

PRIMO WATER CORP /CN/

FORM 8-A12B

(Securities Registration (section 12(b)))

Filed 05/04/18

Address 4221 W. BOY SCOUT BLVD.

SUITE 400

TAMPA, FL, 33607

Telephone 813-313-1732

CIK 0000884713

Symbol PRMW

SIC Code 2086 - Bottled and Canned Soft Drinks and Carbonated Waters

Industry Non-Alcoholic Beverages

Sector Consumer Non-Cyclicals

Fiscal Year 12/28

UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-A

FOR REGISTRATION OF CERTAIN CLASSES OF SECURITIES PURSUANT TO SECTION 12(b) OR (g) OF THE SECURITIES EXCHANGE ACT OF 1934

Cott Corporation

(Exact name of Registrant as specified in its charter)

Canada98-0154711(State or other jurisdiction of incorporation or organization)(I.R.S. Employer Identification No.)

1200 Britannia Rd., East Mississauga, Ontario, Canada

Corporate Center III 4221 W. Boy Scout Blvd., Suite 400 Tampa, Florida, United States

33607 (Zip Code)

(Address of principal executive offices)

Securities to be registered pursuant to Section 12(b) of the Act: Title of each class to be so registered

Name of each exchange on which each class is to be registered

L4W 4T5

Rights attached to Common Shares without nominal or par value pursuant to Rights Agreement New York Stock Exchange

If this form relates to the registration of a class of securities pursuant to Section 12(b) of the Exchange Act and is effective pursuant to General Instruction A.(c) or (e), check the following box. [X]

If this form relates to the registration of a class of securities pursuant to Section 12(g) of the Exchange Act and is effective pursuant to General Instruction A.(d) or (e), check the following box. []

If this form relates to the registration of a class of securities concurrently with a Regulation A offering, check the following box. []

Securities Act registration statement or Regulation A offering statement file number to which this form relates: Not Applicable

Securities to be registered pursuant to Section 12(g) of the Act: None

Item 1. Description of Registrant's Securities to be registered

On May 1, 2018, Cott Corporation (the "Company") entered into a Shareholder Rights Plan Agreement with Computershare Investor Services Inc. (the "Rights Agreement"), which provides for the issuance of one right (a "Right") for each outstanding common share without nominal or par value outstanding at the close of business on May 1, 2018.

The description of the Rights Agreement and the Rights is included in the Company's Current Report on Form 8-K, filed with the U.S. Securities and Exchange Commission on May 4, 2018 (the "Form 8-K") and is incorporated by reference herein. Such description does not purport to be complete and is qualified in its entirety by reference to the Rights Agreement (including the form of Rights Certificate, attached thereto as Exhibit A) which is incorporated herein by reference to Exhibit 4.1 to the Form 8-K.

Item 2. Exhibits

			Incorporated by Reference			Filed Herewith
Exhibit No.	Description	<u>Form</u>	Exhibit	Filing Date	File No.	
3.1	Articles of Amalgamation of Cott Corporation, as amended					*
3.2	By-laws of Cott Corporation, as amended					*
4.1	Shareholder Rights Plan Agreement, dated as of May 1, 2018, between Cott	8-K	4.1	5/4/18	001-31410	
	Corporation and Computershare Investor Services Inc., as Rights Agent					

SIGNATURE

Pursuant to the requirements of Section 12 of the Securities Exchange Act of 1934, the registrant has duly caused this registration statement to be signed on its behalf by the undersigned, thereto duly authorized.

Cott Corporation

May 4, 2018 By: /s/ Marni Morgan Poe

Name: Marni Morgan Poe

Title: Vice President, General Counsel and Secretary

Certificate of Amalgamation

Canada Business Corporations Act

Certificat de fusion

Loi canadienne sur les sociétés par actions

COTT CORPORATION CORPORATION COTT	
Corporate name / Dénomination sociale	
985163-1	
Corporation number / Numéro de société	

I HEREBY CERTIFY that the above-named corporation resulted from an amalgamation, under section 185 of the *Canada Business Corporations Act*, of the corporations set out in the attached articles of amalgamation.

JE CERTIFIE que la société susmentionnée est issue d'une fusion, en vertu de l'article 185 de la *Loi canadienne sur les sociétés par actions*, des sociétés dont les dénominations apparaissent dans les statuts de fusion ci-joints.

/s/ Virginie Ethier

Virginie Ethier

Director / Directeur
2016-08-02

Date of Amalgamation (YYYY-MM-DD)
Date de fusion (AAAA-MM-JJ)





Canada Business Corporations Act (CBCA) FORM 9 ARTICLES OF AMALGAMATION (Section 185)

1 – Corporate name of the amalgamated corporation COTT CORPORATION CORPORATION COTT							
2 – The province or territory in Canada where the registered office is situated (do not indicate the full address) Province of Quebec							
3 – The classes and any maximum number of shares that the corporation is authorized to issue See Schedule "A" annexed to these articles of amalgamation.							
4 – Restrictions, if any, on share transfers None							
5 – Minimum and maximum number of directors (for a fixed number of directors, please indicate the same number in both boxes) Minimum Number 3 Maximum Number 15							
6 – Restrictions, if any, on the business the corporation may carry on None							
7 – Other provisions, if any See Schedule "B" annexed to these articles of amalgamation.							
8 – The amalgamation has been applindicated as follows:	proved pursuant to that section	or subsection of the Act which is					
183 – Long form: approved by special resolution of shareholders	184(1) – Vertical short-for approved by resolution of						
9 - Declaration I hereby certify that I am a director or an authorized officer of the following corporation:							
Name of the amalgamating corporations COTT CORPORATION CORPORATION COTT	Corporation number 440482-3	Signature /s/ Marni Morgan Poe					
Cott Finance Corporation	978054-8	/s/ Marni Morgan Poe					
	fence and, on summary conviction, a	person is liable to a fine not exceeding \$5,000					



SCHEDULE "A" TO THE ARTICLES OF AMALGAMATION OF COTT CORPORATION CORPORATION COTT

- 3. The classes and any maximum number of shares that the Corporation is authorized to issue: The Corporation is authorized to issue the following classes of shares:
 - (a) An unlimited number of first preferred shares, issuable in series ("First Preferred Shares");
 - (b) an unlimited number of first series of First Preferred Shares, designated as Series A Convertible First Preferred Shares ("Series A First Preferred Shares");
 - (c) an unlimited number of second series of First Preferred Shares, designated as Series B Non-Convertible First Preferred Shares ("Series B First Preferred Shares");
 - (d) an unlimited number of Second Preferred Shares, issuable in series ("Second Preferred Shares");
 - (e) an unlimited number of first series of Second Preferred Shares, designated as convertible, participating voting Second Preferred Shares, Series 1 ("Second Preferred Shares, Series 1"); and
 - (f) an unlimited number of common shares (" **Common Shares**").

The rights, privileges, restrictions and conditions attaching to each of the foregoing classes of shares are as follows:

1.0 FIRST PREFERRED SHARES

The First Preferred Shares shall, as a class, carry and be subject to the following rights, privileges, restrictions and conditions:

- 1.1 The First Preferred Shares may be issued at any time and from time to time in one or more series, each series to consist of such number of First Preferred Shares as may, before the issue thereof, be determined by resolution passed by the Board of Directors of the Corporation. The number of shares of any series may from time to time be increased by the Board of Directors of the Corporation upon compliance with the same conditions as are applicable to the issue of shares of a new series.
- 1.2 The Board of Directors of the Corporation shall, subject as hereinafter provided and subject to the provisions of the *Canada Business Corporations Act*, fix, by resolution duly passed before the issue of the First Preferred Shares of each series, the designation, rights, privileges, restrictions and conditions to be attached to the First Preferred Shares of such series, including, but without in any way limiting or restricting the generality of the foregoing:

- (i) provisions, if any, with respect to the rights of the holders of the First Preferred Shares of such series to receive notice of or to attend any meeting of the shareholders of the Corporation or to vote at any such meeting;
- (ii) whether any dividends shall be payable on the First Preferred Shares of such series and, if dividends are to be payable thereon, the rate, amount or method of calculation of preferential dividends, whether fixed or fluctuating, whether cumulative or non-cumulative, whether such dividends are payable in money or by the issue of fully paid shares of the Corporation, the currency or currencies of payment, the date or dates and places of payment of preferential dividends and the date or dates from which such preferential dividends shall accrue;
- (iii) the rights of the Corporation if any, to purchase or redeem the First Preferred Shares of such series, and the purchase or redemption price or the method of calculating the same, and the terms and conditions of any such purchase or redemption;
- (iv) provisions, if any, with respect to the rights of the holders of the First Preferred Shares of such series to tender such shares to the Corporation for purchase by the Corporation and to oblige the Corporation to make such purchase;
- (v) the conversion rights, if any;
- (vi) the terms and conditions of any share purchase plan or sinking fund with respect to the First Preferred Shares of such series; and
- (vii) the restrictions, if any, respecting payment of dividends on the Second Preferred Shares, the Common Shares or on any other shares of the Corporation ranking junior to the First Preferred Shares;

the whole subject to articles of amendment setting forth the designation, rights, privileges, restrictions and conditions to be attached to the First Preferred Shares of such series and the issue of a certificate of amendment in respect thereof.

1.3 The First Preferred Shares shall, with respect to the payment of dividends, be entitled to preference over the Second Preferred Shares, the Common Shares and over any other shares of the Corporation ranking junior to the First Preferred Shares, and no dividends (other than stock dividends payable in shares of the Corporation ranking junior to the First Preferred Shares) shall at any time be declared or paid or set apart for payment on the Second Preferred Shares, the Common Shares or on any other shares of the Corporation ranking junior to the First Preferred Shares, nor shall the Corporation call for redemption or purchase any of the First Preferred Shares (less than the total number of First Preferred Shares then outstanding) or any shares of the Corporation ranking junior to the First Preferred Shares unless at the date of such declaration or call for redemption or purchase, as the case may be, all cumulative dividends up to and including the dividend payment for the last completed period for which such cumulative dividends shall be payable shall have been declared and paid or set apart for payment in respect of each series of cumulative First Preferred Shares then issued and outstanding and any declared and unpaid non-

cumulative dividends shall have been paid or set apart for payment in respect of each series of non-cumulative First Preferred Shares then issued and outstanding.

- 1.4 In the event of the liquidation, dissolution or winding up of the Corporation or other distribution of assets of the Corporation among shareholders for the purpose of winding up its affairs, the holders of the First Preferred Shares shall be entitled to receive, before any amount shall be paid to, or any property or assets of the Corporation distributed among, the holders of the Second Preferred Shares, the Common Shares or any other shares of the Corporation ranking junior to the First Preferred Shares, (i) the amount paid up on such First Preferred Shares, together with, in the case of cumulative First Preferred Shares, all unpaid cumulative dividends (which for such purpose shall be calculated as if such cumulative dividends were accruing from day to day for the period from the expiration of the last period for which cumulative dividends have been paid up to and including the date of distribution) and, in the case of non-cumulative First Preferred Shares, all declared and unpaid non-cumulative dividends, and (ii) if such liquidation, dissolution, winding up or distribution shall be voluntary, an additional amount equal to the premium, if any, which would have been payable on the redemption of said First Preferred Shares respectively if they had been called for redemption by the Corporation on the date of distribution and, if said First Preferred Shares could not be redeemed on such date, then an additional amount equal to the greatest premium, if any, which would have been payable on the redemption of said First Preferred Shares respectively. After payment to the holders of the First Preferred Shares of the amounts so payable to them, they shall not be entitled to share in any further distribution of the property or assets of the Corporation.
- 1.5 The First Preferred Shares of each series shall rank on a parity with the First Preferred Shares of every other series with respect to the payment of dividends and the distribution of assets in the event of the liquidation, dissolution or winding up of the Corporation, whether voluntary or involuntary, or other distribution of assets of the Corporation among shareholders for the purpose of winding up its affairs, provided, however, that in case such assets are insufficient to pay in full the amount due on all the First Preferred Shares, then such assets shall be applied firstly, to the payment equally and rateably of an amount equal to the amount paid up on the First Preferred Shares of each series and the premium thereon, if any, and, secondly, pro rata in the payment of accrued and unpaid cumulative dividends and declared and unpaid non- cumulative dividends.
- 1.6 The holders of the First Preferred Shares shall not, as such, be entitled as of right to subscribe for, or to purchase or receive the whole or any part of any issue of any shares, or of any bonds, debentures or other securities of the Corporation now or hereafter authorized, otherwise than in accordance with the exercise of the conversion rights, if any, which may from time to time attach to any series of first preferred shares.
- 1.7 The provisions contained in Section 1.1 to 1.6 inclusive, and in this Section 1.7 may be repealed or amended in whole or in part by articles of amendment and the issue of a certificate of amendment in respect thereof, but only with the prior approval of the holders of the first preferred shares given as hereinafter specified in addition to any other approval required under the *Canada Business Corporations Act*.

The approval of the holders of the First Preferred Shares as to any and all matters hereinbefore referred to may be given in writing by a resolution signed by all the holders of the First Preferred Shares or by resolution passed by not less than two-thirds (2/3) of the votes cast at a meeting of the holders of the First Preferred Shares duly called for the purpose at which meeting, when originally held, the holders of not less than a majority of the outstanding First Preferred Shares are present in person or represented by proxy in accordance with the by-laws of the Corporation. If at any such meeting, the holders of a majority of the outstanding First Preferred Shares are not present in person or represented by proxy within thirty (30) minutes after the time fixed for the meeting, then the meeting shall be adjourned to such date, being no less than fifteen (15) days later, and to such time and place as may be fixed by the chairman of the meeting and, at such adjourned meeting, the holders of First Preferred Shares present in person or represented by proxy, whether or not they hold more or less than a majority of all First Preferred Shares then outstanding, may transact the business for which the meeting was originally called, and a resolution duly passed thereat by not less than two-thirds (2/3) of the votes cast at such adjourned meeting shall constitute the approval of the holders of the First Preferred Shares hereinbefore mentioned. Notice of any such original meeting of the holders of the First Preferred Shares shall be given not less than twenty-one (21) days nor more than fifty (50) days prior to the date fixed for such meeting and shall state the nature of the business to be transacted and the text of any resolution to be submitted to the meeting. Notice of any such adjourned meeting shall be given: (i) not less than seven (7) days prior to the date fixed for such adjourned meeting, if the original meeting was adjourned by one or more adjournments for an aggregate of less than thirty (30) days, but it shall not be necessary to state in such notice the business for which the adjourned meeting is called or the text of any resolution to be submitted to the adjourned meeting, and (ii) as required by the Canada Business Corporations Act, if the original meeting was adjourned by one or more adjournments for an aggregate of thirty (30) days or more. The formalities to be observed with respect to the giving of notice of any such original meeting or adjourned meeting or any other meeting of the holders of First Preferred Shares and the conduct thereof shall be those from time to time prescribed in the by-laws of the Corporation with respect to meetings of shareholders or in the laws governing the Corporation.

If the repeal or amendment of the provisions hereinbefore contained affects the rights of the holders of First Preferred Shares of any series in a manner different from that in which the rights of the holders of First Preferred Shares of any other series are affected, then such repeal or amendment shall, in addition to being approved by the holders of the First Preferred Shares as hereinabove set forth, be approved by the holders of the First Preferred Shares of such series so affected, and the provisions of this Section 1.7 shall apply, *mutatis mutandis*, with respect to the giving of such approval.

Any meeting of the holders of the outstanding First Preferred Shares may be held at any time and for any purpose, without notice, if all holders of First Preferred Shares entitled to vote at the meeting waive notice of the meeting in writing. For the purpose of waiver of notice, the words "in writing" shall, without limitation, include the sending of a telegram, telex, cable or any other form of written communication by a shareholder. Any holder of First Preferred Shares may waive notice of any meeting either before or after the meeting is held.

Irregularities in the notice or in the giving thereof as well as the accidental omission to give notice of any meeting to, or the non-receipt of any notice by, any holder of First Preferred Shares, shall not invalidate any action taken at any meeting.

At any meeting of the holders of First Preferred Shares without distinction as to series, each holder of First Preferred Shares shall be entitled to 1/25 of a vote for each \$1 (with the Canadian dollar and the United States dollar being deemed to be at par for the purposes of this Section 1.7) paid up on each First Preferred Share held by him. At any meeting of the holders of First Preferred Shares of any particular series, each holder shall be entitled to one (1) vote in respect of each First Preferred Share of such series held by him.

2.0 SERIES A CONVERTIBLE FIRST PREFERRED SHARES

- Authorized. The Corporation shall be authorized to issue an unlimited number of Series A First Preferred Shares. Series A First Preferred Shares converted, redeemed, purchased, or otherwise acquired by the Corporation shall be cancelled and not restored to the status of authorized but unissued shares of the series. The number of authorized Series A First Preferred Shares may be increased or decreased by the approval of the holders of a majority of the outstanding Series A First Preferred Shares, voting as a separate series, provided that there shall be no decrease in the number of authorized Series A First Preferred Shares to a number less than the number of such series then issued and outstanding.
- 2.2 <u>Currency</u>. All payments and distributions on Series A First Preferred Shares or with respect thereto shall be made in United States currency. All references herein to "\$" or "dollars" refer to United States currency.

2.3 Ranking.

- (i) The Series A First Preferred Shares shall, with respect to the payment of dividends and upon Liquidation rank senior to each other class or series of shares of the Corporation (other than the Series B First Preferred Shares), whether now outstanding or hereafter issued, including, without limitation, the Common shares of the Corporation, the Second Preferred Shares and the Second Preferred Shares, Series 1 (all such junior shares being referred to hereinafter collectively as "Junior Shares").
- (ii) The Series A First Preferred Shares shall, with respect to the payment of dividends and rights upon Liquidation rank equally with the Series B First Preferred Shares of the Corporation, whether now outstanding or hereafter issued.

2.4 <u>Dividends</u>.

- (i) The holders of Series A First Preferred Shares shall be entitled to receive a quarterly fixed cumulative dividend in an amount equal to the applicable Dividend Rate (as defined herein) then in effect (the "**Dividends**"),
- (ii) Subject to applicable law, Dividends shall be declared and payable quarterly in arrears on January 1, April 1, July 1 and October 1 of each year (unless any such

day is not a Business Day, in which event such Dividends shall be payable on the next succeeding Business Day, without accrual to the actual payment date), commencing on January 1, 2015 (each such payment date being a " **Dividend Payment Date**," and the period from the Issue Date of the Series A First Preferred Shares to the first Dividend Payment Date and each such quarterly period thereafter being a " **Dividend Period**"). The amount of Dividends payable on the Series A First Preferred Shares for any period shall be computed on the basis of a 360-day year and the actual number of days elapsed. Subject to Section 2.6 (iii)(G)(2) herein, such Dividends shall be paid to holders of Series A First Preferred Shares before any dividends shall be declared, set apart for or paid upon any Junior Shares.

- (iii) Dividends, whether or not declared, shall begin to accrue and be cumulative from the Issue Date and shall compound at the relevant rate on each subsequent Dividend Payment Date (i.e., no Dividends shall accrue on another Dividend unless and until any Dividend Payment Date for such other Dividends has passed without such other Dividends having been paid on such date, in which case Dividends will accrue on such unpaid Dividends). If the Corporation does not pay any Dividend in full on any scheduled Dividend Payment Date, such unpaid Dividend will accrue at the then applicable Dividend Rate plus an additional two and one half percent (2.5%) per annum of the Redemption Value of each Series A First Preferred Share from such scheduled Dividend Payment Date to the date that all accumulated Dividends on the Series A First Preferred Shares have been paid in cash in full. Any Dividends accrued and unpaid pursuant to this Section 2.4(iii) on each Dividend Payment Date shall be added to the Redemption Value until such Dividends are paid in cash in full. For the avoidance of doubt, Dividends shall accumulate whether or not in any Dividend Period there have been funds of the Corporation legally available for the payment of such Dividends.
- (iv) Except as otherwise provided herein, if at any time the Corporation pays less than the total amount of Dividends then accumulated, such payment shall be distributed pro rata among the holders of Series A First Preferred Shares based upon the Redemption Value on the Series A First Preferred Shares then held by each such holder. When Dividends are not paid in full, all Dividends and any dividends on Series B First Preferred Shares shall be paid pro rata so that the amount of Dividends and any dividends on Series B First Preferred Shares shall in all cases bear to each other the same ratio as accumulated Dividends and accumulated dividends on Series B First Preferred Shares bear to each other.
- (v) All Dividends shall be paid in cash.
- (vi) Dividends shall be payable to the holders of record of Series A First Preferred Shares as they appear on the register of the Corporation at the Close of Business on the relevant record date, which shall be not more than thirty (30) days nor less than ten (10) days preceding the applicable Dividend Payment Date.
- 2.5 <u>Liquidation, Dissolution or Winding Up</u>.

- (i) Upon any voluntary or involuntary liquidation, dissolution or winding up of the Corporation or any return of capital or other distribution of assets of the Corporation among its shareholders for the purpose of winding up its affairs (each, a "**Liquidation**"), the holders of Series A First Preferred Shares shall be entitled to receive, out of the assets of the Corporation or proceeds thereof (whether capital or surplus) legally available therefor, an amount per Series A First Preferred Share equal to the greater of:
 - (A) the Redemption Value per share, plus an amount equal to any Dividends accumulated but unpaid thereon (whether or not declared) through the date of Liquidation; and
 - (B) the payment that holders of Series A First Preferred Shares would have received had such holders, immediately prior to such Liquidation converted their respective Series A First Preferred Shares into Common Shares (at the then applicable Conversion Rate) pursuant to Section 2.7 immediately prior to such Liquidation.

(the greater of (A) and (B) is referred to herein as the "Liquidation Preference"). Holders of Series A First Preferred Shares will not be entitled to any other amounts from the Corporation after they have received the full amounts provided for in this Section 2.5(i) and will have no right or claim to any of the Corporation's remaining assets.

- (ii) If, in connection with any distribution described in Section 2.5(i) above, the assets of the Corporation or proceeds thereof are not sufficient to pay in full the Liquidation Preference payable on the Series A First Preferred Shares and the corresponding amounts payable on the Series B First Preferred Shares, then such assets, or the proceeds thereof, shall be paid pro rata in accordance with the full respective amounts which would be payable on such shares if all amounts payable thereon were paid in full.
- (iii) For purposes of this Section 2.5, the merger or consolidation of the Corporation with or into any other corporation or other entity, or the sale, conveyance, lease or other disposition of all or substantially all of the assets of the Corporation, shall not constitute a Liquidation of the Corporation.

2.6 <u>Voting Rights</u>.

- (i) Subject to Section 2.6(ii) and in addition to class voting rights pursuant to applicable law, the holders of Series A First Preferred Shares shall be entitled to
 - (A) vote (in person or by proxy) with the holders of the Common Shares, voting together as a single class, on all matters submitted for a vote of holders of Common Shares, (B) that number of votes per Series A First Preferred Share as is equal to the number of Common Shares into which each such Series A First Preferred Share is then convertible at the time of the relevant record date (at the then applicable Conversion Rate) and (C) notice of all shareholders' meetings (or pursuant to any action by written

consent) in accordance with the Corporation's Articles of Amalgamation and By-laws as if the holders of Series A First Preferred Shares were holders of Common Shares.

- (ii) Subject to applicable law, Section 2.6(iii), Section 2.6(iv) and any other provisions to the contrary herein:
 - (A) from and after the Issue Date until June 12, 2016, the holders of shares of Series A First Preferred Shares shall have no voting rights pursuant to Sections 2.6(i)(A) and (B) above;
 - (B) from June 13, 2016 until December 12, 2017, the holders of Series A First Preferred Shares shall have voting rights pursuant to Sections 2.6(i)(A) and (B) above; **provided**, **however**, that, with respect to the election of directors to the Board only and subject to Section 2.6(iv), each holder of Series A First Preferred Shares shall cast votes on his, her or its shares in the same proportion as the Corporation's holders of Common Shares (e.g., if a director nominee receives ninety percent (90%) of the votes cast by the holders of Common Shares, with five percent (5%) against and five percent (5%) abstaining, then each holder of Series A First Preferred Shares shall cast ninety percent (90%) of his, her or its votes for such director nominee, five percent (5%) against, and abstain from voting Series A First Preferred Shares equivalent to five percent (5%) of the votes on an as-converted basis); and
 - (C) after December 12, 2017, the holders of shares of Series A First Preferred Shares shall have voting rights pursuant to Sections 2.6(i)(A) and (B) above without restriction.
- (iii) In addition to, and not in substitution for, the rights and privileges afforded to the holders of a class of issued shares under applicable law, for so long as any Series A First Preferred Shares remain outstanding, the Corporation shall not and shall not permit any direct or indirect Subsidiary of the Corporation to, without first obtaining the written consent or affirmative vote at a meeting called for that purpose by holders of at least two-thirds (2/3) of the then outstanding shares of Series A First Preferred Shares voting separately as a series, take any of the following actions:
 - (A) Any change, amendment, alteration or repeal (including, without limitation, as a result of a merger, consolidation, or other similar or extraordinary transaction) of any provisions of the Corporation's Articles of Amalgamation or By-laws that amends, modifies or adversely affects the rights, preferences, privileges, conditions or voting powers of the Series A First Preferred Shares;
 - (B) Any issuance or reclassification of shares of the Corporation (including, without limitation, as a result of a merger, consolidation, or other similar

or extraordinary transaction) that would rank equal or senior to the Series A First Preferred Shares, with respect to the Liquidation of the Corporation or with respect to the declaration or payment of dividends, including, for the avoidance of doubt, the issuance of any Series A First Preferred Shares or Series B First Preferred Shares;

- (C) Any voluntary initiation of any Liquidation of the Corporation or any of its Significant Subsidiaries or any group of Subsidiaries that, taken together, would constitute a Significant Subsidiary or commencement of a proceeding for bankruptcy, insolvency, receivership or similar action with respect to the Corporation or any of its Significant Subsidiaries or any group of Subsidiaries that, taken together, would constitute a Significant Subsidiary;
- (D) Any material change to the tax or accounting policies of the Corporation, except any changes required by applicable law or pursuant to GAAP;
- (E) Except as provided in Section 2.6(iv) or Section 2.6(xi) below, any increase in the number of members of the Board if such increase would result in the Board having more than eleven (11) members;
- (F) Any incurrence, assumption or refinancing of any Indebtedness of the Corporation or any of its Subsidiaries to the extent that, immediately after giving effect to such incurrence, assumption or refinancing, the ratio of (1) the consolidated Indebtedness of the Corporation and its Subsidiaries that would be stated on the balance sheet of the Corporation and its Subsidiaries as of the date of determination (determined on a consolidated basis in accordance with GAAP) minus the lesser of (i) the amount of cash and cash equivalents (except for Restricted Cash) that would be stated on the balance sheet of the Corporation and its Subsidiaries and held by the Corporation and its Subsidiaries as of the date of determination (determined on a consolidated basis in accordance with GAAP) and (ii) \$75,000,000 to (2) the EBITDA of the Corporation and its Subsidiaries for the most recently ended four (4) fiscal quarters for which internal financial statements are available immediately preceding the date on which such additional Indebtedness is incurred, assumed or refinanced (the "Reference **Period** "), would not be less than a ratio of 5.0-to-1.0 on a pro forma basis (calculated in accordance with GAAP and Regulation S-X promulgated by the United States Securities and Exchange Commission) after giving effect to such incurrence, assumption or refinancing and the use of the proceeds therefrom as if such events had occurred on the first day of the Reference Period, provided, however, in the case of any revolving credit facility, Indebtedness shall be deemed to be incurred on the date that the commitments are entered into and not on the date of subsequent borrowings and shall be deemed to be outstanding continuously until such revolving credit facility is terminated and, for this

purpose, the amount of the commitment under the Credit Agreement shall be deemed to be equal to \$328,000,000.

- (G) Any declaration or payment of any dividend or distribution (or any money made available for a sinking fund for the payment of any dividend or distribution) upon any Junior Shares, or the redemption of any Junior Shares, or the purchase for cancellation, payment or acquisition for any consideration (or any money made available for a sinking fund for the redemption) of any Junior Shares by the Corporation, directly or indirectly; <u>provided</u>, <u>however</u>, that the foregoing limitation shall not apply to:
 - (1) purchases, redemptions or other acquisitions of Junior Shares in connection with any employment contract, benefit plan or other similar arrangement with or for the benefit of any one or more employees, officers, directors, managers or consultants of or to the Corporation or any of its Subsidiaries; or
 - regular quarterly dividends to holders of Common Shares in an amount not to exceed US\$0.06 per share; <u>provided</u>, <u>however</u>, that no dividends shall be declared or paid without a vote of the holders of Series A First Preferred Shares pursuant to this Section 2.6(iii) if one or more quarterly Dividends have not been paid in full on any Dividend Payment Date until such time as all such Dividends have been paid in full; or
- (H) Any issuance of Equity Securities by any Subsidiary of the Corporation to any Person other than the Corporation or a direct or indirect wholly-owned Subsidiary of the Corporation, any acquisition of Equity Securities of any Person other than a wholly-owned Subsidiary of the Corporation or any of its direct or indirect wholly-owned Subsidiaries, or any transfer or other transaction (including, without limitation, as a result of a merger, consolidation, or other similar or extraordinary transaction) resulting in the Corporation or any of its Subsidiaries owning Equity Securities of any Person other than a wholly- owned Subsidiary of the Corporation or any of its direct or indirect wholly-owned Subsidiaries.
- (iv) Notwithstanding Section 2.6(iii)(E) above and any other restrictions on the right to vote set forth herein, if at any time Dividends payable in respect of the Series A First Preferred Shares have not been declared and paid for the equivalent of six (6) or more Dividend Periods, whether or not for consecutive Dividend Periods (a "Preferred Dividend Default"), then the holders of Series A First Preferred Shares shall be entitled to, voting separately as a series (in person or by proxy), nominate and elect two (2) additional directors to serve on the Board (each, a "Preferred Share Director") and are entitled to receive notice of and attend all meetings of shareholders of the Corporation at which directors are to be elected. In such a case, the number of directors serving on the Board at that time shall be

increased by two (2), and, for clarity, no sitting director of the Corporation will be forced to resign to accommodate a Preferred Share Director.

- (v) The election of Preferred Share Directors upon a Preferred Dividend Default shall take place:
 - (A) if the Corporation receives the request for a special meeting more than ninety (90) calendar days before the date fixed for the Corporation's next annual meeting of shareholders, at a special meeting for such purpose called by the holders of at least ten percent (10%) of the outstanding Series A First Preferred Shares, or, if the Corporation receives the request for a special meeting within ninety (90) calendar days before the date fixed for the Corporation's next annual or special meeting of shareholders, at such annual or special meeting of shareholders; and
 - (B) at each subsequent meeting (or special meeting held in its place) until all accrued and unpaid Dividends for all complete Dividend Periods prior to the date of payment plus the Dividend for the then current Dividend Period have been paid in full, or declared and a sum sufficient for such payment in cash is set aside for payment.
- (vi) At any meeting held for the purpose of electing a Preferred Share Director, the presence in person or by proxy of the holders of at least a majority of the outstanding Series A First Preferred Shares shall be required to constitute a quorum and the Preferred Share Directors shall be elected by a majority of the votes cast (in person or by proxy) by the holders of Series A First Preferred Shares, voting separately as a series. Despite the prior sentence, if proper notice of a meeting of holders of Series A First Preferred Shares is given, specifying the purpose of or the business to be transacted at the meeting, and a quorum of holders of Series A First Preferred Shares is not present, a second meeting of holders of Series A First Preferred Shares may be held on forty eight (48) hours written notice to transact the business specified in the original notice. Subject to applicable law, the holders of Series A First Preferred Shares present at the second meeting constitute a quorum and the business specified in the original notice may be transacted by a majority vote of those holders of Series A First Preferred Shares in attendance (in person or by proxy) at the second meeting, voting as a class. In addition to the foregoing, holders of Series A First Preferred Shares shall be entitled to one (1) vote (in person or by proxy) for each Series A First Preferred Share held on any matter submitted for a vote by the Board.
- (vii) Preferred Share Directors shall hold office from election until the earlier of:
 - (A) his or her resignation, death, bankruptcy or incapacity;
 - (B) the election of his/her successor in the manner as provided herein;
 - (C) his/her removal by holders of a majority of Series A First Preferred Shares, voting as a separate class; or

- (D) the date on which all such unpaid Dividends have been paid in full, or declared and a sum sufficient for such payment in cash is set aside for payment to the holders of the Series A First Preferred Shares and the Series B First Preferred Shares.
- (viii) Any vacancy occurring among the Preferred Share Directors may only be filled by the holders of the Series A First Preferred Shares, voting separately as a class.
- (ix) The right of the holders of the Series A First Preferred Shares to elect the Preferred Share Directors continues until the date on which all such unpaid Dividends have been paid in full, or declared and a sum sufficient for such payment in cash is set aside for payment to the holders of the Series A First Preferred Shares and shall not apply at any meeting if or to the extent that the Preferred Share Directors, whose terms of office do not expire at such meeting, have been previously elected by the holders of the Series A First Preferred Shares, voting separately as a class. Notwithstanding anything contained in the Corporation's Articles of Amalgamation or By-laws, upon any termination of the right of the holders of Series A First Preferred Shares to elect the Preferred Share Directors as provided in this section, the term of office of any Preferred Share Director shall terminate and the entire Board shall be reduced accordingly so that there is no vacancy on the Board.
- (x) If and when all such Dividends have been paid in full, or declared and a sum sufficient for such payment in cash is set aside for payment, holders of the Series A First Preferred Shares shall be divested of the voting rights set forth in Section 2.6(iv) (subject to re-vesting in the event of any subsequent Preferred Dividend Defaults).
- (xi) Section 2.6(iii)(E) shall not apply to any increase in the number of members of the Board resulting from contractual rights of the initial holders of Series A Preferred Shares to appoint directors to the Board after conversion of the Series A Preferred Shares to Common Shares.

2.7 <u>Conversion</u>.

(i) Optional Conversion. At any time following the third (3rd) anniversary of the Issue Date, each holder of Series A First Preferred Shares shall have the right, at such holder's option, to convert any or all of such holder's Series A First Preferred Shares into a number of Common Shares equal to the product of the aggregate Redemption Value of the Series A First Preferred Shares to be converted divided by US\$1,000.00 multiplied by the Conversion Rate then in effect, plus cash in lieu of fractional shares, as set out in Section 2.10(ix) plus an amount in cash per Series A First Preferred Share equal to all accrued but unpaid Dividends (whether or not declared) on such Series A First Preferred Share to but excluding the applicable Conversion Date, out of funds legally available therefor. If the Corporation does not pay in cash such accrued but unpaid dividends on such shares, then, without limiting the right of the holders of Series A First Preferred

Shares hereunder, at the option of each holder of the Series A First Preferred Shares, such amount may be added to the Redemption Value of the Series A First Preferred Shares for the purposes of determining the amount of Common Shares issuable upon conversion of the Series A First Preferred Shares.

- (ii) <u>Conversion Rate</u>. The "Conversion Rate" means 159.236 shares, subject to adjustment in accordance with the rights, privileges, restrictions and conditions of the Series A First Preferred Shares as set out herein.
- (iii) <u>Conversion Procedures</u>. A holder must do each of the following in order to convert its Series A First Preferred Shares pursuant to this Section 2.7:
 - (A) complete and manually sign the conversion notice provided by the Conversion Agent, and deliver such notice to the Conversion Agent;
 - (B) deliver to the Conversion Agent the certificate or certificates representing the Series A First Preferred Shares to be converted (or, if such certificate or certificates have been lost, stolen or destroyed, a lost certificate affidavit and indemnity in form and substance reasonably acceptable to the Corporation);
 - (C) if required, furnish appropriate endorsements and transfer documents in form and substance reasonably acceptable to the Corporation; and
 - (D) if required, pay any share transfer, documentary, stamp or similar taxes not payable by the Corporation pursuant to Section 2.8.

The "Conversion Date" means the date on which a holder complies in all respects with the procedures set forth in this Section 2.7(iii)

- (iv) <u>Effect of Conversion</u>. Effective immediately prior to the Close of Business on the Conversion Date applicable to any Series A First Preferred Shares, but subject to the actual conversion occurring, Dividends shall no longer accrue or be declared on any Series A First Preferred Shares so converted and such Series A First Preferred Shares shall cease to be outstanding.
- (v) Record Holder of Underlying Securities as of Conversion Date. The Person or Persons entitled to receive the Common Shares and, to the extent applicable, cash, issuable upon conversion of Series A First Preferred Shares on a Conversion Date shall be treated for all purposes as the record holder(s) of such Common Shares and/or cash as of the Close of Business on such Conversion Date. As promptly as practicable on or after the Conversion Date and compliance by the applicable holder with the relevant conversion procedures contained in Section 2.7(iii) (and in any event no later than three (3) Trading Days thereafter), the Corporation shall issue the number of Common Shares issuable upon conversion (and deliver payment of cash in lieu of fractional shares). Such delivery of Common Shares and, if applicable, cash shall be made, at the option of the applicable holder, in certificated form or by book-entry. Any such certificate or certificates shall be

delivered by the Corporation to the appropriate holder on a book-entry basis or by mailing certificates evidencing the shares to the holders at their respective addresses as set forth in the conversion notice. If fewer than all of the Series A First Preferred Shares held by any holder hereto are converted pursuant to Section 2.7(i) then a new certificate representing the unconverted Series A First Preferred Shares shall be issued to such holder concurrently with the issuance of the certificates (or book-entry shares) representing the applicable Common Shares. In the event that a holder shall not by written notice designate the name in which Common Shares and, to the extent applicable, cash to be delivered upon conversion of Series A First Preferred Shares should be registered or paid, or the manner in which such shares and, if applicable, cash should be delivered, the Corporation shall be entitled to register and deliver such shares and, if applicable, cash in the name of the holder and in the manner shown on the records of the Corporation.

(vi) <u>Status of Converted or Acquired Shares</u>. Series A First Preferred Shares duly converted in accordance with the rights, privileges, restrictions and conditions of the Series A First Preferred Shares as set out herein, or otherwise acquired by the Corporation in any manner whatsoever, shall be retired promptly after the acquisition thereof.

2.8 <u>Taxes</u>.

- (i) Absent a change in law or Internal Revenue Service practice, or a contrary determination (as defined in Section 1313(a) of the United States Internal Revenue Code of 1986, as amended (the "Code")), each holder of Series A First Preferred Shares and the Corporation agree not to treat the Series A First Preferred Shares (based on their terms as set forth herein) as "preferred stock" within the meaning of Section 305 of the Code, and Treasury Regulation Section 1.305-5 for United States federal income tax and withholding tax purposes and shall not take any position inconsistent with such treatment.
- (ii) The Corporation shall pay any and all documentary, stamp and similar issue or transfer tax due on (x) the issue of the Series A First Preferred Shares and (y) the issue of Common Shares upon conversion of the Series A First Preferred Shares. However, in the case of conversion of Series A First Preferred Shares, the Corporation shall not be required to pay any tax or duty that may be payable in respect of any transfer involved in the issue and delivery of Common Shares or Series A First Preferred Shares in a name other than that of the holder of the shares to be converted, and no such issue or delivery shall be made unless and until the person requesting such issue has paid to the Corporation the amount of any such tax or duty, or has established to the satisfaction of the Corporation that such tax or duty has been paid.
- (iii) Each holder of Series A First Preferred Shares and the Corporation agree to cooperate with each other in connection with any redemption of part of Series A First Preferred Shares and to use good faith efforts to structure such redemption

so that such redemption may be treated as a sale or exchange pursuant to Section 302 of the Code; <u>provided</u>, that nothing in this Section 2.8 shall require the Corporation to purchase any Series A First Preferred Shares; and <u>provided</u>, <u>further</u>, that the Corporation makes no representation or warranty in this Section 2.8 regarding the tax treatment of any redemption of Series A First Preferred Shares.

2.9 <u>Redemption and Repurchase</u>.

- (i) Optional Redemption. The Series A First Preferred Shares may be redeemed, in whole or in part, at any time, at the option of the Corporation, upon giving notice of redemption pursuant to Section 2.9(iii), at a redemption price per share equal to (A) the sum of the Redemption Value per share of the Series A First Preferred Shares to be redeemed plus (B) an amount equal to any Dividends accumulated but unpaid thereon (whether or not declared) to but excluding the date of redemption; provided, however, that the Series A First Preferred Shares shall not be redeemed, in whole or in part, at any time any Series B First Preferred Shares of the Corporation remain issued and outstanding. Notwithstanding anything herein to the contrary, at any time the Series A First Preferred Shares are convertible pursuant to Section 2.7(i) the holders of Series A First Preferred Shares may, at their option, elect to exercise any conversion rights for a period of thirty (30) days following the receipt of notice of redemption as provided in Section 2.9(iii) in lieu of the redemption of any Series A First Preferred Shares.
- (ii) Repurchase at the Option of the Holder Upon a Change of Control. Upon the occurrence of a Change of Control, each holder of Series A First Preferred Shares shall have the right to require the Corporation to repurchase, by irrevocable written notice to the Corporation, all or any portion of such holder's Series A First Preferred Shares at a purchase price per share equal to the sum of (A) the Redemption Value per share of the Series A First Preferred Shares plus (B) an amount equal to accrued but unpaid Dividends (whether or not declared) on such Series A First Preferred Shares to but excluding the date of repurchase.

No earlier than sixty (60) days and no later than thirty (30) days prior to the occurrence of a Change of Control, the Corporation shall send notice by first- class mail, postage prepaid, addressed to the holders of record of the Series A First Preferred Shares at their respective last addresses appearing on the books of the Corporation stating (A) that a Change of Control is expected to occur, (B) the date that the Change of Control is expected to occur, (C) that all Series A First Preferred Shares tendered prior to a Business Day no earlier than thirty (30) days nor later than sixty (60) days from the date such notice is mailed shall be accepted for repurchase and (D) the procedures that holders of the Series A First Preferred Shares must follow in order for their Series A First Preferred Shares to be repurchased, including the place or places where certificates for such shares are to be surrendered for payment of the repurchase price. Any notice mailed as provided in this Subsection shall be conclusively presumed to have been duly given, whether or not the holder receives such notice, but failure duly to give such

notice by mail, or any defect in such notice or in the mailing thereof, to any holder of Series A First Preferred Shares designated for repurchase shall not affect the validity of the proceedings for the repurchase of any other Series A First Preferred Shares. The repurchase of Series A First Preferred Shares pursuant to this Section 2.9(ii) shall occur simultaneously with the consummation of the Change of Control.

- Notice of Redemption at the Option of the Corporation . Notice of every redemption of Series A First Preferred (iii) Shares pursuant to Section 2.9(i) shall be given by first-class mail, postage prepaid, addressed to the holders of record of the shares to be redeemed at their respective last addresses appearing on the books of the Corporation. Such mailing shall be at least thirty (30) days and not more than sixty (60) days before the date fixed for redemption. Any notice mailed as provided in this Section 2.9(iii) shall be conclusively presumed to have been duly given, whether or not the holder receives such notice, but failure duly to give such notice by mail, or any defect in such notice or in the mailing thereof, to any holder of Series A First Preferred Shares designated for redemption shall not affect the validity of the proceedings for the redemption of any other Series A First Preferred Shares. Each notice of redemption given to a holder shall state: (A) the redemption date; (B) the number of the Series A First Preferred Shares to be redeemed and, if less than all the shares held by such holder are to be redeemed, the number of such shares to be redeemed from such holder; (C) the redemption price; and (D) the place or places where certificates for such shares are to be surrendered for payment of the redemption price in respect of such shares. Notwithstanding the foregoing, at any time after the Series A First Preferred Shares become convertible pursuant to Section 2.7, the Corporation may not redeem any Series A First Preferred Shares unless such holders have failed to exercise their conversion rights pursuant to Section 2.7(i) for a period of thirty (30) days following receipt of a notice of redemption.
- (iv) <u>Partial Redemption</u>. In case of any redemption of part of the Series A First Preferred Shares at the time outstanding pursuant to this Section 2.9, the shares to be redeemed shall be redeemed pro rata. Subject to the provisions hereof, the Corporation shall have full power and authority to prescribe the terms and conditions upon which the Series A First Preferred Shares shall be redeemed from time to time. If fewer than all the shares represented by any certificate are redeemed, a new certificate shall be issued representing the unredeemed shares without charge to the holder thereof.
- (v) <u>Effectiveness of Redemption</u>. If notice of redemption has been duly given and if on or before the redemption date specified in the notice all funds necessary for the redemption have been deposited by the Corporation, in trust for the pro rata benefit of the holders of the shares called for redemption, with a bank or trust company doing business in the Borough of Manhattan, The City of New York, and having a capital and surplus of at least US \$1 billion and selected by the Board, so as to be and continue to be available solely therefor, then, notwithstanding that any certificate for any share so called for redemption has not

been surrendered for cancellation, on and after the redemption date dividends shall cease to accrue on all shares so called for redemption, all shares so called for redemption shall no longer be deemed outstanding and all rights with respect to such shares shall forthwith on such redemption date cease and terminate, except only the right of the holders thereof to receive the amount payable on such redemption from such bank or trust company, without interest. Any funds unclaimed at the end of three (3) years from the redemption date shall, to the extent permitted by law, be released to the Corporation, after which time the holders of the shares so called for redemption shall look only to the Corporation for payment of the redemption price of such shares. Notwithstanding the foregoing, at any time after the Series A First Preferred Shares become convertible pursuant to Section 2.7, the Corporation may not redeem any Series A First Preferred Shares unless such holders have failed to exercise their conversion rights pursuant to Section 2.7(i) for a period of thirty (30) days following receipt of a notice of redemption.

2.10 Anti-Dilution Provisions.

- (i) <u>Adjustments</u>. The Conversion Rate will be subject to adjustment, without duplication, under the following circumstances:
 - (A) the issuance of Common Shares as a dividend or distribution to all or substantially all holders of Common Shares, or a subdivision or combination of Common Shares or a reclassification of Common Shares into a greater or lesser number of Common Shares, in which event the Conversion Rate will be adjusted based on the following formula:

$$CR_i = CR_0 \times \frac{OS_i}{OS_0}$$

where,

CR 0 = the Conversion Rate in effect immediately prior to the Close of Business on (1) the Record Date for such dividend or distribution, or (2) the effective date of such subdivision, combination or reclassification;

CR 1 = the new Conversion Rate in effect immediately after the Close of Business on (1) the Record Date for such dividend or distribution, or (2) the effective date of such subdivision, combination or reclassification;

OS 0 = the number of Common Shares outstanding immediately prior to the Close of Business on (1) the Record Date for such dividend or distribution or (2) the effective date of such subdivision, combination or reclassification; and

OS 1 = the number of Common Shares that would be outstanding immediately after, and solely as a result of, the completion of such event

(including, for the avoidance of doubt, a number of Common Shares equal to OS 0 in the event of a dividend or distribution that does not involve the surrender or exchange of Common Shares).

Any adjustment made pursuant to this clause (A) shall be effective immediately prior to the Open of Business on the Trading Day immediately following the Record Date, in the case of a dividend or distribution, or the effective date in the case of a subdivision, combination or reclassification. If any such event is declared but does not occur, the Conversion Rate shall be readjusted, effective as of the date the Board announces that such event shall not occur, to the Conversion Rate that would then be in effect if such event had not been declared.

(B) the dividend, distribution or other issuance to all or substantially all holders of Common Shares of rights (other than a distribution of rights issued pursuant to a shareholders rights plan, to the extent such rights are attached to Common Shares (in which event the provisions of Section 2.10(i)(E) shall apply)), options or warrants entitling them to subscribe for or purchase Common Shares for a period expiring sixty (60) days or less from the date of issuance thereof, at a price per share that is less than the Market Price on the Trading Day immediately preceding the Record Date for such issuance, in which event the Conversion Rate will be increased based on the following formula:

$$CR_{+} = CR_{0} \times \frac{(OS_{0} + X)}{(OS_{0} + Y)}$$

where,

CR 0 = the Conversion Rate in effect immediately prior to the Close of Business on the Record Date for such dividend, distribution or issuance;

CR 1 = the new Conversion Rate in effect immediately following the Close of Business on the Record Date for such dividend, distribution or issuance;

OS 0 = the number of Common Shares outstanding immediately prior to the Close of Business on the Record Date for such dividend, distribution or issuance;

X = the total number of Common Shares issuable pursuant to such rights, options or warrants; and

Y = the number of Common Shares equal to the aggregate price payable to exercise such rights, options or warrants divided by the Market Price on the Trading Day immediately preceding the Record Date for such dividend, distribution or issuance.

For purposes of this clause (B), in determining whether any rights, options or warrants entitle the holders to purchase the Common Shares at a price per share that is less than the Market Price on the Trading Day immediately preceding the Record Date for such dividend, distribution or issuance, there shall be taken into account any consideration the Corporation receives for such rights, options or warrants, and any amount payable on exercise thereof, with the value of such consideration, if other than cash, to be the fair market value thereof (as determined in good faith by the Board).

Any adjustment made pursuant to this clause (B) shall become effective immediately prior to the Open of Business on the Trading Day immediately following the Record Date for such dividend, distribution or issuance. In the event that such rights, options or warrants are not so issued, the Conversion Rate shall be readjusted, effective as of the date the Board publicly announces its decision not to issue such rights, options or warrants, to the Conversion Rate that would then be in effect if such dividend, distribution or issuance had not been declared. To the extent that such rights, options or warrants are not exercised prior to their expiration or Common Shares are otherwise not delivered pursuant to such rights, options or warrants upon the exercise of such rights, options or warrants, the Conversion Rate shall be readjusted to the Conversion Rate that would then be in effect had the adjustments made upon the dividend, distribution or issuance of such rights, options or warrants been made on the basis of the delivery of only the number of Common Shares actually delivered.

(C) the Corporation or one or more of its subsidiaries make purchases of Common Shares pursuant to a tender offer or exchange offer by the Corporation or a subsidiary of the Corporation for all or any portion of the Common Shares, where the cash and value of any other consideration included in the payment per Common Share validly tendered or exchanged exceeds the Market Price of the Common Shares on the Trading Day prior to the last day (the "Expiration Date") on which tenders or exchanges may be made pursuant to such tender or exchange offer (as it may be amended), in which event the Conversion Rate will be increased based on the following formula:

$$CR_{+} = CR_{0} \times \frac{(AC + (SP_{+} \times OS_{+}))}{(OS_{0} \times SP_{+})}$$

where,

CR 0 = the Conversion Rate in effect immediately prior to the Close of Business on the Expiration Date;

CR 1 = the new Conversion Rate in effect immediately after the Close of Business on the Expiration Date;

AC = the fair market value (as determined in good faith by the Board), on the Expiration Date, of the aggregate value of all cash and any other consideration paid or payable for shares validly tendered or exchanged and not withdrawn as of the Expiration Date (the "**Purchased Shares**");

OS 1 = the number of Common Shares outstanding as of the last time tenders or exchanges may be made pursuant to such tender or exchange offer (the "**Expiration Time**"), excluding any Purchased Shares;

OS 0 = the number of Common Shares outstanding immediately before the Expiration Time, including any Purchased Shares; and

SP 1 = the arithmetic average of the VWAP (as defined below) for each of the ten (10) consecutive full Trading Days ending on the Trading Day immediately succeeding the Expiration Date.

Any adjustment made pursuant to this clause (C) shall become effective immediately prior to the Open of Business on the Trading Day immediately following the Expiration Date. In the event that the Corporation or any of its subsidiaries is obligated to purchase Common Shares pursuant to any such tender offer or exchange offer but is permanently prevented by applicable law from effecting any such purchases, or all such purchases are rescinded, then the Conversion Rate shall be readjusted to be the Conversion Rate that would then be in effect if such tender offer or exchange offer had not been made.

(D) the Corporation shall, by dividend or otherwise, distribute to all or substantially all holders of its Common Shares (subject to an exception for cash in lieu of fractional shares) shares of any class of Equity Securities (other than Common Shares as covered by Section 2.10(i)(A)), evidences of its indebtedness, cash, assets, other property or securities or rights, options or warrants to acquire Equity Securities or other securities, but excluding (A) dividends or distributions referred to in Section 2.10 (i)(A) hereof, (B) rights, options or warrants referred to in Section 2.10(i)(B) hereof or distributed in connection with a shareholder rights plans (in which event the provisions of Section 2.10(i)(E) to the extent applicable shall apply) and (3) regular quarterly dividends or distributions paid exclusively in cash in an amount not to exceed US\$0.06 per share, then, in each such case the Conversion Rate shall be adjusted based on the following formula:

$$CR_+ = CR_0 \times \frac{SP_0}{(SP_0 - C)}$$

where,

CR 0 = the Conversion Rate in effect immediately prior to the Close of Business on the Record Date for such dividend or distribution;

CR1 = the new Conversion Rate in effect immediately after the Close of Business on the Record Date for such dividend or distribution;

SP 0 = the Market Price on the Trading Day immediately preceding the Record Date for such dividend or distribution; and

C = the fair market value (as determined in good faith by the Board) of the portion of Distributed Property distributed with respect to each outstanding Common Share on the Record Date for such dividend or distribution; <u>provided</u>, that, if C is equal to or greater than SP 0, then in lieu of the foregoing adjustment, the Corporation shall distribute to each holder of Series A First Preferred Shares on the date the applicable Distributed Property is distributed to holders of Common Shares, but without requiring such holder to convert its Series A First Preferred Shares the amount of Distributed Property such holder would have received had such holder owned a number of Common Shares equal to the Conversion Rate on the Record Date fixed for determination for shareholders entitled to receive such distribution.

(E) If the Corporation has a shareholder rights plan in effect with respect to the Common Shares on the Conversion Date, upon conversion of any Series A First Preferred Shares, holders of such shares will receive, in addition to the shares of Common Shares, the rights under such rights plan relating to such Common Shares, unless, prior to the Conversion Date, the rights have (1) become exercisable or (2) separated from the Common Shares (the first of such events to occur being the "Trigger Event"), in either of which cases the Conversion Rate will be adjusted, effective automatically at the time of such Trigger Event, as if the Corporation had made a distribution of such rights to all holders of the Common Shares as described in Section 2.10(i)(B) (without giving effect to the sixty (60) day limit on the exercisability of rights, options and warrants ordinarily subject to such Section 2.10(i)(B)), subject to appropriate readjustment in the event of the expiration, termination or redemption of such rights prior to the exercise, deemed exercise or exchange thereof. Notwithstanding the foregoing, to the extent any such shareholder rights are exchanged by the Corporation for Common Shares, the Conversion Rate shall be appropriately readjusted as if such shareholder rights had not been issued, but the Corporation had instead issued the Common Shares issued upon such exchange as a dividend or distribution of Common Shares subject to Section 2.10(i)(A). Notwithstanding the preceding provisions of this paragraph, no adjustment shall be required to be made to the Conversion Rate with respect to any holder of Series A First Preferred Shares which is, or is an "affiliate" or "associate" of, an "acquiring person" under such shareholder rights plans or with respect to any direct or indirect transferee

of such holder who receives Series A First Preferred Shares in such transfer after the time such holder becomes, or its affiliate or associate becomes, an "acquiring person."

(F) If the Corporation, at any time or from time to time while any of the Series A First Preferred Shares is outstanding, shall issue Common Shares or any other security convertible into, exercisable or exchangeable for Common Shares (such Common Shares or other security, " **Equity Linked Securities**") (other than (1) an Excluded Issuance, (2) Common Shares issued upon conversion of the Series A First Preferred Shares and (3) rights, options, warrants or other distributions referred to in Sections 2.10(i)(B) and 2.10(i)(D)), the Conversion Rate shall be increased based on the following formula:

$$CR_1 = CR_0 \times \frac{OS_1}{OS_0 + (AC/SP)}$$

where,

CR 0 = the Conversion Rate in effect immediately prior to the issuance of such Equity Linked Securities;

CR 1 = the new Conversion Rate in effect immediately after the issuance of such Equity Linked Securities:

AC = the aggregate consideration paid or payable for such Equity Linked Securities;

OS 0 = the number of Common Shares outstanding immediately before the issuance of Equity Linked Securities;

OS 1 = the number of Common Shares outstanding immediately after the issuance of Equity Linked Securities and giving effect to any Common Shares issuable upon conversion, exercise or exchange of such Equity Linked Securities; and

SP = the Market Price on the date of issuance of such Equity Linked Securities.

The adjustment shall become effective immediately after such issuance.

(ii) <u>Calculation of Adjustments</u>. All adjustments to the Conversion Rate shall be calculated by the Corporation to the nearest 1/10,000th of one (1) Common Share (or if there is not a nearest 1/10,000th of a share, to the next lower 1/10,000th of a share). No adjustment to the Conversion Rate will be required unless such adjustment would require an increase or decrease of at least one percent (1%) of the Conversion Rate; <u>provided</u>, <u>however</u>, that any such adjustment that is not

required to be made will be carried forward and taken into account in any subsequent adjustment; <u>provided</u>, <u>further</u>, that any such adjustment of less than one percent (1%) that has not been made will be made upon any Conversion Date.

(iii) When No Adjustment Required.

- (A) Except as otherwise provided in this Section 2.10 the Conversion Rate will not be adjusted for the issuance of Common Shares or any securities convertible into or exchangeable for Common Shares or carrying the right to purchase any of the foregoing, or for the repurchase of Common Shares.
- (B) Except as otherwise provided in this Section 2.10, no adjustment of the Conversion Rate shall be made as a result of the issuance of, the distribution of separate certificates representing, the exercise or redemption of, or the termination or invalidation of, rights pursuant to any shareholder rights plans.
- (C) Notwithstanding the foregoing, no adjustment to the Conversion Rate shall be made:
 - (1) upon the issuance of any Common Shares or options or rights to purchase such shares pursuant to any present or future employee, director, manager or consultant benefit plan or program of or assumed by the Corporation or any of its Subsidiaries or of any employee agreements or arrangements or programs;
 - (2) upon the issuance of any Common Shares pursuant to any option, warrant, right, or exercisable, exchangeable or convertible security outstanding as of the Issue Date; or
 - (3) for accrued and unpaid dividends on the Series A First Preferred Shares.
- (iv) <u>Successive Adjustments</u>. After an adjustment to the Conversion Rate under this Section 2.10, any subsequent event requiring an adjustment under this Section 2.10 shall cause an adjustment to each such Conversion Rate as so adjusted.
- (v) <u>Multiple Adjustments</u>. For the avoidance of doubt, if an event occurs that would trigger an adjustment to the Conversion Rate pursuant to this Section 2.10 under more than one (1) subsection hereof (other than where holders of Series A First Preferred Shares are entitled to elect the applicable adjustment, in which case such election shall control), such event, to the extent fully taken into account in a single adjustment, shall not result in multiple adjustments hereunder; <u>provided</u>, <u>however</u>, that if more than one (1) subsection of this Section 2.10 is applicable to a single event, the subsection shall be applied that produces the largest adjustment.

- (vi) Other Adjustments. The Corporation may, but shall not be required to, make such increases in the Conversion Rate, in addition to those required by this Section 2.10, as the Board considers to be advisable in order to avoid or diminish any income tax to any holders of Common Shares resulting from any dividend or distribution of shares or issuance of rights or warrants to purchase or subscribe for shares or from any event treated as such for income tax purposes or for any other reason.
- (vii) Notice of Adjustments. Whenever the Conversion Rate is adjusted as provided under this Section 2.10, the Corporation shall as soon as reasonably practicable following the occurrence of an event that requires such adjustment (or if the Corporation is not aware of such occurrence, as soon as reasonably practicable after becoming so aware) or the date the Corporation makes an adjustment pursuant to Section 2.10(vi):
 - (A) compute the adjusted applicable Conversion Rate in accordance with this Section 2.10 and prepare and transmit to the Conversion Agent an officer's certificate setting forth the applicable Conversion Rate, the method of calculation thereof in reasonable detail, and the facts requiring such adjustment and upon which such adjustment is based; and
 - (B) provide a written notice to the holders of the Series A First Preferred Shares of the occurrence of such event and a statement in reasonable detail setting forth the method by which the adjustment to the applicable Conversion Rate was determined and setting forth the adjusted applicable Conversion Rate.
- Conversion Agent. The Conversion Agent shall not at any time be under any duty or responsibility to any holder of Series A First Preferred Shares to determine whether any facts exist that may require any adjustment of the applicable Conversion Rate or with respect to the nature or extent or calculation of any such adjustment when made, or with respect to the method employed in making the same. The Conversion Agent shall be fully authorized and protected in relying on any officer's certificate delivered pursuant to Section 2.10(vii) and any adjustment contained therein and the Conversion Agent shall not be deemed to have knowledge of any adjustment unless and until it has received such certificate. The Conversion Agent shall not be accountable with respect to the validity or value (or the kind or amount) of any Common Shares, or of any securities or property, that may at the time be issued or delivered with respect to any Series A First Preferred Shares; and the Conversion Agent makes no representation with respect thereto. The Conversion Agent, if other than the Corporation, shall not be responsible for any failure of the Corporation to issue, transfer or deliver any Common Shares pursuant to the conversion of Series A First Preferred Shares or to comply with any of the duties, responsibilities or covenants of the Corporation contained in this Section 2.10.

(ix) <u>Fractional Shares</u>. No fractional Common Shares will be delivered to the holders of Series A First Preferred Shares upon conversion. In lieu of fractional shares otherwise issuable, holders of Series A First Preferred Shares will be entitled to receive an amount in cash equal to the fraction of Common Shares, multiplied by the Closing Price of the Common Shares on the Trading Day immediately preceding the applicable Conversion Date. In order to determine whether the number of Common Shares to be delivered to a holder of Series A First Preferred Shares upon the conversion of such holder's Series A First Preferred Shares will include a fractional share (in lieu of which cash would be paid hereunder), such determination shall be based on the aggregate number of Series A First Preferred Shares of such holder that are being converted on any single Conversion Date.

(x) <u>Reorganization Events</u>. In the event of:

- (A) any reclassification, statutory exchange, merger, consolidation or other similar business combination of the Corporation with or into another Person, in each case, pursuant to which the Common Shares (but not the Series A First Preferred Shares) are changed or converted into, or exchanged for, cash, securities or other property of the Corporation or another person;
- (B) any sale, transfer, lease or conveyance to another Person of all or substantially all the property and assets of the Corporation, in each case pursuant to which the Common Shares (but not the Series A First Preferred Shares) are converted into cash, securities or other property; or
- (C) any statutory exchange of securities of the Corporation with another Person (other than in connection with a merger or acquisition) or reclassification, recapitalization or reorganization of the Common Shares (but not the Series A First Preferred Shares) into other securities,

(each of which is referred to as a "Reorganization Event") each Series A First Preferred Share outstanding immediately prior to such Reorganization Event will, without the consent of the holders of Series A First Preferred Shares unless required under Section 2.6(iii) and subject to Section 2.10(xi), remain outstanding but shall become convertible into, out of funds legally available therefor, the number, kind and amount of securities, cash and other property (the "Exchange Property") (without any interest on such Exchange Property and without any right to dividends or distribution on such Exchange Property which have a record date that is prior to the applicable Conversion Date) that the holder of such Series A First Preferred Share would have received in such Reorganization Event had such holder converted its Series A First Preferred Shares into the applicable number of Common Shares immediately prior to the effective date of the Reorganization Event, assuming that such holder is not a Person with which the Corporation consolidated or into which the Corporation merged or which merged into the Corporation or to which such sale or transfer was made, as the case may be (any such Person, a "Constituent Person"), or an Affiliate of a Constituent

Person to the extent such Reorganization Event provides for different treatment of Common Shares held by Affiliates of a Constituent Person and non-Affiliates; <u>provided</u>, that if the kind or amount of securities, cash and other property receivable upon such Reorganization Event is not the same for each Common Share held immediately prior to such Reorganization Event by a Person other than a Constituent Person or an Affiliate thereof, then for the purpose of this Section 2.10(x), the kind and amount of securities, cash and other property receivable upon such Reorganization Event will be deemed to be the weighted average of the types and amounts of consideration received by the holders of Common Shares.

- Exchange Property Election. In the event that the holders of Common Shares have the opportunity to elect the form of consideration to be received in such Reorganization Event, the Exchange Property that the holders of Series A First Preferred Shares shall be entitled to receive shall be determined by the holders of a majority of the outstanding Series A First Preferred Shares on or before the earlier of (A) the deadline for elections by holders of Common Shares and (B) two (2) Business Days before the anticipated effective date of such Reorganization Event. The number of units of Exchange Property for each Series A First Preferred Share converted following the effective date of such Reorganization Event shall be determined from among the choices made available to the holders of the Common Shares and based on the per share amount as of the effective date of the Reorganization Event, determined as if the references to "Common Shares" herein were to "units of Exchange Property."
- (xii) <u>Successive Reorganization Events</u>. The above provisions of Section 2.10(x) and Section 2.10(xi) shall similarly apply to successive Reorganization Events and the provisions of this Section 2.10 shall apply to any shares in the capital (or in the capital of any other issuer) received by the holders of the Common Shares in any such Reorganization Event.
- (xiii) Reorganization Event Notice. The Corporation (or any successor) shall, no less than twenty (20) Business Days prior to the occurrence of any Reorganization Event, provide written notice to the holders of Series A First Preferred Shares of such occurrence of such event and of the kind and amount of the cash, securities or other property that constitutes the Exchange Property. Failure to deliver such notice shall not affect the operation of this Section 2.10.
- (xiv) Reorganization Event Agreements. The Corporation shall not enter into any agreement for a transaction constituting a Reorganization Event unless (A) such agreement provides for or does not interfere with or prevent (as applicable) conversion of the Series A First Preferred Shares into the Exchange Property in a manner that is consistent with and gives effect to this Section 2.10, and (B) to the extent that the Corporation is not the surviving corporation in such Reorganization Event or will be dissolved in connection with such Reorganization Event, proper provision shall be made in the agreements governing such Reorganization Event for the conversion of the Series A First Preferred Shares into shares of the Person surviving such Reorganization Event or such other

- continuing entity in such Reorganization Event, or in the case of a Reorganization Event described in Section 2.10(x)(B), an exchange of Series A First Preferred Shares for the shares of the Person to whom the Corporation's assets are conveyed or transferred, having voting powers, preferences, and relative, participating, optional or other special rights as nearly equal as possible to those provided herein.
- Restriction on Adjustments and Transactions. Notwithstanding the foregoing, if any of the transactions described in this Section 2.10 would result in an adjustment to the Conversion Rate such that the Series A First Preferred Shares would be convertible in the aggregate into more than 19.90% of the Common Shares outstanding on the Issue Date (subject to adjustment as described in this Section 2.10), then the Company shall not effect such transaction without first obtaining the written consent or affirmative vote at a meeting called for that purpose by holders of Common Shares required by a securities exchange on which the Common Shares are listed or admitted to trading in order to permit the Series A First Preferred Shares to be convertible into such number of Common Shares. Notwithstanding anything herein to the contrary, any right of the Series A First Preferred Shares in these Articles of Amalgamation may be waived by the written consent or affirmative vote at a meeting called for that purpose by holders of at least two-thirds (2/3) of the then outstanding shares of Series A First Preferred Shares, voting separately as a series.
- 2.11 <u>Reservation of Shares</u>. The Corporation shall at all times when the Series A First Preferred Shares shall be outstanding reserve and keep available, free from preemptive rights, for issuance upon the conversion of Series A First Preferred Shares, such number of its authorized but unissued Common Shares as will from time to time be sufficient to permit the conversion of all outstanding Series A First Preferred Shares. Prior to the delivery of any securities which the Corporation shall be obligated to deliver upon conversion of the Series A First Preferred Shares, the Corporation shall comply with all applicable laws and regulations which require action to be taken by the Corporation.
- Notices. Any and all notices or other communications or deliveries hereunder shall be in writing and shall be deemed given and effective on the earliest of (i) the date of transmission, if such notice or communication is delivered via facsimile at the facsimile number specified in this Section 2.12 prior to or at the Close of Business on a Business Day and electronic confirmation of receipt is received by the sender, (ii) the next Business Day after the date of transmission, if such notice or communication is delivered via facsimile at the facsimile number specified in this Section on a day that is not a Business Day or later than the Close of Business on any Business Day, (iii) the Business Day following the date of mailing, if sent by a nationally recognized overnight courier service, or (iv) upon actual receipt by the party to whom such notice is required to be given. The addresses for such communications shall be: (i) if to the Corporation, attention: Chief Executive Officer and General Counsel, or (ii) if to a holder of Series A First Preferred Shares, to the address or facsimile number appearing on the Corporation's shareholder records or such other address or facsimile number as such holder may provide to the Corporation in accordance with this Section 2.12. Notwithstanding the foregoing, if the Series A First Preferred Shares are issued in book- entry form through The Depository Trust Corporation or any similar

facility, such notices may be given to the holders of the Series A First Preferred Shares in any manner permitted by such facility.

- 2.13 <u>Certain Definitions</u>. As used herein, the following terms shall have the following meanings, unless the context otherwise requires:
- "Affiliate" with respect to any person shall have the meaning specified pursuant to part 1.3 of National Instrument 46-106 *Prospectus and Registrations Exemptions*, promulgated under the *Securities Act* (Ontario); provided, however, that (i) portfolio companies in which any person or any of its Affiliates has an investment shall not be deemed an Affiliate of such person, and (ii) the Corporation, any of its Subsidiaries, or any of the Corporation's other controlled Affiliates, in each case, will not be deemed to be Affiliates of Crestview for purposes of the rights, privileges, restrictions and conditions of the Series A First Preferred Shares. For purposes of this definition, "control" (including, with correlative meaning, the terms "controlled by" and "under common control with") when used in respect of any Person shall have the meaning specified pursuant to part 1.4 of National Instrument 46-106 Prospectus and Registrations Exemptions, promulgated under the Securities Act (Ontario).
- "Beneficially Own" shall mean to have "beneficial ownership" as such term is defined in the Canada Business Corporations Act and shall include, without limitation, any Person that "beneficially owns" securities for the purposes of Multilateral Instrument 61-101 Protection of Minority Security Holders in Special Transactions, promulgated under the Securities Act (Ontario).
 - "Board" shall mean the Board of Directors of the Corporation.
- "Business Day" shall mean a day that is a Monday, Tuesday, Wednesday, Thursday or Friday and is not a day on which banking institutions in New York, New York or Toronto, Ontario generally are authorized or obligated by law, regulation or executive order to close.
 - "Change of Control" shall mean the occurrence of any of the following:
 - (i) any Person (other than Crestview) shall Beneficially Own, directly or indirectly, through a purchase, merger or other acquisition transaction or series of transactions, shares of the Corporation's Equity Securities entitling such Person to exercise fifty percent (50%) or more of the total voting power of all classes of Voting Shares of the Corporation, which shall not include the Series A First Preferred Shares, other than an acquisition by the Corporation, any of the Corporation's Subsidiaries or any of the Corporation's employee benefit plans (for purposes of this clause (i), "Person" shall include, without limitation, any syndicate or group);
 - (ii) the Corporation (A) merges or consolidates with or into any other Person, another Person merges with or into the Corporation, or the Corporation conveys, sells, transfers or leases all or substantially all of the Corporation's assets to another Person or (B) engages in any recapitalization, reclassification or other transaction in which all or substantially all of the Common Shares are exchanged for or

converted into cash, securities or other property, in each case other than a merger or consolidation:

- (1) that does not result in a reclassification, conversion, exchange or cancellation of the Corporation's outstanding Common Shares; or
- which is effected solely to change the Corporation's jurisdiction of incorporation and results in a reclassification, conversion or exchange of outstanding Common Shares solely into common shares of the surviving entity; or
- (3) where the Voting Shares outstanding immediately prior to such transaction are converted into or exchanged for Voting Shares of the surviving or transferee Person constituting a majority of the outstanding Voting Shares of such surviving or transferee Person (immediately after giving effect to such issuance) and no Person has the right to exercise fifty percent (50%) or more of the total voting power of all classes of Voting Shares of the Corporation which, for the avoidance of doubt, shall not include the Series A First Preferred Shares; or
- (iii) the Common Shares cease to be listed or quoted on any of the New York Stock Exchange, the Nasdaq Global Select Market, Nasdaq Global Market or the Toronto Stock Exchange (or any of their respective successors); or
- (iv) any transaction or series of transactions having substantially the same effect as any of the foregoing.
- "Close of Business" shall mean 5:00 p.m., New York City time, on any Business Day.
- "Closing Price" shall mean the price per share of the final trade of the Common Shares on the applicable Trading Day on the principal United States securities exchange on which the Common Shares are listed or admitted to trading.
 - "Code" shall have the meaning ascribed to it in Section 2.8.
 - "Common Shares" shall have the meaning ascribed to it in Section 2.3.
 - "Constituent Person" shall have the meaning ascribe to it in Section 2.10(x).
 - "Conversion Agent" shall have the meaning ascribed to it in Section 2.17.
 - "Conversion Date" shall have the meaning ascribed to it in Section 2.7(iii).
 - "Conversion Rate" shall have the meaning ascribed to it in Section 2.7(ii).
 - "Corporation" shall mean Cott Corporation.

- "Credit Agreement" shall mean the Credit Agreement, dated as of August 17, 2010, among the Corporation, Cott Beverages Inc., a Georgia corporation, Cliffstar LLC, a Delaware limited liability company, and Cott Beverages Limited, a company organized under the laws of England and Wales, as Borrowers, the other Loan Parties party thereto, the Lenders party thereto, JPMorgan Chase Bank, N.A., London Branch, as UK Security Trustee, JPMorgan Chase Bank, N.A., as Administrative Agent and Administrative Collateral Agent, and General Electric Capital Corporation, as Co- Collateral Agent (as amended by that certain Amendment No. 1 to Credit Agreement, dated as of April 19, 2012, as further amended by that certain Amendment No. 2 to Credit Agreement, dated as of July 19, 2012, as further amended by that certain Amendment No. 3 to Credit Agreement, dated as of October 22, 2013, as further amended by that certain Amendment No. 4 to Credit Agreement, dated as of May 28, 2014, and as amended on the Issue Date pursuant to that certain Amendment No. 5 but for the avoidance of doubt, without giving effect to any subsequent amendment, restatement, supplement or modification thereof).
- "Crestview" shall mean Crestview DSW Investors, L.P. and any of its Affiliates, any successor entity and any other investment fund, vehicle or similar entity of which such person or an Affiliate, advisor or manager of such person serves as the general partner, manager or advisor.
 - "Distributed Property" shall have the meaning ascribed to it in Section 2.10(i)(D).
 - "Dividend" shall have the meaning ascribed to it in Section 2.4(i).
 - "Dividend Payment Date" shall have the meaning ascribed to it in Section 2.4(ii).
 - "Dividend Period" shall have the meaning ascribed to it in Section 2.4(ii).
 - " Dividend Rate" shall mean, in each case:

From the Issue Date until the first anniversary of the Issue Date,	9.0% per annum of the Redemption Value of each Series A First Preferred Share.
From the day after the first anniversary of the Issue Date,	10.0% per annum of the Redemption Value of each Series A First Preferred Share.
From the day after the second anniversary of the Issue Date,	11.0% per annum of the Redemption Value of each Series A First Preferred Share.
From the day after the third anniversary of the Issue Date,	12.0% per annum of the Redemption Value of each Series A First Preferred Share.
From the day after the fourth anniversary of the Issue Date,	13.0% per annum of the Redemption Value of each Series A First Preferred Share.
From the day after the fifth anniversary of the Issue Date,	14.0% per annum of the Redemption Value of each Series A First Preferred Share.

[&]quot;EBITDA" shall have the meaning ascribed to it in the Credit Agreement.

- "Equity Linked Securities" shall have the meaning ascribed to it in Section 2.10(i)(F).
- "Equity Securities" shall mean any and all shares, interests, rights to purchase, warrants, options, participations or other equivalents of or interests in (however designated) equity securities issued by the Corporation.
 - "Exchange Property" shall have the meaning ascribed to it in Section 2.10(x).
- "Excluded Issuance" shall mean, any issuances of (i) Equity Securities or options to purchase Equity Securities to employees, directors, managers, officers or consultants of or to the Corporation or any of its Subsidiaries pursuant to an option or incentive compensation or similar plan outstanding as of the date hereof or, subsequent to the date hereof, approved by the Board or a duly authorized committee of the Board, (ii) securities as consideration for any bona fide merger, joint venture, partnership, consolidation, dissolution, liquidation, tender offer, recapitalization, reorganization, share exchange, business combination or similar transaction or any other direct or indirect acquisition by the Corporation, whereby the Corporation's securities comprise, in whole or in part, the consideration paid by the Corporation in such transaction, (iii) Common Shares issued at a price to the public (i.e., without giving effect to any underwriter or placement agent discounts, commissions or fees or issuer costs, whether or not such issuance is actually a public offering) equal to or greater than the Market Price on the Trading Day immediately preceding the earlier of (x) the date on which the price for such sale or issuance is agreed or fixed, and Equity Linked Securities issued with an exercise or conversion price equal to or greater than the Market Price on the Trading Day immediately preceding the earlier of (x) the date on which the sale or issuance is publicly announced and (y) the date on which the price for such sale or issuance is agreed or fixed.
 - "Expiration Date" shall have the meaning ascribed to it in Section 2.10(i)(C).
 - "Expiration Time" shall have the meaning ascribed to it in Section 2.10(i)(C).
 - "GAAP" shall mean generally accepted accounting principles in the United States.
 - "Indebtedness" shall have the meaning ascribed to it in the Credit Agreement.
 - "Issue Date" shall mean December 12, 2014.
 - "Junior Shares" shall have the meaning ascribed to it in Section 2.3.
 - "Liquidation" shall have the meaning ascribed to it in Section 2.5(i).
 - "Liquidation Preference" shall have the meaning ascribed to it in Section 2.5(i).
- "Market Price" shall mean, as of any Trading Day, the lower of (i) the Closing Price and (ii) the trailing fifteen (15) day VWAP of the Common Shares of the Corporation.
 - "Open of Business" shall mean 9:00 a.m., New York City time, on any Business Day.

- "Person" shall mean any individual, company, partnership, limited liability company, joint venture, association, joint stock company, trust, unincorporated organization, government or agency or political subdivision thereof or any other entity.
 - "Preferred Dividend Default" shall have the meaning ascribed to it in Section 2.6(iv).
- "Preferred Shares" shall mean any and all series of preferred shares of the Corporation, including the Series A First Preferred Shares and the Series B First Preferred Shares.
 - "Preferred Share Director" shall have the meaning ascribed to it in Section 2.6(iv).
 - "Purchased Shares" shall have the meaning ascribed to it in Section 2.10(i)(C).
- "Record Date" shall mean, with respect to any dividend, distribution or other transaction or event in which the holders of Common Shares have the right to receive any cash, securities or other property or in which the Common Shares (or other applicable security) are exchanged for or converted into any combination of cash, securities or other property, the date fixed for determination of shareholders entitled to receive such cash, securities or other property (whether such date is fixed by the Board or by statute, contract, the constating documents of the Corporation or otherwise and, with respect to the right of a shareholder to vote on any matter at a meeting of shareholders, the date fixed by the Corporation as the record date for the determination of shareholders entitled to vote at the meeting or, if no such record date is established, the date such vote is taken or any written resolution of shareholders is established).
- "Redemption Value" shall initially mean US \$1,000.00 per share of Series A First Preferred Shares, as adjusted pursuant to Section 2.4(iii).
 - "Reorganization Event" shall have the meaning ascribed to it in Section 2.10(x).
- "Restricted Cash" shall mean (i) any cash or cash equivalents identified on the balance sheet of the Corporation and its Subsidiaries as "restricted" and (ii) any amount to the extent that the distribution thereof to the Corporation or use thereof by the Corporation for the repayment of its consolidated Indebtedness is restricted or prohibited by law or contract.
 - "Series A First Preferred Shares" shall have the meaning ascribed to it in Paragraph 3(b).
 - "Series B First Preferred Shares" shall have the meaning ascribed to it in Paragraph 3(c).
- "Significant Subsidiary" shall mean any Subsidiary of the Corporation that would be a "Significant Subsidiary" of the Corporation within the meaning of Rule 1-02 under Regulation S-X promulgated by the Securities and Exchange Commission, determined as of the date of the latest audited consolidated financial statements of the Corporation and its consolidated Subsidiaries.
- "Subsidiary" shall mean any company or corporate entity for which the Corporation owns, directly or indirectly, an amount of the voting securities, other voting rights or voting

partnership interests sufficient to elect at least a majority of its board of directors or other governing body (or, if there are no such voting interests, more than fifty percent (50%) of the equity interests of such company or corporate entity).

- "Trading Day" shall mean any Business Day on which the Common Shares are traded, or able to be traded, on the principal United States securities exchange on which the Common Shares are listed or admitted to trading.
 - "Trigger Event" shall have the meaning ascribed to it in Section 2.10(i)(E).
- "Voting Shares" shall mean Equity Securities of the class or classes pursuant to which the holders thereof have the general voting power under ordinary circumstances (determined without regard to any classification of directors) to elect one or more members of the Board of the Corporation (without regard to whether or not, at the relevant time, Equity Securities of any other class or classes (other than Common Shares) shall have or might have voting power by reason of the happening of any contingency).
- "VWAP" shall mean the volume-weighted average prices for a Common Share or similar equity interest distributed to holders of Common Shares on the principal United States securities exchange on which such Common Share or equity interest trades, as reported by Bloomberg, L.P. (or, if Bloomberg ceases to publish such price, any successor service reasonably chosen by the Corporation) in respect of the period from the open of trading on the relevant Trading Day until the close of trading on such Trading Day (or if such volume-weighted average price is unavailable, the market price of one Common Share or equity interest on such Trading Day determined, using a volume- weighted average method by a nationally recognized investment banking firm (unaffiliated with the Corporation) retained for such purpose by the Corporation).
- 2.14 <u>Headings</u>. The headings of the paragraphs hereof are for convenience of reference only and shall not define, limit or affect any of the provisions hereof.
- 2.15 <u>Record Holders</u>. To the fullest extent permitted by applicable law, the Corporation may deem and treat the record holder of any Series A First Preferred Share as the true and lawful owner thereof for all purposes, and the Corporation shall not be affected by any notice to the contrary.
- 2.16 <u>Replacement Certificates</u>. The Corporation shall replace any mutilated certificate at the holder's expense upon surrender of that certificate to the Corporation. The Corporation shall replace certificates that become destroyed, stolen or lost at the holder's expense upon delivery to the Corporation of reasonably satisfactory evidence that the certificate has been destroyed, stolen or lost, together with any indemnity that may be required by the Corporation.
- 2.17 <u>Transfer Agent, Conversion Agent, Registrar and Paying Agent</u>. The duly appointed transfer agent, conversion agent (the "Conversion Agent"), registrar and paying agent for the Series A First Preferred Shares shall initially be the Corporation. The Corporation may, in its sole discretion, remove the transfer agent in accordance with the agreement between the Corporation and the transfer agent; <u>provided</u>, that the Corporation shall appoint a successor transfer agent who shall accept such appointment prior to the effectiveness of such removal.

Upon any such removal or appointment, the Corporation shall send notice thereof by first-class mail, postage prepaid, to the holders of the Series A First Preferred Shares.

3.0 SERIES B NON-CONVERTIBLE FIRST PREFERRED SHARES

- 3.1 <u>Authorized</u>. The Corporation shall be authorized to issue an unlimited number of Series B First Preferred Shares. Series B First Preferred Shares redeemed, purchased, or otherwise acquired by the Corporation shall be cancelled and not restored to the status of authorized but unissued shares of the series. The number of authorized Series B First Preferred Shares may be increased or decreased by the approval of the holders of a majority of the outstanding Series B First Preferred Shares, voting as a separate series, provided that there shall be no decrease in the number of authorized Series B First Preferred Shares to a number less than the number of such series then issued and outstanding.
- 3.2 <u>Currency</u>. All payments and distributions on Series B First Preferred Shares or with respect thereto shall be made in United States currency. All references herein to "\$" or "dollars" refer to United States currency.

3.3 Ranking.

- (i) The Series B First Preferred Shares shall, with respect to the payment of dividends and upon Liquidation rank senior to each other class or series of shares of the Corporation (other than the Series A First Preferred Shares), whether now outstanding or hereafter issued, including, without limitation, the common shares of the Corporation (the "Common Shares"), the Second Preferred Shares and the Second Preferred Shares Series 1 (all such junior shares being referred to hereinafter collectively as "Junior Shares").
- (ii) The Series B First Preferred Shares shall, with respect to the payment of dividends and rights upon Liquidation rank equally with the Series A First Preferred Shares of the Corporation, whether now outstanding or hereafter issued.

3.4 <u>Dividends</u>.

- (i) The holders of Series B First Preferred Shares shall be entitled to receive a quarterly fixed cumulative dividend in an amount equal to the applicable Dividend Rate (as defined herein) then in effect (the "**Dividends**").
- (ii) Subject to applicable law, Dividends shall be declared and payable quarterly in arrears on January 1, April 1, July 1 and October 1 of each year (unless any such day is not a Business Day, in which event such Dividends shall be payable on the next succeeding Business Day, without accrual to the actual payment date), commencing on January 1, 2015 (each such payment date being a " **Dividend Payment Date**," and the period from the Issue Date of the Series B First Preferred Shares to the first Dividend Payment Date and each such quarterly period thereafter being a " **Dividend Period**"). The amount of Dividends payable on the Series B First Preferred Shares for any period shall be computed on the basis of a 360-day year and the actual number of days elapsed. Subject to Section

- 3.6(ii)(G)(2) herein, such Dividends shall be paid to holders of Series B First Preferred Shares before any dividends shall be declared, set apart for or paid upon any Junior Shares.
- (iii) Dividends, whether or not declared, shall begin to accrue and be cumulative from the Issue Date and shall compound at the relevant rate on each subsequent Dividend Payment Date (i.e., no Dividends shall accrue on another Dividend unless and until any Dividend Payment Date for such other Dividends has passed without such other Dividends having been paid on such date, in which case Dividends will accrue on such unpaid Dividends). If the Corporation does not pay any Dividend in full in cash on any scheduled Dividend Payment Date, such Dividends will accrue at the then applicable Dividend Rate plus an additional two and one half percent (2.5%) per annum of the Redemption Value of each Series B First Preferred Share from such scheduled Dividend Payment Date to the date that all accumulated Dividends on the Series B First Preferred Shares have been paid in cash in full. Any Dividends accrued and not paid in cash pursuant to this Section 3.4(iii) on each Dividend Payment Date shall be added to the Redemption Value until such Dividends are paid in cash in full. For the avoidance of doubt, Dividends shall accumulate whether or not in any Dividend Period there have been funds of the Corporation legally available for the payment of such Dividends.
- (iv) Except as otherwise provided herein, if at any time the Corporation pays less than the total amount of Dividends then accumulated in cash, such cash dividend payment shall be distributed pro rata among the holders of Series B First Preferred Shares based upon the Redemption Value on the Series B First Preferred Shares then held by each such holder. When Dividends are not paid in full in cash, all Dividends paid in cash and any dividends paid in cash on Series A First Preferred Shares shall be paid pro rata so that the amount of Dividends paid in cash and any dividends on Series A First Preferred Shares paid in cash shall in all cases bear to each other the same ratio as accumulated Dividends and accumulated dividends on Series A First Preferred Shares bear to each other.
- (v) All Dividends shall be paid in cash or accrued in accordance with Section 3.4(iii).
- (vi) Dividends shall be payable to the holders of record of Series B First Preferred Shares as they appear on the register of the Corporation at the Close of Business on the relevant record date, which shall be not more than thirty (30) days nor less than ten (10) days preceding the applicable Dividend Payment Date.

3.5 <u>Liquidation, Dissolution or Winding Up</u>.

(i) Upon any voluntary or involuntary liquidation, dissolution or winding up of the Corporation or any return of capital or other distribution of assets of the Corporation among its shareholders for the purpose of winding up its affairs (each, a " **Liquidation**"), the holders of Series B First Preferred Shares shall be entitled to receive, out of the assets of the Corporation or proceeds thereof

(whether capital or surplus) legally available therefor, an amount per Series B First Preferred Share equal to the Redemption Value per share, plus an amount equal to any Dividends accumulated but unpaid thereon (whether or not declared) through the date of Liquidation (the "Liquidation Preference"). Holders of Series B First Preferred Shares will not be entitled to any other amounts from the Corporation after they have received the full amounts provided for in this Section 3.5(i) and will have no right or claim to any of the Corporation's remaining assets.

- (ii) If, in connection with any distribution described in Section 3.5(i) above, the assets of the Corporation or proceeds thereof are not sufficient to pay in full the Liquidation Preference payable on the Series B First Preferred Shares and the corresponding amounts payable on the Series A First Preferred Shares, then such assets, or the proceeds thereof, shall be paid pro rata in accordance with the full respective amounts which would be payable on such shares if all amounts payable thereon were paid in full.
- (iii) For purposes of this Section 3.5, the merger or consolidation of the Corporation with or into any other corporation or other entity, or the sale, conveyance, lease or other disposition of all or substantially all of the assets of the Corporation, shall not constitute a Liquidation of the Corporation.

3.6 <u>Voting Rights; Consent Rights</u>.

- (i) The holders of Series B First Preferred Shares shall be entitled to notice of all meetings at which holders of Common Shares shall be entitled to vote; provided, however that, notwithstanding any such notice but subject to Section 3.6(ii) and without limiting any class voting rights pursuant to applicable law, the holders of Series B First Preferred Shares shall not be entitled to vote on any matter presented to the holders of Common Shares of the Corporation for their action or consideration.
- (ii) In addition to, and not in substitution for, the rights and privileges afforded to the holders of a class of issued shares under applicable law, for so long as any Series B First Preferred Shares remain outstanding, the Corporation shall not and shall not permit any direct or indirect Subsidiary of the Corporation to, without first obtaining the written consent or affirmative vote at a meeting called for that purpose by holders of at least two-thirds (2/3) of the then outstanding shares of Series B First Preferred Shares voting separately as a series, take any of the following actions:
 - (A) Any change, amendment, alteration or repeal (including, without limitation, as a result of a merger, consolidation, or other similar or extraordinary transaction) of any provisions of the Corporation's Articles of Amalgamation or By-laws that amends, modifies or adversely affects the rights, preferences, privileges, conditions or voting powers of the Series B First Preferred Shares;

- (B) Any issuance or reclassification of shares of the Corporation (including, without limitation, as a result of a merger, consolidation, or other similar or extraordinary transaction) that would rank equal or senior to the Series B First Preferred Shares, with respect to the Liquidation of the Corporation or with respect to the declaration or payment of dividends, including, for the avoidance of doubt, the issuance of any Series B First Preferred Shares or Series A First Preferred Shares;
- (C) Any voluntary initiation of any Liquidation of the Corporation or any of its Significant Subsidiaries or any group of Subsidiaries that, taken together, would constitute a Significant Subsidiary or commencement of a proceeding for bankruptcy, insolvency, receivership or similar action with respect to the Corporation or any of its Significant Subsidiaries or any group of Subsidiaries that, taken together, would constitute a Significant Subsidiary;
- (D) Any material change to the tax or accounting policies of the Corporation, except any changes required by applicable law or pursuant to GAAP;
- (E) Except where holders of another series of First Preferred Shares of the Corporation have the right to appoint directors to the Board pursuant to these Articles of Amalgamation or contractual rights of the initial holders of Series A First Preferred Shares to appoint directors to the Board after conversion of the Series A First Preferred Shares to Common Shares, any increase in the number of members of the Board if such increase would result in the Board having more than eleven (11) members;
- (F) Any incurrence, assumption or refinancing of any Indebtedness of the Corporation or any of its Subsidiaries to the extent that, immediately after giving effect to such incurrence, assumption or refinancing, the ratio of (1) the consolidated Indebtedness of the Corporation and its Subsidiaries that would be stated on the balance sheet of the Corporation and its Subsidiaries as of the date of determination (determined on a consolidated basis in accordance with GAAP) minus the lesser of (i) the amount of cash and cash equivalents (except for Restricted Cash) that would be stated on the balance sheet of the Corporation and its Subsidiaries and held by the Corporation and its Subsidiaries as of the date of determination (determined on a consolidated basis in accordance with GAAP) and (ii) \$75,000,000 to (2) the EBITDA of the Corporation and its Subsidiaries for the most recently ended four fiscal quarters for which internal financial statements are available immediately preceding the date on which such additional Indebtedness is incurred, assumed or refinanced (the "Reference **Period** "), would not be less than a ratio of 5.0-to-1.0 on a pro forma basis (calculated in accordance with GAAP and Regulation S-X promulgated by the United States Securities and Exchange Commission) after giving effect to such incurrence, assumption or refinancing and the use of the proceeds therefrom as if such events had occurred on the first

day of the Reference Period, provided, however, in the case of any revolving credit facility, Indebtedness shall be deemed to be incurred on the date that the commitments are entered into and not on the date of subsequent borrowings and shall be deemed to be outstanding continuously until such revolving credit facility is terminated and, for this purpose, the amount of the commitment under the Credit Agreement shall be deemed to be equal to \$328,000,000.

- (G) Any declaration or payment of any dividend or distribution (or any money made available for a sinking fund for the payment of any dividend or distribution) upon any Junior Shares, or the redemption of any Junior Shares or any Series A First Preferred Shares, or the purchase for cancellation, payment or acquisition for any consideration (or any money made available for a sinking fund for the redemption) of any Junior Shares or any Series A First Preferred Shares by the Corporation, directly or indirectly; provided, however, that the foregoing limitation shall not apply to:
 - (1) purchases, redemptions or other acquisitions of Junior Shares in connection with any employment contract, benefit plan or other similar arrangement with or for the benefit of any one or more employees, officers, directors, managers or consultants of or to the Corporation or any of its Subsidiaries; or
 - regular quarterly dividends to holders of Common Shares in an amount not to exceed US\$0.06 per share; provided, however, that no dividends shall be declared or paid without a vote of the holders of Series B First Preferred Shares pursuant to this Section 3.6(ii) if one or more quarterly Dividends have not been paid in full in cash on any Dividend Payment Date until such time as all such Dividends have been paid in full in cash; or
- (H) Any issuance of Equity Securities by any Subsidiary of the Corporation to any Person other than the Corporation or a direct or indirect wholly-owned Subsidiary of the Corporation, any acquisition of Equity Securities of any Person other than a wholly owned Subsidiary of the Corporation or any of its direct or indirect wholly owned Subsidiaries, or any transfer or other transaction (including, without limitation, as a result of a merger, consolidation, or other similar or extraordinary transaction) resulting in the Corporation or any of its Subsidiaries owning Equity Securities of any Person other than a wholly- owned Subsidiary of the Corporation or any of its direct or indirect wholly-owned Subsidiaries.

3.7 <u>Taxes</u>.

(i) The Corporation shall pay any and all documentary, stamp and similar issue or transfer tax due on the issue of the Series B First Preferred Shares.

(ii) Each holder of Series B First Preferred Shares and the Corporation agree to cooperate with each other in connection with any redemption of part of Series B First Preferred Shares and to use good faith efforts to structure such redemption so that such redemption may be treated as a sale or exchange pursuant to Section 302 of the Internal Revenue Code of 1986, as amended; provided, that nothing in this Section 3.7 shall require the Corporation to purchase any Series B First Preferred Shares; and provided, further, that the Corporation makes no representation or warranty in this Section 3.7 regarding the tax treatment of any redemption of Series B First Preferred Shares.

3.8 <u>Redemption and Repurchase</u>.

- Holder Redemption Election. At any time after the date that is the nine (9) year anniversary of the Issue Date, each holder of outstanding Series B First Preferred Shares shall have the right, but not the obligation, to elect to cause the Corporation to redeem any or all of such holder's outstanding Series B First Preferred Shares by delivery to the Corporation of written notice thereof (the "Redemption Election"). Upon receipt of a Redemption Election from a holder of Series B First Preferred Shares, the Corporation shall redeem the number of outstanding Series B First Preferred Shares such holder has elected to redeem pursuant to such Redemption Election at a redemption price per share equal to (A) the sum of the Redemption Value per share of the Series B First Preferred Shares to be redeemed plus (B) an amount equal to any Dividends accumulated but unpaid thereon (whether or not declared) to but excluding the date of redemption.
- (ii) <u>Corporation Redemption Election</u>. The Series B First Preferred Shares may be redeemed, in whole or in part, at any time, at the option of the Corporation, upon giving notice of redemption pursuant to Section 3.8(iv), at a redemption price per share equal to (A) the sum of the Redemption Value per share of the Series B First Preferred Shares to be redeemed plus (B) an amount equal to any Dividends accumulated but unpaid thereon (whether or not declared) to but excluding the date of redemption.
- (iii) Repurchase at the Option of the Holder Upon a Change of Control. Upon the occurrence of a Change of Control, each holder of Series B First Preferred Shares shall have the right to require the Corporation to repurchase, by irrevocable written notice to the Corporation, all or any portion of such holder's Series B First Preferred Shares at a purchase price per share equal to the sum of (A) the Redemption Value per share of the Series B First Preferred Shares plus (B) an amount equal to accrued but unpaid Dividends (whether or not declared) on such Series B First Preferred Shares to but excluding the date of repurchase.

No earlier than sixty (60) days and no later than thirty (30) days prior to the occurrence of a Change of Control, the Corporation shall send notice by first- class mail, postage prepaid, addressed to the holders of record of the Series B First Preferred Shares at their respective last addresses appearing on the books of the Corporation stating (A) that a Change of Control is expected to occur, (B) the

date that the Change of Control is expected to occur, (C) that all Series B First Preferred Shares tendered prior to a Business Day no earlier than thirty (30) days nor later than sixty (60) days from the date such notice is mailed shall be accepted for repurchase and (D) the procedures that holders of the Series B First Preferred Shares must follow in order for their Series B First Preferred Shares to be repurchased, including the place or places where certificates for such shares are to be surrendered for payment of the repurchase price. Any notice mailed as provided in this Subsection shall be conclusively presumed to have been duly given, whether or not the holder receives such notice, but failure duly to give such notice by mail, or any defect in such notice or in the mailing thereof, to any holder of Series B First Preferred Shares designated for repurchase shall not affect the validity of the proceedings for the repurchase of any other Series B First Preferred Shares. The repurchase of Series B First Preferred Shares pursuant to this Section 3.8(iii) shall occur simultaneously with the consummation of the Change of Control.

- Notice of Redemption at the Option of the Corporation. Notice of every redemption of Series B First Preferred Shares pursuant to Section 3.8(ii) shall be given by first-class mail, postage prepaid, addressed to the holders of record of the shares to be redeemed at their respective last addresses appearing on the books of the Corporation. Such mailing shall be at least thirty (30) days and not more than sixty (60) days before the date fixed for redemption. Any notice mailed as provided in this Section 3.8(iv) shall be conclusively presumed to have been duly given, whether or not the holder receives such notice, but failure duly to give such notice by mail, or any defect in such notice or in the mailing thereof, to any holder of Series B First Preferred Shares designated for redemption shall not affect the validity of the proceedings for the redemption of any other Series B First Preferred Shares. Each notice of redemption given to a holder shall state: (A) the redemption date; (B) the number of the Series B First Preferred Shares to be redeemed and, if less than all the shares held by such holder are to be redeemed, the number of such shares to be redeemed from such holder; (C) the redemption price; and (D) the place or places where certificates for such shares are to be surrendered for payment of the redemption price in respect of such shares.
- (v) Partial Redemption. In case of any redemption of part of the Series B First Preferred Shares at the time outstanding pursuant to this Section 3.8, the shares to be redeemed shall be redeemed pro rata. Subject to the provisions hereof, the Corporation shall have full power and authority to prescribe the terms and conditions upon which the Series B First Preferred Shares shall be redeemed from time to time. If fewer than all the shares represented by any certificate are redeemed, a new certificate shall be issued representing the unredeemed shares without charge to the holder thereof. Without limiting any remedies that may be available to any holder of Series B First Preferred Shares, at law, in equity or otherwise, if on any date of redemption, the Corporation does not have sufficient funds legally available to redeem all of the outstanding Series B First Preferred Shares to be redeemed on such date in accordance herewith, the Corporation shall redeem only the number of such Series B First Preferred Shares that it has

sufficient funds to redeem with such amount to be apportioned on a pro rata basis among the holders of such Series B First Preferred Shares, and the Corporation shall redeem such Series B First Preferred Shares that are not redeemed on such redemption date as soon as practicable after the Corporation has funds legally available therefor.

- (vi) Effectiveness of Redemption. If notice of redemption has been duly given and if on or before the redemption date specified in the notice all funds necessary for the redemption have been deposited by the Corporation, in trust for the pro rata benefit of the holders of the shares called for redemption, with a bank or trust company doing business in the Borough of Manhattan, The City of New York, and having a capital and surplus of at least US \$1 billion and selected by the Board, so as to be and continue to be available solely therefor, then, notwithstanding that any certificate for any share so called for redemption has not been surrendered for cancellation, on and after the redemption date dividends shall cease to accrue on all shares so called for redemption, all shares so called for redemption shall no longer be deemed outstanding and all rights with respect to such shares shall forthwith on such redemption date cease and terminate, except only the right of the holders thereof to receive the amount payable on such redemption from such bank or trust company, without interest. Any funds unclaimed at the end of three (3) years from the redemption date shall, to the extent permitted by law, be released to the Corporation, after which time the holders of the shares so called for redemption shall look only to the Corporation for payment of the redemption price of such shares.
- Notices. Any and all notices or other communications or deliveries hereunder shall be in writing and shall be deemed given and effective on the earliest of (i) the date of transmission, if such notice or communication is delivered via facsimile at the facsimile number specified in this Section 3.9 prior to or at the Close of Business on a Business Day and electronic confirmation of receipt is received by the sender, (ii) the next Business Day after the date of transmission, if such notice or communication is delivered via facsimile at the facsimile number specified in this Section on a day that is not a Business Day or later than the Close of Business on any Business Day, (iii) the Business Day following the date of mailing, if sent by a nationally recognized overnight courier service, or (iv) upon actual receipt by the party to whom such notice is required to be given. The addresses for such communications shall be: (i) if to the Corporation, attention: Chief Executive Officer and General Counsel, or (ii) if to a holder of Series B First Preferred Shares, to the address or facsimile number appearing on the Corporation's shareholder records or such other address or facsimile number as such holder may provide to the Corporation in accordance with this Section 3.9. Notwithstanding the foregoing, if the Series B First Preferred Shares are issued in book- entry form through The Depository Trust Corporation or any similar facility, such notices may be given to the holders of the Series B First Preferred Shares in any manner permitted by such facility.
- 3.10 <u>Certain Definitions</u>. As used herein, the following terms shall have the following meanings, unless the context otherwise requires:

- "Affiliate" with respect to any person shall have the meaning specified pursuant to part 1.3 of National Instrument 46-106 Prospectus and Registrations Exemptions, promulgated under the Securities Act (Ontario); provided, however, that (i) portfolio companies in which any person or any of its Affiliates has an investment shall not be deemed an Affiliate of such person, and (ii) the Corporation, any of its Subsidiaries, or any of the Corporation's other controlled Affiliates, in each case, will not be deemed to be Affiliates of Crestview for purposes of the rights, privileges, restrictions and conditions of the Series B First Preferred Shares. For purposes of this definition, "control" (including, with correlative meaning, the terms "controlled by" and "under common control with") when used in respect of any Person shall have the meaning specified pursuant to part 1.4 of National Instrument 46-106 Prospectus and Registrations Exemptions, promulgated under the Securities Act (Ontario).
- "Beneficially Own" shall mean to have "beneficial ownership" as such term is defined in the Canada Business Corporations Act and shall include, without limitation, any Person that "beneficially owns" securities for the purposes of Multilateral Instrument 61-101 Protection of Minority Security Holders in Special Transactions, promulgated under the Securities Act (Ontario).
 - "Board" shall mean the Board of Directors of the Corporation.
- "Business Day" shall mean a day that is a Monday, Tuesday, Wednesday, Thursday or Friday and is not a day on which banking institutions in New York, New York or Toronto, Ontario generally are authorized or obligated by law, regulation or executive order to close.
 - "Change of Control" shall mean the occurrence of any of the following:
 - (i) any Person (other than Crestview) shall Beneficially Own, directly or indirectly, through a purchase, merger or other acquisition transaction or series of transactions, shares of the Corporation's Equity Securities entitling such Person to exercise fifty percent (50%) or more of the total voting power of all classes of Voting Shares of the Corporation, which shall not include the Series A First Preferred Shares, other than an acquisition by the Corporation, any of the Corporation's Subsidiaries or any of the Corporation's employee benefit plans (for purposes of this clause (i), "Person" shall include, without limitation, any syndicate or group):
 - (ii) the Corporation (A) merges or consolidates with or into any other Person, another Person merges with or into the Corporation, or the Corporation conveys, sells, transfers or leases all or substantially all of the Corporation's assets to another Person or (B) engages in any recapitalization, reclassification or other transaction in which all or substantially all of the Common Shares are exchanged for or converted into cash, securities or other property, in each case other than a merger or consolidation:
 - (1) that does not result in a reclassification, conversion, exchange or cancellation of the Corporation's outstanding Common Shares; or

- which is effected solely to change the Corporation's jurisdiction of incorporation and results in a reclassification, conversion or exchange of outstanding Common Shares solely into common shares of the surviving entity; or
- where the Voting Shares outstanding immediately prior to such transaction are converted into or exchanged for Voting Shares of the surviving or transferee Person constituting a majority of the outstanding Voting Shares of such surviving or transferee Person (immediately after giving effect to such issuance) and no Person has the right to exercise fifty percent (50%) or more of the total voting power of all classes of Voting Shares of the Corporation which, for the avoidance of doubt, shall not include the Series A First Preferred Shares; or
- (iii) the Common Shares cease to be listed or quoted on any of the New York Stock Exchange, the Nasdaq Global Select Market, Nasdaq Global Market or the Toronto Stock Exchange (or any of their respective successors); or
- (iv) any transaction or series of transactions having substantially the same effect as any of the foregoing.
- "Close of Business" shall mean 5:00 p.m., New York City time, on any Business Day.
- "Common Shares" shall have the meaning ascribed to it in Section 3.3(i).
- "Corporation" shall mean Cott Corporation.
- "Credit Agreement" shall mean the Credit Agreement, dated as of August 17, 2010, among the Corporation, Cott Beverages Inc., a Georgia corporation, Cliffstar LLC, a Delaware limited liability company, and Cott Beverages Limited, a company organized under the laws of England and Wales, as Borrowers, the other Loan Parties party thereto, the Lenders party thereto, JPMorgan Chase Bank, N.A., London Branch, as UK Security Trustee, JPMorgan Chase Bank, N.A., as Administrative Agent and Administrative Collateral Agent, and General Electric Capital Corporation, as Co- Collateral Agent (as amended by that certain Amendment No. 1 to Credit Agreement, dated as of April 19, 2012, as further amended by that certain Amendment No. 2 to Credit Agreement, dated as of July 19, 2012, as further amended by that certain Amendment No. 3 to Credit Agreement, dated as of October 22, 2013, as further amended by that certain Amendment No. 4 to Credit Agreement, dated as of May 28, 2014, and as amended on the Issue Date pursuant to that certain Amendment No. 5 but for the avoidance of doubt, without giving effect to any subsequent amendment, restatement, supplement or modification thereof).
- "Crestview" shall mean Crestview DSW Investors, L.P. and any of its Affiliates, any successor entity and any other investment fund, vehicle or similar entity of which such person or an Affiliate, advisor or manager of such person serves as the general partner, manager or advisor.
 - " **Dividend**" shall have the meaning ascribed to it in Section 3.4(i)

"Dividend Payment Date" shall have the meaning ascribed to it in Section 3.4(ii). "Dividend Period" shall have the meaning ascribed to it in Section 3.4(ii). "Dividend Rate" shall mean, in each case:

From the Issue Date until the first anniversary of the Issue Date,	10.0% per annum of the Redemption Value of each Series B First Preferred Share.
From the day after the first anniversary of the Issue Date,	11.0% per annum of the Redemption Value of each Series B First Preferred Share.
From the day after the second anniversary of the Issue Date,	12.0% per annum of the Redemption Value of each Series B First Preferred Share.
From the day after the third anniversary of the Issue Date,	13.0% per annum of the Redemption Value of each Series B First Preferred Share.
From the day after the fourth anniversary of the Issue Date,	14.0% per annum of the Redemption Value of each Series B First Preferred Share.
From the day after the fifth anniversary of the Issue Date,	15.0% per annum of the Redemption Value of each Series B First Preferred Share.

- "EBITDA" shall have the meaning ascribed to it in the Credit Agreement.
- "Equity Securities" shall mean any and all shares, interests, rights to purchase, warrants, options, participations or other equivalents of or interests in (however designated) equity securities issued by the Corporation.
- "GAAP" shall mean generally accepted accounting principles in the United States. "Indebtedness" shall have the meaning ascribed to it in the Credit Agreement. "Issue Date" shall mean December 12, 2014.
- "**Junior Shares**" shall have the meaning ascribed to it in Section 3.3(i). "Liquidation" shall have the meaning ascribed to it in Section 3.5(i). "Liquidation Preference" shall have the meaning ascribed to it in Section 3.5(i).
- "Person" shall mean any individual, company, partnership, limited liability company, joint venture, association, joint stock company, trust, unincorporated organization, government or agency or political subdivision thereof or any other entity.
- "Preferred Shares" shall mean any and all series of preferred shares of the Corporation, including the Series A First Preferred Shares and the Series B First Preferred Shares.
- "Redemption Value" shall initially mean US \$1,000.00 per share of Series B First Preferred Shares, as adjusted pursuant to Section 3.4(iii).
 - "Reference Period" shall have the meaning ascribed to it in Section 3.6(ii)(F).

- "Restricted Cash" shall mean (i) any cash or cash equivalents identified on the balance sheet of the Corporation and its Subsidiaries as "restricted" and (ii) any amount to the extent that the distribution thereof to the Corporation or use thereof by the Corporation for the repayment of its consolidated Indebtedness is restricted or prohibited by law or contract.
 - "Series A First Preferred Shares" shall have the meaning ascribed to it in Paragraph 3(b).
- "Series B First Preferred Shares" shall have the meaning ascribed to it in Paragraph "Significant Subsidiary" shall mean any Subsidiary of the Corporation that would be a 3(c).
- "Significant Subsidiary" of the Corporation within the meaning of Rule 1-02 under Regulation S-X promulgated by the Securities and Exchange Commission, determined as of the date of the latest audited consolidated financial statements of the Corporation and its consolidated Subsidiaries.
- "Subsidiary" shall mean any company or corporate entity for which the Corporation owns, directly or indirectly, an amount of the voting securities, other voting rights or voting partnership interests sufficient to elect at least a majority of its board of directors or other governing body (or, if there are no such voting interests, more than fifty percent (50%) of the equity interests of such company or corporate entity).
- "Voting Shares" shall mean Equity Securities of the class or classes pursuant to which the holders thereof have the general voting power under ordinary circumstances (determined without regard to any classification of directors) to elect one or more members of the Board of the Corporation (without regard to whether or not, at the relevant time, Equity Securities of any other class or classes (other than Common Shares) shall have or might have voting power by reason of the happening of any contingency).
- 3.11 <u>Headings</u>. The headings of the paragraphs hereof are for convenience of reference only and shall not define, limit or affect any of the provisions hereof.
- 3.12 <u>Record Holders</u>. To the fullest extent permitted by applicable law, the Corporation may deem and treat the record holder of any Series B First Preferred Share as the true and lawful owner thereof for all purposes, and the Corporation shall not be affected by any notice to the contrary.
- 3.13 <u>Replacement Certificates</u>. The Corporation shall replace any mutilated certificate at the holder's expense upon surrender of that certificate to the Corporation. The Corporation shall replace certificates that become destroyed, stolen or lost at the holder's expense upon delivery to the Corporation of reasonably satisfactory evidence that the certificate has been destroyed, stolen or lost, together with any indemnity that may be required by the Corporation.
- 3.14 <u>Transfer Agent, Registrar and Paying Agent</u>. The duly appointed transfer agent, registrar and paying agent for the Series B First Preferred Shares shall initially be the Corporation. The Corporation may, in its sole discretion, remove the transfer agent in accordance with the agreement between the Corporation and the transfer agent; provided, that the Corporation shall appoint a successor transfer agent who shall accept such appointment prior to the effectiveness of

such removal. Upon any such removal or appointment, the Corporation shall send notice thereof by first class mail, postage prepaid, to the holders of the Series B First Preferred Shares.

4.0 SECOND PREFERRED SHARES

The Second Preferred Shares shall, as a class, rank as to the payment of dividends and the distribution of assets in the event of the liquidation, dissolution or winding up of the Corporation or other distribution of assets of the Corporation among shareholders for the purpose of winding up its affairs, after the first preferred shares and shall carry and be subject to the following rights, privileges, restrictions and conditions:

- 4.1 The Second Preferred Shares may be issued at any time and from time to time in one or more series, each series to consist of such number of second preferred shares as may, before the issue thereof, be determined by resolution passed by the Board of Directors of the Corporation. The number of shares of any series may from time to time be increased by the Board of Directors of the Corporation upon compliance with the same conditions as are applicable to the issue of shares of a new series.
- 4.2 The Board of Directors of the Corporation shall, subject as hereinafter provided and subject to the provisions of the *Canada Business Corporations Act*, fix, by resolution duly passed before the issue of the Second Preferred Shares of each series, the designation, rights, privileges, restrictions and conditions to be attached to the Second Preferred Shares of such series, including, but without in any way limiting or restricting the generality of the foregoing:
 - (i) provisions, if any, with respect to the rights of the holders of the Second Preferred Shares of such series to receive notice of or to attend any meeting of the shareholders of the Corporation or to vote at any such meeting;
 - (ii) whether any dividends shall be payable on the Second Preferred Shares of such series and, if dividends are to be payable thereon, the rate, amount or method of calculation of preferential dividends, whether fixed or fluctuating, whether cumulative or non-cumulative, whether such dividends are payable in money or by the issue of fully paid shares of the Corporation, the currency or currencies of payment, the date or dates and places of payment of preferential dividends and the date or dates from which such preferential dividends shall accrue;
 - (iii) the rights of the Corporation, if any, to purchase or redeem the Second Preferred Shares of such series, and the purchase or redemption price or the method of calculating the same, and the terms and conditions of any such purchase or redemption;
 - (iv) provisions, if any, with respect to the rights of the holders of the Second Preferred Shares of such series to tender such shares to the Corporation for purchase by the Corporation and to oblige the Corporation to make such purchases;
 - (v) the conversion rights, if any;

- (vi) the terms and conditions of any share purchase plan or sinking fund with respect to the Second Preferred Shares of such series; and
- (vii) the restrictions, if any, respecting payment of dividends on the Common Shares or on any other shares of the Corporation ranking junior to the Second Preferred Shares;

the whole subject to articles of amendment setting forth the designation, rights, privileges, restrictions and conditions to be attached to the Second Preferred Shares of such series and the issue of a certificate of amendment in respect thereof, and subject to the provisions now or hereafter attaching to the First Preferred Shares as a class or to any series thereof.

- The Second Preferred Shares shall, with respect to the payment of dividends, be entitled to preference over the Common Shares and over any other shares of the Corporation ranking junior to the Second Preferred Shares, but shall be subject to the prior rights in respect of the payment of dividends attaching to the First Preferred Shares, and no dividends (other than stock dividends payable in shares of the Corporation ranking junior to the Second Preferred Shares) shall at any time be declared or paid or set apart for payment on the Common Shares or on any other shares of the Corporation ranking junior to the Second Preferred Shares (less than the total number of Second Preferred Shares then outstanding) or any shares of the Corporation ranking junior to the Second Preferred Shares unless at the date of such declaration or call for redemption or purchase as the case may be, all cumulative dividends up to and including the dividend payment for the last completed period for which such cumulative dividends shall have been declared and paid or set apart for payment in respect of each series of cumulative Second Preferred Shares then issued and outstanding and any declared and unpaid non-cumulative dividends shall have been paid or set apart for payment in respect of each series of non-cumulative Second Preferred Shares then issued and outstanding.
- In the event of the liquidation, dissolution or winding up of the Corporation or other distribution of assets of the Corporation among shareholders for the purpose of winding up its affairs, the holders of the Second Preferred Shares shall be entitled to receive, subject to prior payment in full to the holders of the First Preferred Shares of all amounts payable in such circumstances on the First Preferred Shares but before any amount shall be paid to, or any property or assets of the Corporation distributed among, the holders of the Common Shares or any other shares of the Corporation ranking junior to the Second Preferred Shares, (i) the amount paid up on such Second Preferred Shares, together with, in the case of cumulative Second Preferred Shares, all unpaid cumulative dividends (which for such purpose shall be calculated as if such cumulative dividends were accruing from day to day for the period from the expiration of the last period for which cumulative dividends have been paid up to and including the date of distribution) and, in the case of non-cumulative Second Preferred Shares, all declared and unpaid non-cumulative dividends, and (ii) if such liquidation, dissolution, winding up or distribution shall be voluntary, an additional amount equal to the premium, if any, which would have been payable on the redemption of said Second Preferred Shares could not be redeemed on such date, then an additional amount equal to the greatest premium, if any, which would have been payable on the redemption of said Second Preferred

Shares respectively. After payment to the holders of the Second Preferred Shares of the amounts so payable to them, they shall not be entitled to share in any further distribution of the property or assets of the Corporation.

- 4.5 The Second Preferred Shares of each series shall rank on a parity with the Second Preferred Shares of every other series with respect to the payment of dividends and the distribution of assets in the event of the liquidation, dissolution or winding up of the Corporation, whether voluntary or involuntary, or other distribution of assets of the Corporation among shareholders for the purpose of winding up its affairs, provided, however, that in case such assets are insufficient to pay in full the amount due on all the Second Preferred Shares, then such assets shall be applied, firstly, to the payment equally and rateably of an amount equal to the amount paid up on the Second Preferred Shares of each series and the premium thereon, if any, and, secondly, *pro rata* in the payment of accrued and unpaid cumulative dividends and declared and unpaid non-cumulative dividends.
- 4.6 The holders of the Second Preferred Shares shall not, as such, be entitled as of right to subscribe for, or to purchase or receive the whole or any part of any issue of any shares, or of any bonds, debentures or other securities of the Corporation now or hereafter authorized, otherwise than in accordance with the exercise of the conversion rights, if any, which may from time to time attach to any series of Second Preferred Shares.
- 4.7 The provisions contained in Sections 4.1 to 4.6 inclusive, and in this Section 4.7 may be repealed or amended in whole or in part by articles of amendment and the issue of a certificate of amendment in respect thereof, but only with the prior approval of the holders of the Second Preferred Shares given as hereinafter specified in addition to any other approval required under the *Canada Business Corporations Act*.

The approval of the holders of the Second Preferred Shares as to any and all matters hereinbefore referred to may be given in writing by a resolution signed by all the holders of the Second Preferred Shares or by resolution passed by not less than two-thirds (2/3) of the votes cast at a meeting of the holders of the Second Preferred Shares duly called for the purpose at which meeting, when originally held, the holders of not less than a majority of the outstanding Second Preferred Shares are present in person or represented by proxy in accordance with the by-laws of the Corporation. If at any such meeting, the holders of a majority of the outstanding Second Preferred Shares are not present in person or represented by proxy within thirty (30) minutes after the time fixed for the meeting, then the meeting shall be adjourned to such date, being no less than fifteen (15) days later, and to such time and place as may be fixed by the chairman of the meeting and, at such adjourned meeting, the holders of Second Preferred Shares present in person or represented by proxy, whether or not they hold more or less than a majority of all Second Preferred Shares then outstanding, may transact the business for which the meeting was originally called, and a resolution duly passed thereat by not less than two-thirds (2/3) of the votes cast at such adjourned meeting shall constitute the approval of the holders of the Second Preferred Shares shall be given not less than twenty-one (21) days or more than fifty (50) days prior to the date fixed for such meeting and shall state the nature of the business to be transacted and the text of any resolution to be submitted to the meeting. Notice of any such adjourned meeting shall be given: (i) not less than seven (7) days prior to the date fixed for such

adjourned meeting, if the original meeting was adjourned by one or more adjournments for an aggregate of less than thirty (30) days, but it shall not be necessary to state in such notice the business for which the adjourned meeting is called or the text of any resolution to be submitted to the adjourned meeting, and (ii) as required by the *Canada Business Corporations Act*, if the original meeting was adjourned by one or more adjournments for an aggregate of thirty (30) days or more. The formalities to be observed with respect to the giving of notice of any such original meeting or adjourned meeting or any other meeting of the holders of Second Preferred Shares and the conduct thereof shall be those from time to time prescribed in the by-laws of the Corporation with respect to meetings of shareholders or in the laws governing the Corporation.

If the repeal or amendment of the provisions hereinbefore contained affects the rights of the holders of Second Preferred Shares of any series in a manner different from that in which the rights of the holders of Second Preferred Shares of any other series are affected, then such repeal or amendment shall, in addition to being approved by the holders of the Second Preferred Shares as hereinabove set forth, be approved by the holders of the Second Preferred Shares of such series so affected, and the provisions of this Section 4.7 shall apply, *mutatis mutandis*, with respect to the giving of such approval.

Any meeting of the holders of the outstanding Second Preferred Shares may be held at any time and for any purpose, without notice, if all holders of Second Preferred Shares entitled to vote at the meeting waive notice of the meeting in writing. For the purpose of waiver of notice, the words "in writing" shall, without limitation, include the sending of a telegram, telex, cable or any other form of written communication by a shareholder. Any holder of Second Preferred Shares may waive notice of any meeting either before or after the meeting is held.

Irregularities in the notice or in the giving thereof as well as the accidental omission to give notice of any meeting to, or the non-receipt of any notice by, any holder of Second Preferred Shares, shall not invalidate any action taken at any meeting.

At any meeting of the holders of Second Preferred Shares without distinction as to series, each holder of Second Preferred Shares shall be entitled to 1/25 of a vote for each \$1 (with the Canadian dollar and the United States dollar being deemed to be at par for the purposes of this Section 4.7) paid up on each Second Preferred Share held by him. At any meeting of the holders of Second Preferred Shares of any particular series, each holder shall be entitled to one (1) vote in respect of each Second Preferred Share of such series held by him.

5.0 SECOND PREFERRED SHARES SERIES 1

The first series of Second Preferred Shares, designated as Convertible Participating Voting Second Preferred Shares Series 1 (the "Second Preferred Shares Series 1"), shall consist of an unlimited number of shares. The Second Preferred Shares Series 1 shall, in addition to the rights, privileges, restrictions and conditions attaching to the Second Preferred Shares as a class (collectively, the "Second Preferred Shares Class Provisions") carry and be subject to the following rights, privileges, restrictions and conditions (collectively, the "Second Preferred Shares"):

5.1 <u>Defined Terms</u>. The following words and phrases whenever used in the Second Preferred Shares Series 1 Provisions shall have the following meanings, unless there be something in the context otherwise inconsistent therewith:

"Adjusted Conversion Value" means \$7.75 less:

- (i) upon conversion occurring at the option of the Corporation;
- (ii) after a Change-in-Control Transaction; or
- (iii) after the exercise by the holders of their conversion privilege provided herein following delivery by the Corporation of the notice of redemption contemplated by Section 5.1,

in each case, prior to the Fourth Anniversary Date, the Adjustment Amount;

"Adjusted Redemption Price" means:

- (i) until the Fourth Anniversary Date, an amount equal to the sum of (A) \$10.00 per Second Preferred Share Series 1, plus (B) the Redemption Premium per Second Preferred Share Series 1, plus (C) an amount equal to the Adjustment Amount multiplied by the number of common shares into which such Second Preferred Share Series 1 would be convertible (assuming conversion at the option of the holders); and
- (ii) from and after the Fourth Anniversary Date, an amount equal to the sum of (A) \$10.00 per Second Preferred Share Series 1, plus (B) the Redemption Premium per Second Preferred Share Series 1, plus (C) any accrued but unpaid dividends thereon;
- "Adjustment Amount" in respect of each Second Preferred Share Series 1 means, at the time of the applicable event giving rise to the calculation of the Adjustment Amount (the "Applicable Time"), an amount equal to (i) the U.S. Dollar Equivalent of the dividend paid per common share in the most recently completed fiscal quarter of the Corporation (calculated on the date of payment of the dividend) multiplied by (ii) a number which is the sum of the number of Common Shares into which 4,000,000 Second Preferred Shares Series 1 would be convertible (assuming conversion prior to the Fourth Anniversary Date at the option of the Corporation) on each successive Adjustment Date set forth in the table contained in Section 5.15(ii)(A) following the Applicable Time until the Fourth Anniversary Date, divided by 5,161,290.3;
- "business day" means a day other than a Saturday, a Sunday or any other day on which principal commercial banks are not permitted to be open in Toronto, Ontario;
- "Canadian Dollar Equivalent" means, on any day, with respect to any amount in United States dollars, the equivalent amount in Canadian dollars, converted at the Bank of Canada's noon rate of exchange for that day;

- "Change-in-Control Transaction" means a transaction or series of related transactions as a result of which any person other than Thomas H. Lee Company, and its shareholders, officers, directors, affiliates, and entities controlled or administered by any of the foregoing acquire shares carrying, in the aggregate, more than 50% of the votes attaching to all voting shares in the capital of the Corporation;
- "Closing Price" with respect to any securities on any Trading Day shall mean the closing sale price on such day or, in case no such sale takes place on such Trading Day, the average of the reported closing bid and asked prices, in each case on the Recognized Exchange on which the highest volume of trading in such securities took place on that day or, if no trading in such securities took place on that day, on the prior Trading Day on which trading took place;
- "Common Shares" shall mean common shares of the Corporation as such shares were constituted on the date of issuance of the Second Preferred Shares Series 1 and shares of any other class resulting from any reclassification or change of such shares;
- "Common Equivalent Redemption Price" means, as of any date, an amount per Second Preferred Share Series 1 equal to the amount which would have been distributable with respect to the number of common shares into which such Second Preferred Share Series 1 would be convertible as of such date had the Corporation been liquidated as of such date. Such Common Equivalent Redemption Price shall be based on the value of the Corporation on a liquidated basis and shall be as agreed upon by the Corporation and holders of the Second Preferred Shares Series 1 or, absent such agreement, within fifteen (15) days following the date on which the applicable event which gives rise to the valuation occurs, as determined by an independent appraiser selected by the Corporation, the fees and expenses of which appraiser shall be borne by the Corporation;
 - "Conversion Basis" has the meaning ascribed to it in Section 5.12(i);
 - "Conversion Factor" has the meaning ascribed to it in Section 5.15(i);
- "Current Market Value" means the U.S. Dollar Equivalent of the Closing Price of the common shares on the Recognized Exchange determined on the last Trading Day prior to the issuance of common shares (or other event in connection with which Current Market Value is to be determined) or, if no trading in the common shares took place on that day, on the prior Trading Day on which trading took place;
- "Dividend Payment Date" means the first business day following the first day of each of the first and third fiscal quarters in each fiscal year of the Corporation commencing on or about six months following the Fourth Anniversary Date;
 - "Fourth Anniversary Date" means July 7, 2002;
- "Liquidation Premium" means, at any time, an amount per Second Preferred Share Series 1 equal to the Conversion Factor less \$10.00;

- "Maximum Common Share Number" means 7,688,508 Common Shares, or such greater number of Common Shares as may be permitted to be issued by all applicable regulatory authorities;
- "Public Offering" means a sale of Common Shares in the capital of the Corporation (or securities exchangeable for or convertible into such Common Shares) pursuant to a prospectus filed under applicable securities laws and underwritten by one or more independent underwriters, where the aggregate proceeds to the Corporation are at least U.S.\$25,000,000, and not more than 25% of the shares are sold to any one purchaser or affiliated group of purchasers, other than underwriters;
- "Recognized Exchange" means any recognized stock exchange on which the common shares are listed from time to time, and on which the highest aggregate volume of trading in the common shares has taken place over the applicable period or on the applicable day, including, without limitation, the NASDAQ National Market, The Toronto Stock Exchange and The Montreal Exchange;
- "Redemption Premium" means, on any date, an amount per Second Preferred Share Series 1 equal to the Conversion Factor less \$10.00;
- "Surviving Person" means the continuing or surviving person of a merger, consolidation or other continuation with the Corporation;
- "**Trading Day**" means, if the applicable security is listed or admitted for trading on a recognized stock exchange on which the common shares are listed from time to time, a day on which trades may be made thereon; and
- "U.S. Dollar Equivalent" means, on any day, with respect to any amount in Canadian dollars, the equivalent amount in United States dollars, converted at the Bank of Canada's noon rate of exchange for that day.
- 5.2 Currency. All amounts payable pursuant hereto shall be payable in lawful money of the United States of America.
- 5.3 <u>Non-Business Day</u>. If any day on which any payment is to be made or action taken hereunder in connection with the Second Preferred Shares Series 1 is not a business day, then such payment shall be made or action taken on the next succeeding day that is a business day.
- 5.4 <u>Herein, hereto, etc.</u> The words "herein", "hereto", "hereof" and similar words refer, unless the context clearly indicates the contrary, to the whole of the Second Preferred Shares Class Provisions and not to any particular section, clause or paragraph thereof.
- 5.5 <u>Number and Gender</u>. Words importing the singular number only shall include the plural and vice versa, words importing the use of any gender shall include all genders, and words importing persons shall include firms and corporations and vice versa.
- $\underline{\text{Adjustments}}$. Without limiting or duplicating the effect of Section 5.15(ii)(C), all numbers of shares and dollar amounts used herein shall be adjusted appropriately to reflect any

stock split, consolidation, combination, stock dividend or other form of recapitalization or restructuring occurring after the date hereof.

5.7 <u>Actions by Holders</u>. Any action required or permitted to be taken hereunder by holders of Second Preferred Shares Series 1, including, without limitation, election to choose between or among alternative forms of payment, shall be deemed to be valid if approved or taken by holders of in excess of fifty percent (50%) in number of such shares then outstanding, and shall be binding upon all holders of Second Preferred Shares Series 1. In connection with the exercise by the holders of the option contemplated by Section 5.11, if no approval is obtained in the manner contemplated by this Section 5.7 within the thirty day period provided for in Section 5.11, the holders shall be deemed to have elected to receive the Adjusted Redemption Price.

5.8 Dividends.

- (i) Cash and Other Dividends
 - (A) If and whenever the Corporation shall at any time or from time to time declare and pay a cash dividend on its outstanding common shares, then the holders of Second Preferred Shares Series 1 shall be entitled to receive from the Corporation, with respect to each Second Preferred Share Series 1 held, a preferential dividend equal in amount to one-half of the same dividend to be received by a holder of the number of common shares into which such Second Preferred Share Series 1 is convertible on the record date for such dividend (assuming conversion at the option of the holder), out of which any applicable withholding taxes will be withheld. Any such dividend shall be paid on the Second Preferred Shares Series 1 at the same time such dividend shall be paid on the common shares.
 - (B) The Corporation shall declare and pay any dividend to the holders of the Second Preferred Shares Series 1 contemplated by Sections 5.16 and 5.25(iii), out of which dividend amount applicable withholding taxes will be withheld.
- Paid-in-Kind Dividends. In addition to the dividends provided for in Section 5.8(i), the holders of Second Preferred Shares Series 1 shall, from and after the Fourth Anniversary Date, be entitled to receive a cumulative preferential non-cash paid-in-kind dividend payable in additional Second Preferred Shares Series 1. Such dividend shall be payable on each Dividend Payment Date to the holders of record at the close of business on the third business day immediately preceding such Dividend Payment Date, and shall be at the rate of two and one-half percent (2 1/2%) for each six (6) months, compounded semi-annually, with daily accrual. With respect to the first dividend payable after the Fourth Anniversary Date, such dividend shall be paid on the number of Second Preferred Shares Series 1 outstanding on such Fourth Anniversary Date, and such dividends payable thereafter shall be paid on the number of such Second Preferred Shares Series 1 outstanding on the immediately preceding Dividend Payment Date. Notwithstanding the foregoing:

- (A) none of the dividends contemplated by this Section 5.8(ii) shall be payable until the holders of Second Preferred Shares Series 1 provide reasonably satisfactory evidence to the Corporation that the holders have paid or made provision for the payment, within the applicable statutory time periods, of any applicable withholding taxes eligible in connection therewith; and
- (B) any dividends contemplated by this Section 5.8(ii) which are not permitted to be paid by virtue of the maximum number of common shares into which the Second Preferred Shares Series 1 may be converted exceeding the Maximum Common Share Number shall be paid in cash at the rate of two and one-half (2 1/2%) for each six (6) months, compounded semi-annually, with daily accrual, based on the Adjusted Redemption Price, out of which any applicable withholding taxes will be withheld.
- (iii) No Additional Dividends. The holders of the Second Preferred Shares Series 1 shall not be entitled to any dividends other than or in excess of the dividends provided for in this Section 5.8.
- 5.9 <u>Liquidation, Dissolution Or Winding-Up</u>. In the event of the liquidation, dissolution or winding-up of the Corporation, whether voluntary or involuntary, or any other distribution of assets of the Corporation among its shareholders for the purpose of winding up its affairs (in any case, a "**Liquidation Event**"), each holder of the Second Preferred Shares Series 1 in accordance with the Second Preferred Shares Class Provisions shall be entitled to receive the greater of:
 - (i) an amount equal to the sum of: (A) \$10.00 per share, plus (B) the Liquidation Premium per share, plus (C) if the Liquidation Event occurs after the Fourth Anniversary Date, any accrued but unpaid dividend, before any amount shall be paid or any property or assets of the Corporation shall be distributed to the holders of the common shares or to the holders of any other shares ranking junior to the Second Preferred Shares Series 1 in any respect, out of which aggregate amount any applicable withholding taxes will be withheld; or
 - (ii) pari passu and rateably with the holders of the common shares, the Common Equivalent Redemption Price per share, out of which any applicable withholding taxes will be withheld, and less the amount of any taxes required to be paid by the Corporation pursuant to Part VI.1 of the *Income Tax Act* (Canada) as a result of such payment to such holder

After payment to the holders of the Second Preferred Shares Series 1 of the amount so payable to them, they shall not be entitled to share in any further distribution of the property or assets of the Corporation.

5.10 <u>Voting</u>. So long as any Second Preferred Share Series 1 is outstanding, each holder thereof shall be entitled to vote at all meetings of the shareholders of the Corporation. With respect to any such vote, each holder of Second Preferred Shares Series 1 shall be entitled to the number of votes equal to the number of votes to which a holder of the number of common shares

into which such Second Preferred Shares Series 1 are convertible on the record date for such vote is entitled (assuming conversion at the option of the holder).

5.11 <u>Redemption</u>.

- (i) Subject to Section 5.11(ii) and the provisions of the *Canada Business Corporations Act*, the Corporation, in the manner provided in the provisions attaching to the Second Preferred Shares as a class may, upon not less than sixty (60) days' prior written notice, redeem all, but not less than all, of the then outstanding Second Preferred Shares Series 1 on payment for each share to be redeemed of an amount per share equal to, at the option of holders of the Second Preferred Shares Series 1 given within thirty days after receipt from the Corporation of its notice of proposed redemption, either:
 - (A) the Adjusted Redemption Price; or
 - (B) the Common Equivalent Redemption Price;

in either case out of which such payment any applicable withholding tax shall be withheld. If the holders elect to receive the Common Equivalent Redemption Price, the Corporation may, at its election, pay such amount either in cash or by issuing to such holders a number of Common Shares having a value, based on the average Closing Price over the fifteen (15) consecutive Trading Days immediately preceding the date of redemption, equal to the Common Equivalent Redemption Price.

(ii) <u>Restriction on Redemption</u>. The Corporation shall not be entitled to redeem any of the Second Preferred Shares Series 1 at any time prior to the Fourth Anniversary Date unless the Common Shares shall have traded, at any time since July 7, 1998, on a Recognized Exchange at a simple average Closing Price per share of not less than \$13.00 (or, if applicable, the Canadian Dollar Equivalent thereof) during any one hundred and twenty (120) consecutive Trading Days prior to the date of the notice of redemption.

5.12 Right of Conversion

- (i) The Second Preferred Shares Series 1 shall, subject as hereinafter provided, be convertible into Common Shares on the following conversion basis (the "Conversion Basis"), namely the number of Common Shares for each Second Preferred Share Series 1 converted shall be determined by dividing the Conversion Factor (as defined in Section 5.15) in effect at the time of conversion by the Adjusted Conversion Value.
- (ii) The conversion provided for in Section 5.12(i) may be effected by:
 - (A) any of the holders of Second Preferred Shares Series 1 at any time, and
 - (B) the Corporation

- at any time prior to the Fourth Anniversary Date if the common shares shall have traded at any time, since July 7, 1998, on a Recognized Exchange at a simple average Closing Price per share of not less than \$13.00 (or, if applicable, the Canadian Dollar Equivalent thereof) during any one hundred and twenty (120) consecutive Trading Days, and
- (2) at any time or from time to time on or after the Fourth Anniversary Date.

5.13 Conversion by Holder.

(i) The conversion privilege herein provided for may be exercised by a holder of Second Preferred Shares Series 1 by notice in writing given to the transfer agent for the Second Preferred Shares Series 1 at any office for the transfer of the Second Preferred Shares Series 1 or to the Corporation at its registered office accompanied by the certificate or certificates representing Second Preferred Shares Series 1 in respect of which the holder thereof desires to exercise such right of conversion. Such notice shall be signed by such holder or his duly authorized attorney or agent and shall specify the number of Second Preferred Shares Series 1 which the holder desires to have converted. The transfer form on the certificate or certificates in question need not be endorsed, except in the circumstances hereinafter contemplated. If less than all the Second Preferred Shares Series 1 represented by a certificate or certificates accompanying any such notice are to be converted, the holder shall be entitled to receive, at the expense of the Corporation, a new certificate representing the Second Preferred Shares Series 1 comprised in the certificate or certificates surrendered as aforesaid which are not to be converted.

On any conversion of Second Preferred Shares Series 1, the share certificates for common shares of the Corporation resulting therefrom shall be issued in the name of the registered holder of the Second Preferred Shares Series 1 converted or in such name or names as such registered holder may direct in writing (either in the notice referred to above or otherwise); in any such case the transfer form on the back of the certificate in question shall be endorsed by the registered holder of the Second Preferred Shares Series 1 or his duly authorized attorney, with signature guaranteed in a manner satisfactory to the Corporation.

(ii) In the case of any Second Preferred Shares Series 1 which may be called for redemption, the right of conversion thereof shall, notwithstanding anything herein contained, cease and terminate at the close of business on the business day immediately preceding the date fixed for redemption, provided, however, that if the Corporation shall fail to redeem such Second Preferred Shares Series 1 in accordance with the notice of redemption, the right of conversion shall thereupon be restored.

(iii) Subject as hereinafter provided in this Section 5.13(iii), the right of a holder of Second Preferred Shares Series 1 to convert the same into Common Shares shall be deemed to have been exercised, and the registered holder of such Second Preferred Shares Series 1 (or any person or persons in whose name or names any such registered holder of Second Preferred Shares Series 1 shall have directed certificates representing Common Shares to be issued as provided in Section 5.13(i)) shall be deemed to have become a holder of Common Shares of record of the Corporation for all purposes on the date of surrender of certificates representing the Second Preferred Shares Series 1 to be converted accompanied by notice in writing as provided in Section 5.13(i), notwithstanding any delay in the delivery of certificates representing the Common Shares into which such Second Preferred Shares Series 1 have been converted.

5.14 <u>Conversion by Corporation</u>.

(i) In case the Corporation shall desire to exercise the right to cause conversion of the Second Preferred Shares Series 1 pursuant to Section 5.12(i), it shall fix a date for conversion and it shall deliver by hand or mail or cause to be mailed a notice of such conversion at least twenty (20) days prior to the date fixed for conversion to the holders of Second Preferred Shares Series 1 so to be converted at their last addresses as the same appear on the books of the Corporation. Such mailing shall be by ordinary mail. The notice, if delivered or mailed in the manner herein provided, shall be conclusively presumed to have been duly given, whether or not the holder receives such notice. In any case, failure to deliver or mail such notice by mail or any defect in the notice to the holder of the Second Preferred Shares Series 1 designated for conversion shall not affect the validity of the proceedings for the conversion of any other shares of Second Preferred Shares Series 1.

Each such notice shall specify the date fixed for conversion, the number of Common Shares into which each Second Preferred Share Series 1 is to be converted, the place or places for surrender of certificates representing such Second Preferred Shares Series 1 and that such Common Shares will be delivered upon presentation and surrender of certificates representing such Second Preferred Shares Series 1. If the notice of conversion is delivered prior to the Fourth Anniversary Date, such notice shall be accompanied by a certificate of an officer of the Corporation certifying that the Common Shares have traded at any time since July 7, 1998, on a Recognized Exchange at a simple average Closing Price per share of not less than \$13.00 (or, if applicable, the Canadian Dollar Equivalent thereof) during any consecutive one hundred and twenty (120) Trading Days.

(ii) As promptly as practicable upon receipt of such notice of conversion, each holder of any shares of Second Preferred Shares Series 1 shall surrender the certificate or certificates for such shares of Second Preferred Shares Series 1, duly endorsed, at a place designated for such surrender along with instructions regarding the name or names (with address) in which the certificate or certificates for Common Shares which shall be issuable on such conversion. Each such share surrendered

for conversion shall, unless the shares issuable on conversion are to be issued in the same name as the registration of such share of Second Preferred Shares Series 1, be duly endorsed by, or be accompanied by instruments of transfer in form satisfactory to the Corporation duly executed by, the holder or his duly authorized attorney, with signature guaranteed in a manner satisfactory to the Corporation.

As promptly as practicable after satisfaction of the requirements for surrender set forth above, the Corporation shall issue and shall deliver to such holder at the address designated in such instructions a certificate or certificates for the number of full shares issuable upon the conversion of such shares in accordance with the provisions of Sections 5.12 to 5.23 and a cheque or cash in respect of any fractional interest in respect of a common share arising upon such conversion, as provided in Sections 5.12 to 5.23.

Each conversion shall be deemed to have been effected as to any such certificate on the date on which the requirements set forth above in this Section 5.14 (ii) have been satisfied as to such certificate, and the person in whose name any certificate or certificates for Common Shares shall be issuable upon such conversion shall be deemed to have become on said date the holder of record of the shares represented thereby; provided however, that any such surrender on any date when the share transfer books of the Corporation shall be closed shall constitute the person in whose name the certificates are to be issued as the record holder thereof for all purposes on the next succeeding day on which such share transfer books are open, but such conversion shall be at the Conversion Factor in effect on the date upon which such Second Preferred Shares Series 1 shall have been surrendered.

5.15 Conversion Factor.

- (i) The initial conversion factor shall be \$10.00 (herein called the "Conversion Factor") subject to adjustment as provided in this Section 5.15.
- (ii) The Conversion Factor shall be adjusted from time to time by the Corporation as follows:
 - (A) Subject to Sections 5.15(ii)(B), (C) and (D), the Conversion Factor shall be adjusted semi-annually at the rate of 2.5% for each six (6) month period, compounded semi-annually with daily accrual, until the Fourth Anniversary Date, as follows:

Adjustment Date	Conversion Factor	
January 7, 1999	\$10.25	
July 7, 1999	\$10.51	
January 7, 2000	\$10.77	
July 7, 2000	\$11.04	

Adjustment Date	Conversion Factor	
January 7, 2001	\$11.31	
July 7, 2001	\$11.60	
January 7, 2002	\$11.89	
Fourth Anniversary Date	\$12.18	

For greater certainty, from and after the Fourth Anniversary Date, the Conversion Factor shall be \$12.18.

- (B) If conversion occurs prior to the Fourth Anniversary Date, either, (1) at the option of the Corporation pursuant to Section 5.12(ii)(B), (2) within ninety (90) days following a Change-in- Control Transaction, or (3) following receipt from the Corporation of a notice of redemption pursuant to Section 5.11, the Conversion Factor shall be \$12.18, as adjusted by Sections 5.15(ii)(C) and (D);
- (C) If and whenever the outstanding Common Shares shall be subdivided into a greater number of Common Shares or a stock dividend is declared in respect of Common Shares, the Conversion Factor in effect at the opening of business on the day following the day upon which such subdivision or stock dividend becomes effective shall be proportionately increased, and conversely, in case the outstanding Common Shares shall be combined into a smaller number of Common Shares, the Conversion Factor in effect at the opening of business on the day following the day upon which such combination becomes effective shall be proportionately reduced, such increase or reduction, as the case may be, to become effective immediately after the opening of business on the day following the day upon which such subdivision, stock dividend or combination becomes effective;
- (D) If and whenever the Corporation issues Common Shares (other than (1) pursuant to the exercise of employee stock options, (2) as consideration in connection with acquisitions approved by the Corporation's board of directors, or (3) in a Public Offering) at a gross sale or offering price which is less than 100% of the Current Market Value thereof, the Conversion Factor shall be adjusted to the result obtained by multiplying the Conversion Basis in effect immediately prior to the date of such issuance by a fraction:
 - (1) the numerator of which shall be the number of common shares outstanding immediately after such issuance; and
 - (2) the denominator of which shall be the sum of:

- (I) the number of common shares outstanding immediately prior to such issuance; and
- (II) the number of shares equal to the quotient obtained by dividing
 - x. the aggregate consideration received pursuant to such issuance by
 - y. the Current Market Value per common share

and by multiplying the number thereby determined by \$7.75.

(iii) Notwithstanding Section 5.15(ii), no adjustment in the Conversion Factor shall be required unless such adjustment would require an increase or decrease of at least 1% in such price; provided, however, that any adjustments which by reason of this Section 5.15(iii) are not required to be made shall be carried forward and taken into account in any subsequent adjustment. All calculations under this Section 5.15 shall be made by the Corporation and shall be made to the nearest cent or to the nearest one one-hundredth of a share, as the case may be.

No adjustment need be made for rights to purchase Common Shares pursuant to a Corporation plan for reinvestment of dividends or interest.

(iv) In any case in which this Section 5.15 provides that an adjustment shall become effective immediately after a record date for an event, the Corporation may defer until the occurrence of such event (A) issuing to the holder of any Second Preferred Share Series 1 converted after such record date and before the occurrence of such event the additional Common Shares issuable upon such conversion by reason of the adjustment required by such event over and above the common shares issuable upon such conversion before giving effect to such adjustment and (B) paying to such holder any amount in cash in lieu of any fraction pursuant to Section 5.21.

5.16 Effect of Reclassification.

(i) In the event of any reclassification or change of outstanding Common Shares (other than a change as a result of a subdivision or combination) (a "Reclassification"), each Second Preferred Share Series 1 then outstanding shall thereafter be convertible into the kind and amount of shares and other securities or properties receivable upon such reclassification or change by a holder of a number of Common Shares issuable upon conversion of such Second Preferred Share Series 1, provided that such Second Preferred Shares Series 1 shall not be convertible into any property that would cause such Second Preferred Share Series 1 to be a "short term preferred share" (as defined by the Income Tax Act (Canada)). If, pursuant hereto, any property would otherwise be distributable which would cause such Second Preferred Share Series 1 to be a "short term

preferred share", such property shall be paid as a dividend-in-kind on the day immediately prior to conversion.

(ii) Notwithstanding anything contained herein to the contrary, the Corporation will not effect any Reclassification unless, prior to the consummation hereof, (A) the Surviving Person thereof shall assume, by written instrument mailed to each holder of Second Preferred Shares Series 1 if such shares are held by fifty (50) or fewer holders or groups of affiliated holders or to each transfer agent for the Second Preferred Shares Series 1 if such shares are held by a greater number of holders, the obligation to deliver such holder such shares or other securities or properties or pay a dividend of such other properties with respect to or in exchange for common shares to which, in accordance with the foregoing provisions, such holder is entitled, and (B) proper provision is made to ensure that the holder of Second Preferred Shares Series 1 will be entitled to receive the benefits afforded by this Section 5.16. Such written instrument should provide for adjustments which shall be as nearly as equivalent as may be practicable to the adjustments provided for in this Section 5.16.

The above provisions of this section shall similarly apply to successive reclassifications and changes.

- (iii) If at any time or from time to time the Corporation takes any action that would result in a Reclassification, the Corporation shall cause to be mailed to each holder of Second Preferred Shares Series 1 at his address appearing on the books of the Corporation, as promptly as possible but in any event at least fifteen (15) days prior to the applicable date hereinafter specified, a notice stating the date on which such Reclassification is expected to become effective or occur and the date as of which it is expected that holders of record of Common Shares shall be entitled to exchange their Common Shares for securities or other property deliverable upon such Reclassification. Failure to give such notice, or any defect therein, shall not affect the legality or validity of such Reclassification. Neither the failure to give such notice nor any defect therein shall affect the legality or validity of such Reclassification.
- (iv) If this Section 5.16 applies to any event or occurrence, the adjustments provided for in Section 5.15 shall not apply to such event or occurrence.
- 5.17 <u>Transfer or Similar Taxes on Shares Issued</u>. The issue of share certificates on conversions of Second Preferred Shares Series 1 shall be made without charge to the converting holder of Second Preferred Shares Series 1 for any security transfer or similar tax in respect of the issue thereof. The Corporation shall not, however, be required to pay any such tax which may be payable in respect of any transfer involved in the issue and delivery of shares in any name other than that of the holder of any Second Preferred Shares Series 1 converted, and the Corporation shall not be required to issue or deliver any such share certificate unless and until the person or persons requesting the issue thereof shall have paid to the Corporation the amount of such tax or shall have established to the satisfaction of the Corporation that such tax has been paid.

- 5.18 <u>Shares to be Fully Paid</u>. The Corporation covenants that all Common Shares which may be issued upon conversion of Second Preferred Shares Series 1 will, upon issue, be fully paid and non-assessable by the Corporation and free from all transfer or similar taxes as described in Section 5.17, liens and charges with respect to the issue thereof.
- Reports as to Adjustments. Upon any adjustment of the Conversion Factor then in effect and any increase or decrease in the number of Common Shares issuable upon the operation of the conversion set forth in Sections 5.12 to 5.23, then, and in each such case, the Corporation shall promptly deliver to the transfer agent for the Second Preferred Shares Series 1 and the transfer agent for the Common Shares, a certificate signed by an officer of the Corporation setting forth in reasonable detail the event requiring the adjustment and the method by which such adjustment was calculated and specifying the Conversion Factor then in effect following such adjustment and the increased or decreased number of shares issuable upon the conversion set forth in Sections 5.12 to 5.23. The Corporation shall also promptly after the making of such adjustment cause its independent public accountants to give written notice to the registered holders of the Second Preferred Shares Series 1 at the address of each holder as shown on the books of the Corporation maintained by the transfer agent thereof, which notice shall state the Conversion Factor then in effect, as adjusted, and the increased or decreased number of shares issuable upon the exercise of the right of conversion granted by Sections 5.12 to 5.23, and shall set forth in reasonable detail the method of calculation of each with a brief statement of the facts requiring such adjustment. Where appropriate, such notice to holders of the Second Preferred Shares Series 1 may be given in advance and included as part of the notice required under the provisions of Section 5.14.
- Entitlement to Dividends. A holder of Second Preferred Shares Series 1 on the record date for any dividend declared payable on such share will be entitled to such dividend notwithstanding that such share is converted after such record date and before the payment date of such dividend, and the registered holder of any Common Share resulting from any conversion shall be entitled to rank equally with the registered holders of all other Common Shares in respect of all dividends declared payable to holders of Common Shares of record on any date after the date of conversion. Subject as aforesaid, no payment or adjustment will be made on account of any dividend, accrued or otherwise, on the Second Preferred Shares Series 1 converted or the Common Shares resulting from any conversion.
- 5.21 <u>Avoidance of Fractional Shares</u>. In any case where a fraction of a Common Share or Convertible Preferred Share Series 1 would otherwise be issuable hereunder, whether on conversion of one or more Second Preferred Shares Series 1, as a payment of a dividend or otherwise, the Corporation shall adjust such fractional interest by rounding up or down to the nearest whole share.
- 5.22 <u>Postponement of Issuance of Shares upon Conversion</u>. In any case where the application of the foregoing provisions results in an increase in the Conversion Factor taking effect immediately after the record date for a specific event, if any Second Preferred Shares Series 1 are converted after that record date and prior to completion of the event, the Corporation may postpone the issuance to the holder of the additional Common Shares to which he is entitled by reason of the increase in the Conversion Factor but such additional Common Shares shall be so issued and delivered to that holder upon completion of the event and the Corporation shall, in the

interim, deliver to the holder an appropriate instrument evidencing his right to receive such additional Common Shares.

- 5.23 <u>Certain Covenants</u>. Any registered holder of Second Preferred Shares Series 1 may proceed to protect and enforce its rights and the rights of such holders by any available remedy by proceeding at law or in equity to protect and enforce any such rights, whether for the specific enforcement of any provision herein or in aid of the exercise of any power granted herein, or to enforce any other proper remedy.
- 5.24 <u>Conflict</u>. In the event of any conflict or inconsistency between any of the Second Preferred Shares Series 1 Provisions and the Second Preferred Shares Class Provisions, such conflict or inconsistency shall be resolved in favour of the Second Preferred Shares Class Provisions.
- 5.25 <u>Restrictions</u>. Notwithstanding anything contained in these Second Preferred Shares Series 1 Provisions:
 - (i) the Corporation shall not, without the consent of the holders of the Second Preferred Shares Series 1, issue any equity security, or any security which may be converted into or exchanged for an equity security of the Corporation, in either case where such equity security has liquidation, voting, dividend or redemption rights, terms or privileges which are in priority to those attaching to the Second Preferred Shares Series 1;
 - (ii) the Corporation shall not pay any non-cash paid-in-kind dividends to the holders of the Second Preferred Shares Series 1 to the extent that the maximum aggregate number of Common Shares into which all of the holders' Second Preferred Shares Series 1 from time to time outstanding are convertible (assuming the maximum conversion rate), including the Second Preferred Shares Series 1 issuable on the payment of such dividend, would exceed the Maximum Common Share Number;
 - (iii) the outstanding Second Preferred Shares Series 1 shall not, at any time, be converted into a number of Common Shares which exceeds the Maximum Common Share Number, and any adjustment resulting from such prohibition shall be satisfied by the Corporation declaring and paying, immediately prior to conversion, a dividend in cash, out of which adjustment applicable withholding taxes shall be withheld; and
 - (iv) the Corporation shall not issue Common Shares in a transaction requiring adjustment of the Conversion Factor as contemplated by Section 5.15(ii)(D) to the extent that as a result of such adjustment the common shares into which the outstanding Second Preferred Shares Series 1 would be convertible (assuming the maximum conversion rate) would exceed the Maximum Common Share Number.
- 5.26 <u>Amendments</u>. Sections 5.1 to 5.27 inclusive, of the Second Preferred Shares Series 1 Provisions may be repealed, altered, modified, amended or amplified only with the sanction of the holders of the Second Preferred Shares Series 1 given as hereinafter specified, in addition to any other approval required by the *Canada Business Corporations Act*.

5.27 <u>Sanction by Holders of Second Preferred Shares Series 1</u>. The sanction of holders of the Second Preferred Shares Series 1 as to any and all matters referred to herein or as to any change adversely affecting the rights or privileges of the Second Preferred Shares Series 1 may be given and shall be deemed to have been sufficiently given if given by the holders of the Second Preferred Shares Series 1 in the manner provided in the Second Preferred Shares Series 1 Provisions with respect to the sanction of the holders of any series of the Second Preferred Shares and the said provisions shall apply *mutatis mutandis*.

6.0 COMMON SHARES

The Common Shares shall, as a class, carry and be subject to the following rights, privileges, restrictions and conditions:

- 6.1 Each Common Share of the Corporation shall entitle the holder thereof to one (1) vote at all meetings of shareholders of the Corporation (except meetings at which only holders of another specified class or series of shares are entitled to vote).
- 6.2 Subject to the prior rights with respect to the payment of dividends attaching to the First Preferred Shares, the Second Preferred Shares and to any other class of shares of the Corporation which rank prior to the Common Shares, the holders of the Common Shares shall be entitled to receive, as and when declared by the Board of Directors of the Corporation, dividends which may be paid in money, property or by the issue of fully paid shares of the Corporation.
- 6.3 In the event of the liquidation, dissolution or winding up of the Corporation or other distribution of assets of the Corporation among shareholders for the purpose of winding up its affairs, subject to the rights, privileges, restrictions and conditions attached to the First Preferred Shares and the Second Preferred Shares, either as a class or series, and to any other class or series of shares of the Corporation which rank prior to the Common Shares, the Common Shares shall entitle the holders thereof to receive the remaining property of the Corporation.
- Notwithstanding the provisions attaching to the Common Shares, the Corporation may at any time and from time to time amend the Articles of the Corporation to
 - (i) effect an exchange, reclassification or cancellation of all or part of the Common Shares;
 - (ii) create a new class of shares equal or superior to the Common Shares; or
 - (iii) increase any maximum number of authorized shares of any class of shares having rights or privileges equal or superior to the Common Shares,

without in any of such cases the holders of the Common Shares being entitled to vote separately thereon as a class; provided, however, that the holders of common shares shall always be entitled to vote in any of such cases in accordance with Section 6.1 hereof.

SCHEDULE "B" TO THE ARTICLES OF AMALGAMATION OF COTT CORPORATION CORPORATION COTT

7. Other provisions, if any

Without in any way limiting the powers conferred upon the Corporation and its directors by the *Canada Business Corporations Act*, the Board of Directors of the Corporation may from time to time on behalf of the Corporation:

- (a) borrow money upon the credit of the Corporation and limit or increase the amount to be borrowed;
- (b) issue, reissue, sell or pledge bonds, debentures, notes or other evidences of indebtedness, guarantees or securities of the Corporation, whether secured or unsecured;
- (c) to the extent permitted by the *Canada Business Corporations Act*, give guarantees on behalf of the Corporation to secure performance of any obligation of any person or give, directly or indirectly, financial assistance to any person on behalf of the Corporation by means of a loan, guarantee or otherwise;
- (d) mortgage, hypothecate, pledge or otherwise, the real or personal, movable or immovable property of the Corporation, currently owned or subsequently acquired, including book debts, rights, powers, franchises and undertakings, to secure any present or future debt obligations or any money borrowed or other debt, liability or obligation of the Corporation, including any bonds, debentures, notes, create a security interest in all or any of debenture stock, other evidences of indebtedness, guarantees or securities of the Corporation which it is by law entitled to issue; and
- (e) delegate to one or more of the directors or officers of the Corporation all or any of the powers conferred by the foregoing provisions to such extent and in such manner as the Board of Directors shall determine at the time of each such delegation.

For the purposes of the *Special Corporate Powers Act* of the Province of Quebec and without in any way limiting the powers conferred upon the Corporation and its directors by Section 189 of the *Canada Business Corporations Act*, the Corporation, may for the purpose of securing any bonds, debentures or debenture stock which it is by law entitled to issue, hypothecate, mortgage or pledge, and cede and transfer, any property, movable or immovable, present or future, which it may own.

The directors may appoint one or more additional directors, who shall hold office for a term expiring not later than the close of the next annual meeting of shareholders, but the total number of directors so appointed may not exceed one-third of the number of directors elected at the previous annual meeting of shareholders.

Innovation, Sciences et Développement économique Canada

Form 4 Articles of Amendment

Canada Business Corporations Act (CBCA) (s. 27 or 177)

Formulaire 4 Clauses modificatrices

Loi canadienne sur les sociétés par actions (LCSA) (art. 27 ou 177)

1	Corporate name Dénomination sociale COTT CORPORATION CORPORATION COTT		
2	Corporation number Numéro de la société		

The articles are amended as follows

Les statuts sont modifiés de la façon suivante

The corporation changes the province or territory in Canada where the registered office is situated to: La province ou le territoire au Canada où est situé le siège social est modifié pour :

ON

3

985163-1

The corporation makes other changes as follows: La société apporte d'autres changements aux statuts comme suit :

See attached schedule / Voir l'annexe ci-jointe

Declaration: I certify that I am a director or an officer of the corporation. Déclaration : J'atteste que je suis un administrateur ou un dirigeant de la société.

Original signed by / Original signé par Marni Morgan Poe Marni Morgan Poe 813-313-1800

Misrepresentation constitutes an offence and, on summary conviction, a person is liable to a fine not exceeding \$5000 or to imprisonment for a term not exceeding six months or both (subsection 250(1) of the CBCA)

Faire une fausse déclaration constitue une infraction et son auteur, sur déclaration de culpabilité par procédure sommaire, est passible d'une amende maximale de 5 000 \$ et d'un emprisonnement maximal de six mois, ou l'une de ces peines (paragraphe 250(1) de la LCSA).

You are providing information required by the CBCA. Note that both the CBCA and the Privacy Act allow this information to be disclosed to the public. It will be stored in personal information bank number IC/PPU-049.

Vous fournissez des renseignements exigés par la LCSA. Il est à noter que la LCSA et la Loi sur les renseignements personnels permettent que de tels renseignements soient divulgués au public. Ils seront stockés dans la banque de renseignements personnels numéro IC/PPU-049.



IC 3069 (2008/04)

Schedule / Annexe

Amendment Schedules / Annexes - Modification

The "Other Provisions" being paragraph 7 of the articles of amalgamation of the Corporation is amended by adding the following:

Meetings of shareowners of the Corporation may be held at such place inside or outside Canada, as the directors from time to time may determine.

COTT CORPORATION

CORPORATION COTT

SECOND AMENDED AND RESTATED

BY-LAW NO. 2002-1

being a by-law relating generally to the transaction of the business and affairs of the Corporation, which by-law amends and restates By-Law No. 2002-1 and all prior amendments thereto.

ARTICLE ONE

INTERPRETATION

SECTION 1.01 <u>DEFINITIONS</u>. In the by-laws of the Corporation, unless the context otherwise requires:

- "Act" means the Canada Business Corporations Act and all regulations made pursuant to it, and any statute and regulations that may be substituted therefor, as from time to time amended;
- "appoint" includes "elect" and vice versa;
- "articles" means the articles of amalgamation of the Corporation attached to the certificate of amalgamation dated December 31, 2006, as from time to time amended or restated;
- "board" means the board of directors of the Corporation;
- "by-laws" means this by-law and all other by-laws of the Corporation from time to time in force and effect;
- "Corporation" means Cott Corporation, a corporation amalgamated under the laws of Canada;
- "electronic document" means any form of representation of information or of concepts fixed in any medium in or by electronic, optical or other similar means and that can be read or perceived by a person or by any means;
- "information system" means a system used to generate, send, receive, store, or otherwise process an electronic document;
- "meeting of shareowners" means an annual meeting of shareowners or a special meeting of shareowners;
- "non-business day" means Saturday, Sunday and any other day that is a holiday as defined in the Interpretation Act (Canada);
- "recorded address" means in the case of a shareowner his latest address as recorded in the securities register; and in the case of joint shareowners the address appearing in the securities register in respect of such joint holding or the first address so appearing if there are more than one; and in the case of a director, officer, auditor or member of a committee of the board, his latest address as shown in the records of the Corporation;
- "signing officer" means, in relation to any instrument, any person authorized to sign the same on behalf of the Corporation by section 2.02, or by a resolution passed pursuant thereto;
- save as aforesaid, words and expressions defined in the Act have the same meanings when used herein or in any other by-law; and

words importing the singular number include the plural and vice versa; words importing gender include the masculine, feminine and neuter genders; and words importing persons include individuals, bodies corporate, partnerships, trusts and unincorporated organizations; and a reference to a section means that section in the by-laws in which such section appears.

In the case of any conflict between the articles and the provisions of this or any other by-law the provisions of the articles shall prevail.

ARTICLE TWO

BUSINESS OF THE CORPORATION

SECTION 2.01 <u>REGISTERED OFFICE</u>. Until changed in accordance with the Act, the address of the registered office of the Corporation will be within the place specified in the articles or within articles of amendment changing the place in which its registered office is situated.

SECTION 2.02 EXECUTION OF INSTRUMENTS. Deeds, documents, bonds, debentures, transfers, assignments, contracts, obligations, certificates and other instruments may be signed on behalf of the Corporation by two persons, one of whom holds the office of chairman of the board, chairman of the executive committee, president, vice-president or director and the other of whom holds one of the said offices or the office of secretary, treasurer, assistant-secretary or assistant-treasurer or director or any other office created by by-law or by resolution of the board. Where one person holds more than one office, he may sign any of the above said deeds, documents, bonds, debentures, transfers, assignments, contracts, obligations, certificates and other instruments on behalf of the Corporation in one or more capacities. In addition, the board may from time to time direct the manner in which and the person or persons by whom any particular instrument or class of instruments may or shall be signed. Any signing officer may affix the corporate seal to any instrument requiring the same.

SECTION 2.03 <u>BANKING AND FINANCIAL ARRANGEMENTS</u>. The banking and financial business of the Corporation including, without limitation, the borrowing of money and the giving of security therefor, shall be transacted with such banks, trust companies or other bodies corporate or organizations as may from time to time be designated by or under the authority of the board. Such banking and financial business or any part thereof shall be transacted under such agreements, instructions and delegations of powers as the board may from time to time prescribe or authorize.

SECTION 2.04 <u>VOTING RIGHTS IN OTHER BODIES CORPORATE</u>. The signing officers of the Corporation may execute and deliver proxies and arrange for the issuance of voting certificates or other evidence of the right to exercise the voting rights attaching to any securities held by the Corporation. Such instruments, certificates or other evidence shall be in favour of such person or persons as may be determined by the officers executing such proxies or arranging for the issuance of voting certificates or such other evidence of the right to exercise such voting rights. In addition, the board may from time to time direct the manner in which and the person or persons by whom any particular voting rights or class of voting rights may or shall be exercised.

SECTION 2.05 <u>WITHHOLDING INFORMATION FROM SHAREOWNERS</u>. Subject to the provisions of the Act, no shareowner shall be entitled to discovery of any information respecting any details or conduct of the Corporation's business which, in the opinion of the board, it would be inexpedient in the interests of the shareowners or the Corporation to communicate to the public. The board may from time to time determine whether and to what extent and at what time and place and under what conditions or regulations the accounts, records and documents of the Corporation or any of them shall be open to the inspection of shareowners and no shareowner shall have any right of inspecting any account, record or document of the Corporation except as conferred by the Act or authorized by the board or by resolution passed at a general meeting of shareowners.

SECTION 2.06 <u>DECLARATIONS</u>. Any officer of the Corporation or any other person appointed for the purpose by resolution of the board is authorized and empowered to appear and make answer for, on behalf and in the name of the Corporation, to writs, orders and interrogatories upon articulated facts issued out of any court and to declare for, on behalf and in the name of the Corporation, any answer to writs of attachment by way of garnishment or otherwise and to make all affidavits and sworn declarations in connection therewith or in connection with any and all judicial proceedings. Such officers and persons may make demands of abandonment or petitions for winding-up or bankruptcy orders upon any debtor of the Corporation, may attend and vote at all meetings of creditors of the Corporation's debtors and grant proxies in connection therewith, and may generally do all such things in respect thereof as they deem to be in the best interests of the Corporation.

ARTICLE THREE

BORROWING AND SECURITIES

SECTION 3.01 <u>BORROWING POWER</u>. Without limiting the borrowing powers of the Corporation as set forth in the Act or in the articles, the board may from time to time:

- (a) borrow money upon the credit of the Corporation and limit or increase the amount to be borrowed;
- (b) issue, reissue, sell or pledge bonds, debentures, notes or other evidences of indebtedness, guarantees or securities of the Corporation, whether secured or unsecured;
- (c) to the extent permitted by the Act, give guarantees on behalf of the Corporation to secure performance of an obligation of any person or give, directly or indirectly, financial assistance to any person on behalf of the Corporation by means of a loan, guarantee or otherwise; and
- (d) mortgage, hypothecate, pledge or otherwise create a security interest in all or any of the real or personal, moveable or immoveable property of the Corporation, currently owned or subsequently acquired, including book debts, rights, powers, franchises and undertakings, to secure any present or future debt obligations or any money borrowed or other debt or liability of the Corporation, including any bonds, debentures, notes, debenture stock, other evidences of indebtedness, guarantees or securities of the Corporation which it is by law entitled to issue. Nothing in this section limits or restricts the borrowing of money by the Corporation on bills of exchange or promissory notes made, drawn, accepted or endorsed by or on behalf of the Corporation.

SECTION 3.02 <u>DELEGATION</u>. The board may from time to time delegate to such one or more of the directors or officers of the Corporation as may be designated by the board all or any of the powers conferred on the board by section 3.01 or by the Act to such extent and in such manner as the board shall determine at the time of each such delegation.

ARTICLE FOUR

DIRECTORS

SECTION 4.01 <u>NUMBER OF DIRECTORS AND QUORUM</u>. Until changed in accordance with the Act, the board shall consist of such fixed number, or minimum and maximum number, of directors as may be set out in the articles.

The directors may, from time to time, fix by resolution the quorum for meetings of directors, but until otherwise fixed, a majority of the directors in office from time to time shall constitute a quorum. Subject to the provisions of section 4.06 hereof, any meeting of directors at which a quorum is present shall be competent to exercise all or any of the authorities, powers and discretions by or under the by-laws of the Corporation for the time being vested in or exercisable by the directors generally.

Subject to the Act and to the articles of the Corporation, the directors may appoint one or more directors, who shall hold office for a term expiring not later than the close of the next annual meeting of shareowners, but the total number of directors so appointed may not exceed one-third of the number of directors elected at the previous annual meeting of shareowners.

SECTION 4.02 <u>QUALIFICATION</u>. No person shall be qualified for election as a director if he is less than eighteen (18) years of age; if he is of unsound mind and has been so found by a court in Canada or elsewhere; if he is not an individual; or if he has the status of a bankrupt. A director need not be a shareowner. At least 25% of the directors, or such lesser number as may be permitted by the Act, must be resident Canadians. However, if the Corporation has fewer than four directors, at least one director, or such lesser number as may be permitted by the Act, must be a resident Canadian.

SECTION 4.03 <u>ELECTION AND TERM</u>. The election of directors shall take place at each annual meeting of shareowners at which time all the directors then in office shall cease to hold office, but, if qualified, shall be eligible for re-election. The number of directors to be elected at any such meeting shall be the number of directors then in office unless the directors or the shareowners otherwise determine. The election shall be by resolution. If an election of directors is not held at any such meeting of shareowners, the incumbent directors shall continue in office until their successors are elected.

SECTION 4.04 <u>VACATION OF OFFICE</u>. A director ceases to hold office when he dies; when he is removed from office by the shareowners in accordance with the provisions of the Act; when he ceases to be qualified for election as a director; or when his written resignation is received by the Corporation, or if a time is specified in such resignation, at the time so specified, whichever is later.

SECTION 4.05 <u>VACANCIES</u>. Subject to the Act, a quorum of the board may fill a vacancy in the board, except a vacancy resulting from an increase in the number or minimum number of directors or from a failure of the shareowners to elect the number or minimum number of directors. In the absence of a quorum of the board, or if the vacancy has arisen from a failure of the shareowners to elect the number or minimum number of directors, the board may call a special meeting of shareowners to fill the vacancy. If the board fails to call such meeting or if there are no such directors then in office, any shareowner may call the meeting. Where there is a vacancy in the board, the remaining directors may exercise all the authorities, powers and discretions of the board so long as a quorum remains in office.

SECTION 4.06 <u>CANADIAN RESIDENCY REQUIREMENTS</u>. The board shall not transact business at a meeting, other than filling a vacancy in the board, unless at least 25% of the directors present, or such lesser number as may be permitted by the Act, are resident Canadians, or, if the Corporation has fewer than 4 directors, at least one of the directors present, or such lesser number as may be permitted by the Act, is a resident Canadian. The board may, however, transact business at a meeting of directors where the required number of resident Canadian directors is not present if

- (a) a resident Canadian director who is unable to be present approves in writing or by telephonic, electronic or other communication facilities, the business transacted at the meeting; and
- (b) the required number of resident Canadian directors would have been present had that director been present at the meeting.

SECTION 4.07 <u>MEETINGS BY TELEPHONE</u>. <u>ELECTRONIC OR OTHER COMMUNICATION FACILITY</u>. A director may, to the extent and in the manner permitted by law, participate in a meeting of directors or of a committee of directors by means of a telephonic, electronic or other communication facility that permits all participants to communicate adequately with each other during the meeting, but only if all the directors of the Corporation have consented to that form of participation. A director participating in such a meeting by such means is deemed for the purposes of the Act to be present at that meeting. Any such consent shall be effective whether given before or after the meeting to which it relates and may be given with respect to all meetings of the board and of committees of the board held while a director holds office.

SECTION 4.08 PLACE OF MEETINGS. Meetings of the board may be held at any place in or outside Canada.

SECTION 4.09 <u>CALLING OF MEETINGS</u>. Meetings of the board shall be held from time to time and at such place as the board, the chairman of the executive committee, the president or any two directors may determine.

SECTION 4.10 NOTICE OF MEETING. Notice of the time and place of each meeting of the board shall be given in the manner provided in section 12.01 to each director not less than twenty-four (24) hours before the time when the meeting is to be held. A notice of a meeting of directors need not specify the purpose of or the business to be transacted at the meeting except where the Act requires such purpose or business to be specified. A director may waive notice of or otherwise consent to a meeting of the board. Such a waiver of notice may be sent in any manner, including as an electronic document and at any time before, during or after a meeting of the board. No action taken at any meeting of the board shall be invalidated by the accidental failure to give notice or sufficient notice thereof to any director.

SECTION 4.11 <u>FIRST MEETING OF NEW BOARD</u>. Provided a quorum of directors is present, each newly elected board may without notice hold its first meeting immediately following the meeting of shareowners at which such board is elected.

SECTION 4.12 <u>ADJOURNED MEETING</u>. Notice of an adjourned meeting of the board is not required if the time and place of the adjourned meeting is announced at the original meeting.

SECTION 4.13 <u>REGULAR MEETINGS</u>. The board may appoint a day or days in any month or months for regular meetings of the board at a place and hour to be named. A copy of any resolution of the board fixing the place and time of such regular meetings shall be sent to each director forthwith after being passed, but no other notice shall be required for any such regular meeting except where the Act requires the purpose thereof or the business to be transacted thereat to be specified.

SECTION 4.14 <u>CHAIRMAN</u>. The chairman of any meeting of the board shall be the first mentioned of such of the following officers as have been appointed and who is a director and is present at the meeting: chairman of the board, chairman of the executive committee, president or a vice-president, who is a director. If no such officer is present, the directors present shall choose one of their number to be chairman.

SECTION 4.15 <u>VOTES TO GOVERN</u>. At all meetings of the board every question shall be decided by a majority of the votes cast on the question. In case of an equality of votes the chairman of the meeting shall be entitled to a second or casting vote.

SECTION 4.16 <u>REMUNERATION AND EXPENSES</u>. The directors shall be paid such remuneration for their services as the board may from time to time determine. The directors shall also be entitled to be reimbursed for travelling and other expenses properly incurred by them in attending meetings of the board or any committee thereof. Nothing herein contained shall preclude any director from serving the Corporation in any other capacity and receiving remuneration therefor.

ARTICLE FIVE

COMMITTEES

SECTION 5.01 <u>COMMITTEES OF DIRECTORS</u>. The board may appoint a committee or committees of directors, however designated, and delegate to such committee or committees any of the powers of the board except those which, under the Act, a committee of directors has no authority to exercise.

SECTION 5.02 EXECUTIVE COMMITTEE. The board may designate one of the committees appointed by it as the executive committee. It shall comprise at least three (3) members who shall remain in office at the pleasure of the board and while still directors. It shall, subject to section 5.01, be vested with all the powers and authority of the board between meetings thereof. All acts and proceedings of the executive committee shall be reported to the board at the next meeting thereof, but any right granted or obligation incurred pursuant to the authority of the executive committee shall be treated as valid and binding upon the Corporation.

SECTION 5.03 <u>AUDIT COMMITTEE</u>. The board shall elect from among its number an audit committee to be composed of at least three (3) directors of whom the majority shall not be officers or employees of the Corporation or its affiliates. Members of the audit committee shall remain in office at the pleasure of the board and while still directors.

SECTION 5.04 <u>TRANSACTION OF BUSINESS</u>. Subject to the provisions of section 4.07, the powers of a committee of directors may be exercised by a meeting at which a quorum is present or by resolution in writing signed by all the members of such committee who would have been entitled to vote on that resolution at a meeting of the committee. Meetings of a committee of directors may be held at any place in or outside Canada.

SECTION 5.05 <u>PROCEDURE</u>. Unless otherwise determined by the board, each committee shall have the power to fix its quorum at not less than a majority of its members, to elect its chairman and to regulate its procedure.

ARTICLE SIX

OFFICERS

SECTION 6.01 <u>APPOINTMENT</u>. The board may from time to time appoint a chairman of the board, a chairman of the executive committee, a president, one or more vice-presidents (to which title may be added words indicating seniority or function), a secretary and such other officers as the board may determine, including one or more assistants to any of the officers so appointed. The board may specify the duties of and, in accordance with this by-law and subject to the provisions of the Act, delegate to such officers powers to manage the business and affairs of the Corporation. Subject to section 6.02, an officer may but need not be a director and one person may hold more than one office.

SECTION 6.02 <u>CHAIRMAN OF THE BOARD</u>, <u>CHAIRMAN OF THE EXECUTIVE COMMITTEE AND PRESIDENT</u>. The chairman of the board, chairman of the executive committee and the president shall each be chosen from among the directors and, if appointed, shall have such powers and duties as the board may specify.

SECTION 6.03 <u>VICE-PRESIDENT OR VICE-PRESIDENTS</u>. The vice-president or vice-presidents shall have such powers and duties as the board may specify.

SECTION 6.04 <u>SECRETARY</u>. Except as may be otherwise determined from time to time by the board, the secretary shall attend and be the secretary of all meetings of the board, shareowners and committees of the board and shall enter or cause to be entered in records kept for that purpose minutes of all proceedings thereat; he shall give or cause to be given, as and when instructed, all notices to shareowners, directors, officers, auditors and members of committees of the board; he shall be the custodian of the stamp or mechanical device generally used for affixing the corporate seal of the Corporation and of all books, papers, records, documents and instruments belonging to the Corporation, except

when some other officer or agent has been appointed for that purpose; and he shall have such other powers and duties as the board or the chief executive officer may specify.

SECTION 6.05 <u>POWERS AND DUTIES OF OTHER OFFICERS</u>. The powers and duties of all other officers shall be such as the terms of their engagement call for or as the board or the chief executive officer may specify. Any of the powers and duties of an officer to whom an assistant has been appointed may be exercised and performed by such assistant, unless the board or the chief executive officer otherwise directs.

SECTION 6.06 <u>VARIATION OF POWERS AND DUTIES</u>. The board may from time to time and subject to the provisions of the Act, vary, add to or limit the powers and duties of any officer.

SECTION 6.07 <u>TERM OF OFFICE</u>. The board, in its discretion, may remove and discharge any or all the officers of the Corporation either with or without cause at any meeting called for that purpose and may elect or appoint others in their place or places. Any officer or employee of the Corporation, not being a member of the board, may also be removed and discharged, either with or without cause, by the chairman of the board, chairman of the executive committee or president. If, however, there be a contract with an officer or employee derogating from the provisions of this section such removal or discharge shall be subject to the provisions of such contract. Otherwise each officer appointed by the board shall hold office until his successor is appointed.

SECTION 6.08 <u>TERMS OF EMPLOYMENT AND REMUNERATION</u>. The terms of employment and the remuneration of officers appointed by the board shall be settled by it from -time to time.

SECTION 6.09 <u>AGENTS AND ATTORNEYS</u>. The board, the chairman of the board, the chairman of the executive committee or the president or any person delegated by any of them shall have power from time to time to appoint agents or attorneys for the Corporation in or outside Canada with such powers of management or otherwise (including the power to sub-delegate) as may be thought fit.

SECTION 6.10 <u>FIDELITY BONDS</u>. The board, the chairman of the board, the chairman of the executive committee or the president or any person delegated by any of them may require such officers, employees and agents of the Corporation as the board deems advisable to furnish bonds for the faithful discharge of their powers and duties, in such form and with such surety as the board may from time to time determine.

ARTICLE SEVEN

PROTECTION OF DIRECTORS, OFFICERS AND OTHERS

SECTION 7.01 <u>LIMITATION OF LIABILITY</u>. No director or officer shall be liable for the acts, receipts, neglects or defaults of any other person including any director or officer or employee or agent, or for joining in any receipt or acts for conformity, or for any loss, damage or expense happening to the Corporation through the insufficiency or deficiency of title to any property acquired by or on behalf of the Corporation, or for the insufficiency or deficiency of any security in or upon which any of the moneys of the Corporation shall be invested, or for any loss or damage arising from the bankruptcy, insolvency, delictual, quasi-delictual or tortious acts of any person with whom any of the moneys, securities or other property of the Corporation shall be deposited or for any loss occasioned by an error of judgment or oversight on his part, or for any other loss, damage or misfortune whatever which may arise out of the execution of the duties of his office or in relation thereto, unless the same are occasioned by his own wilful neglect or default; provided that nothing herein shall relieve any director or officer from the duty to act in accordance with the mandatory provisions of the Act or from liability for any breach thereof.

SECTION 7.02 <u>LIMITATION OF LIABILITY</u>. Without in any manner derogating from or limiting the mandatory provisions of the Act but subject to the conditions in this by-law, the Corporation shall indemnify each director and officer of the Corporation, each former director and officer of the Corporation and each individual who acts or acted at the Corporation's request as a director or officer, or each individual acting in a similar capacity, of another entity, against all costs, charges and expenses, including an amount paid to settle an action or satisfy a judgment, reasonably incurred by the individual in respect of any civil, criminal, administrative, investigative or other proceeding in which the individual is involved because of that association with the Corporation or other entity.

SECTION 7.03 <u>ADVANCE OF COSTS</u>. The Corporation may advance moneys to a director, officer or other individual for the costs, charges and expenses of a proceeding referred to in section 7.02. The individual shall repay the moneys if the individual does not fulfil the conditions of section 7.04.

SECTION 7.04 <u>LIMITATION IN INDEMNITY</u>. The Corporation's indemnity applies, however, only to the extent that the individual seeking indemnity:

(a) acted honestly and in good faith with a view to the best interests of the Corporation, or, as the case may be, to

the best interests of the other entity for which the individual acted as director or officer or in a similar capacity at the Corporation's request; and in the case of a criminal or administrative action or proceeding that is enforced by a monetary penalty, the individual had reasonable grounds for believing that the individual's conduct was lawful.

SECTION 7.05 <u>INSURANCE</u>. Subject to the limitations contained in the Act, the Corporation may purchase and maintain such insurance for the benefit of its directors and officers as such, as the board may from time to time determine.

ARTICLE EIGHT

SHARES

SECTION 8.01 <u>ALLOTMENT</u>. Subject to the articles, shares of the Corporation may be issued at such times and to such persons and for such consideration as the board may determine and the board may from time to time allot or grant options or other rights to purchase any of the shares of the Corporation at such times and to such persons and for such consideration as the board shall determine.

SECTION 8.02 <u>COMMISSIONS</u>. Subject to the provisions of the Act, the board may from time to time authorize the Corporation to pay a commission to any person in consideration of his purchasing or agreeing to purchase shares of the Corporation, whether from the Corporation or from any other person, or procuring or agreeing to procure purchasers for any such shares.

SECTION 8.03 <u>REGISTRATION OF TRANSFER</u>. Subject to the provisions of the Act, no transfer of shares shall be registered in a securities register except upon presentation of the certificate representing such shares with a transfer endorsed thereon or delivered therewith duly executed by the registered holder or by his attorney or successor duly appointed, or upon proper instructions from the holder of uncertificated shares, in each case with such reasonable assurance or evidence of signature, identification and authority to transfer as the board may from time to time require, and upon payment of all applicable taxes and any fees required by the board.

SECTION 8.04 <u>TRANSFER AGENTS</u>. REGISTRARS AND DIVIDEND DISBURSING AGENTS. The board may from time to time appoint a registrar to maintain the securities register and a transfer agent to maintain the register of transfers and may also appoint one or more branch registrars to maintain branch securities registers and one or more branch transfer agents to maintain branch registers of transfers. The board may also from time to time appoint a dividend disbursing agent to disburse dividends. One person may be appointed to any number of the aforesaid positions. The board may at any time terminate any such appointment.

SECTION 8.05 <u>CONCLUSIVENESS OF SECURITIES REGISTER</u>. Subject to the provisions of the Act, the Corporation shall treat the person in- whose name any share is registered in the securities register as absolute owner of such share with full legal capacity and authority to exercise all rights of ownership, irrespective of any indication to the contrary through knowledge or notice or description in the Corporation's records or on the share certificate.

SECTION 8.06 SHARE CERTIFICATES. Shares of the capital stock of the Corporation may be certificated or uncertificated. Every holder of one or more shares of the Corporation shall be entitled, at his option, to a share certificate, or to a non-transferable written acknowledgement of his right to obtain a share certificate, stating the number and class or series of shares held by him as shown on the securities register. Share certificates and acknowledgements of a shareowner's right to a share certificate respectively, shall be in such form as the board shall from time to time approve. Any share certificate shall be signed in accordance with section 2.02, and need not be under the corporate seal; provided that, unless the board otherwise determines, certificates representing shares in respect of which a transfer agent and/or registrar has been appointed shall not be valid unless countersigned by or on behalf of such transfer agent and/or registrar. The signature of one of the signing officers or, in the case of share certificates which are not valid unless counter-signed by or on behalf of a transfer agent and/or registrar, the signatures of both signing officers, may be printed or mechanically reproduced upon share certificates and every such printed or mechanically reproduced signature shall for all purposes be deemed to be the signature of the officer whose signature it reproduces and shall be binding upon the Corporation. A share certificate executed as aforesaid shall be valid notwithstanding that one or both of the officers whose facsimile signature appears thereon no longer holds office at the date of issue of the certificate.

SECTION 8.07 <u>REPLACEMENT OF SHARE CERTIFICATES</u>. The board or any officer or agent designated by the board may in its or his discretion direct the issue of a new share certificate in lieu and upon cancellation of a share certificate that has been mutilated or in substitution for a share certificate claimed to have been lost, destroyed or wrongfully taken upon payment of such fee, if any, and on such terms as to indemnity, reimbursement of expenses and

evidence of loss and of title as the board may from time to time prescribe, whether generally or in any particular case.

SECTION 8.08 <u>JOINT SHAREOWNERS</u>. If two or more persons are registered as joint holders of any share, the Corporation shall not be bound to issue more than one certificate in respect thereof, and delivery of such certificate to one of such persons shall be sufficient delivery to all of them. Any one of such persons may give effectual receipts for the certificate issued in respect thereof or for any dividend, bonus, return of capital or other money payable or warrant issuable in respect of such share.

SECTION 8.09 <u>DECEASED SHAREOWNERS</u>. In the event of the death of a holder, or of one of the joint holders, of any share, the Corporation shall not be required to make any entry in the securities register in respect thereof or to make payment of any dividends thereon or other distributions in respect thereof except upon production of all such documents as may be required by law and upon compliance with the reasonable requirements of the Corporation and its transfer agent.

ARTICLE NINE

DIVIDENDS AND RIGHTS

SECTION 9.01 <u>DIVIDENDS</u>. Subject to the provisions of the Act, the board may from time to time declare dividends payable to the shareowners according to their respective rights and interests in the Corporation. Dividends may be paid in money or property or by issuing fully paid shares of the Corporation.

SECTION 9.02 <u>DIVIDEND CHEQUES</u>. A dividend payable in cash shall be paid by cheque drawn on the Corporation's bankers or one of them or those of its dividend disbursing agent to the order of each registered holder of shares of the class or series in respect of which it has been declared and mailed by pre-paid ordinary mail to such registered holder at his recorded address, unless such holder otherwise directs. In the case of joint holders the cheque shall, unless such joint holders otherwise direct, be made payable to the order of all of such joint holders and mailed to them at their recorded address. The mailing of such cheque as aforesaid, unless the same is not paid on due presentation on or after the applicable dividend payment date, shall satisfy and discharge the liability for the dividend to the extent of the sum represented thereby plus the amount of any tax which the Corporation is required to and does withhold.

SECTION 9.03 NON-RECEIPT OF CHEQUES. In the event of non-receipt of any dividend cheque by the person to whom it is sent as aforesaid, the Corporation shall issue to such person a replacement cheque for a like amount on such terms as to indemnity, reimbursement of expenses and evidence of non-receipt and of title as the board may from time to time prescribe, whether generally or in any particular case.

SECTION 9.04 <u>UNCLAIMED DIVIDENDS</u>. Any dividend unclaimed after a period of six (6) years from the date on which the same has been declared to be payable shall be forfeited and shall revert to the Corporation.

ARTICLE TEN

MEETINGS OF SHAREOWNERS

SECTION 10.01 <u>ANNUAL MEETINGS</u>. The annual meeting of shareowners shall be held at such time in each year and, subject to section 10.03, at such place as the board or failing it, the chairman of the board, the chairman of the executive committee or the president may from time to time determine, for the purpose of considering the financial statements and reports required by the Act to be placed before the annual meeting, electing directors, appointing auditors and for the transaction of such other business as may properly be brought before the meeting.

SECTION 10.02 SPECIAL MEETINGS. Subject to compliance with the Act, the board, the chairman of the board, the chairman of the executive committee or the president shall have power to call a special meeting of shareowners at any time.

SECTION 10.03 <u>PLACE OF MEETINGS</u>. Meetings of shareowners shall be held at the place where the registered office of the Corporation is situated or, if the board so determines, at some other place within Canada.

SECTION 10.04 MEETINGS BY TELEPHONE, ELECTRONIC OR OTHER COMMUNICATION FACILITY. Any person entitled to attend a meeting of shareowners may participate in the meeting, to the extent and in the manner permitted by law, by means of a telephonic, electronic or other communication facility that permits all participants to communicate adequately with each other during the meeting, if the Corporation makes available such a communication facility. A person participating in a meeting by such means is deemed for the purposes of the Act to be present at the meeting. The directors or the shareowners of the Corporation who call a meeting of shareowners

pursuant to the Act may determine that the meeting shall be held, to the extent and in the manner permitted by law, entirely by means of a telephonic, electronic or other communication facility that permits all participants to communicate adequately with each other during the meeting.

SECTION 10.05 NOTICE OF MEETINGS. Subject to compliance with the Act, notice of the time and place of each meeting of shareowners shall be given in the manner provided in section 12.01 not less than twenty-one (21) nor more than fifty (50) days before the date of the meeting to each director, to the auditor and to each shareowner who at the close of business on the record date for notice is entered in the securities register as the holder of one or more shares carrying the right to vote at the meeting. A shareowner may in any manner either before, during or after a meeting of shareowners waive notice of or otherwise consent to a meeting of shareowners.

SECTION 10.06 <u>CHAIRMAN</u>, <u>SECRETARY AND SCRUTINEERS</u>. The chairman of any meeting of shareowners shall be the first mentioned of such of the following officers as have been appointed and who is present at the meeting: chairman of the board, chairman of the executive committee, president or a vice-president who is a shareowner. If no such officer is present within fifteen (15) minutes after the time fixed for holding the meeting, the persons present and entitled to vote shall choose one of their number to be chairman. If the secretary and each assistant-secretary of the Corporation is absent, the chairman shall appoint some person, who need not be a shareowner, to act as secretary of the meeting. If desired, one or more scrutineers, who need not be shareowners, may be appointed by a resolution or by the chairman of the meeting.

SECTION 10.07 <u>PERSONS ENTITLED TO BE PRESENT</u>. The only persons entitled to be present at a meeting of shareowners shall be those entitled to vote thereat, the directors and auditors of the Corporation and others who, although not entitled to vote, are entitled or required under any provision of the Act or the articles or by-laws to be present at the meeting. Any other person may be admitted only on the invitation of the chairman of the meeting or with the consent of the meeting.

SECTION 10.08 QUORUM. Except as otherwise provided by law or by the articles, a quorum for the transaction of business at any meeting of shareowners shall be not less than two persons present in person, each being a shareowner entitled to vote thereat or a duly appointed proxy for an absent shareowner so entitled, and holding or representing the holder or holders of shares carrying not less than a majority of the voting power of all issued and outstanding shares of the Corporation entitled to vote on a particular matter to be acted on at the meeting, except that, when specified business is to be voted on by one or more classes or series of shares voting as a class, unless otherwise provided by law, regulatory authority or by the articles, the holders of not less than a majority of the voting power of the shares of such classes or series shall constitute a quorum for the transaction of such matter. If a quorum is present at the opening of the meeting of shareowners, the shareowners present may proceed with the business of the meeting, notwithstanding that a quorum is not present throughout the meeting.

If a quorum is not present at the opening of a meeting of shareowners, the shareowners present in person and entitled to be counted for the purpose of forming a quorum shall have power to adjourn the meeting from time to time to a fixed time and place without notice other than announcement at the meeting until a quorum shall be present, subject to the provisions of the Act, the articles and section 10.16 of this by-law. At any such adjourned meeting, provided a quorum is present, any business may be transacted which might have been transacted at the meeting adjourned.

SECTION 10.09 <u>RIGHT TO VOTE</u>. The shareowners entitled to vote at any meeting of shareowners shall be determined in accordance with the provisions of the Act and the articles.

SECTION 10.10 <u>PROXIES</u>. Every shareowner entitled to vote at a meeting of shareowners may appoint a proxyholder, or one or more alternate proxyholders, who need not be shareowners, to attend and act at the meeting in the manner and to the extent authorized and with the authority conferred by the proxy. A proxy shall be in writing executed by the shareowner or his attorney and shall conform with the requirements of the Act and applicable law.

SECTION 10.11 <u>TIME FOR DEPOSIT OF PROXIES</u>. The board may specify in a notice calling a meeting of shareowners a time, preceding the time of such meeting by not more than forty-eight (48) hours exclusive of non-business days, before which time proxies to be used at such-meeting must be deposited. A proxy shall be acted upon only if, prior to the time so specified, it shall have been deposited with the Corporation or an agent thereof specified in such notice or, if no such time is specified in such notice, unless it has been received by the secretary of the Corporation or by the chairman of the meeting or any adjournment thereof prior to the time of voting.

SECTION 10.12 <u>JOINT SHAREOWNERS</u>. If two or more persons hold shares jointly, any one of them present in person or represented by proxy at a meeting of shareowners may, in the absence of the other or others, vote the shares; but if two or more of those persons who are present, in person or by proxy, vote, they shall vote as one on the shares jointly held by them.

SECTION 10.13 VOTES TO GOVERN. At any meeting of shareowners every question shall, unless otherwise

required by the articles or by-laws or by law, be determined by the majority of the votes cast on the question. In case of an equality of votes, either upon a show of hands or upon a poll, the chairman of the meeting shall be entitled to a second or casting vote.

SECTION 10.14 SHOW OF HANDS. Subject to the provisions of the Act, any question at a meeting of shareowners shall be decided by a show of hands unless a ballot thereon is required or demanded as hereinafter provided. Upon a show of hands every person who is present and entitled to vote shall have one vote. Whenever a vote by show of hands shall have been taken upon a question, unless a ballot thereon is so required or demanded, a declaration by the chairman of the meeting that the vote upon the question has been carried or carried by a particular majority or not carried and an entry to that effect in the minutes of the meeting shall be prima facie evidence of the fact without proof of the number or proportion of the votes recorded in favour of or against any resolution or other proceeding in respect of the said question, and the result of the vote so taken shall be the decision of the shareowners upon the said question.

SECTION 10.15 <u>BALLOTS</u>. On any question proposed for consideration at a meeting of shareowners, and whether or not a show of hands has been taken thereon, any shareowner or proxyholder entitled to vote at the meeting may require or demand a ballot. A ballot so required or demanded shall be taken in such manner as the chairman shall direct. A requirement or demand for a ballot may be withdrawn at any time prior to the taking of the ballot. If a ballot is taken each person present shall be entitled, in respect of the shares which he is entitled to votes at the meeting upon the question, to that number of votes provided by the Act or the articles, and the result of the ballot so taken shall be the decision of the shareowners upon the said question.

SECTION 10.16 <u>ADJOURNMENT</u>. Subject to the articles, if a meeting of shareowners is adjourned for less than thirty (30) days, it shall not be necessary to give notice of the adjourned meeting, other than by announcement at the meeting that is adjourned. If a meeting of shareowners is adjourned by one or more adjournments for an aggregate of thirty (30) days or more, notice of the adjourned meeting shall be given as for required by the Act.

SECTION 10.17 <u>RESOLUTION IN WRITING</u>. A resolution in writing signed by all the shareowners entitled to vote on that resolution at a meeting of shareowners is as valid as if it had been passed at a meeting of the shareowners unless a written statement with respect to the subject matter of the resolution is submitted by a director or the auditors in accordance with the Act.

SECTION 10.18 <u>ELECTRONIC VOTING BY SHAREOWNERS</u>. Any vote at a meeting of the shareowners may be held, to the extent and in the manner permitted by law, entirely by means of a telephonic, electronic or other communication facility, if the Corporation makes available such a communication facility. Any person participating in a meeting of shareowners by electronic means as provided in section 10.04 and entitled to vote at that meeting may vote, to the extent and in the manner permitted by law, by means of the telephonic, electronic or other communication facility that the Corporation has made available for that purpose.

ARTICLE ELEVEN

DIVISIONS AND DEPARTMENTS

SECTION 11.01 <u>CREATION AND CONSOLIDATION OF DIVISIONS</u>. The board may cause the business and operations of the Corporation or any part thereof to be divided or to be segregated into one or more divisions upon such basis as the board may consider appropriate in each case. The board may also cause the business and operations of any such divisions or sub-units to be consolidated upon such basis as the board may consider appropriate in each case.

SECTION 11.02 <u>NAME OF DIVISIONS</u>. Any division or its sub-units may be designated by such name as the board may from time to time determine and may transact business, enter into contracts, sign cheques and other documents of any kind and do all acts and things under such name.

ARTICLE TWELVE

NOTICES

SECTION 12.01 <u>METHOD OF SENDING NOTICE</u>. Subject to compliance with all applicable laws, any notice (which term includes any communication or document) to be sent pursuant to the Act, the articles, the by-laws or otherwise to a shareowner, director, officer, auditor or member of a committee of the board shall be sufficiently sent if (i) delivered personally to the person to whom it is to be sent, (ii) delivered to the recorded address or mailed to the recorded address of that person by prepaid mail (iii) sent to that person at the recorded address by any means of prepaid

transmitted or recorded communication or (iv) provided as an electronic document to the information system of that person. A notice so delivered shall be deemed to have been sent when it is delivered personally or to the recorded address. A notice so mailed shall be deemed to have been sent when deposited in a post office or public letter box and shall be deemed to have been received on the fifth day after so depositing. A notice so sent by any means- of transmitted or recorded communication or provided as an electronic document shall be deemed to have been sent when dispatched by the Corporation if it uses its own facilities or information system and otherwise when delivered to the appropriate communication company or agency or its representative for dispatch. The secretary or assistant secretary may change or cause to be changed the recorded address of any shareowner, director, officer or auditor or member of a committee of the board in accordance with any information believed by him to be reliable. The recorded address of a director shall be his latest address as shown in the records of the Corporation or in the most recent notice filed under the *Corporations Information Act*, whichever is the more current.

SECTION 12.02 <u>ELECTRONIC DOCUMENTS</u>. A requirement under this by-law to provide a person with a notice, document or other information is not satisfied by the provision of an electronic document unless:

- the addressee has consented, in the manner prescribed under the Act, and has designated an information system for the receipt of the electronic document;
- (b) the electronic document is provided to the designated information system, unless otherwise prescribed in the Act;
- (c) in the case of a notice, document or other information that is required by the Act to be provided by registered mail, the provision of such notice, document or other information by the sending of an electronic document is prescribed by the Act;
- (d) the Act has been complied with;
- (e) the information in the electronic document is accessible by the sender so as to be usable for subsequent reference; and
- (f) the information in the electronic document is accessible by the addressee and capable of being retained by the addressee, so as to be usable for subsequent reference.

SECTION 12.03 NOTICE TO JOINT SHAREOWNERS. If two or more persons are registered as joint holders of any share, any notice shall be addressed to all of such joint holders but notice to one of such persons shall be sufficient notice to all of them. The address to be used for the purpose of giving notices shall be the recorded address.

SECTION 12.04 <u>COMPUTATION OF TIME</u>. In computing the date when notice must be given under any provision requiring a specified number of days' notice of any meeting or other event, the date of giving the notice shall be excluded and the date of the meeting or other event shall be included.

SECTION 12.05 <u>UNDELIVERED NOTICES</u>. If any notice given to a shareowner pursuant to section 12.01 is returned on three (3) consecutive occasions because he cannot be found, the Corporation shall not be required to give any further notices to such shareowner until he informs the Corporation in writing of his new address.

SECTION 12.06 <u>OMISSIONS AND ERRORS</u>. The accidental omission to give any notice to any shareowner, director, officer, auditor or member of a committee of the board or the non-receipt of any notice by any such person or any error in any notice not affecting the substance thereof shall not invalidate any action taken at any meeting held pursuant to such notice or otherwise founded thereon.

SECTION 12.07 PERSONS ENTITLED TO SHARES BY DEATH OR OPERATION OF LAW. Every person who, by operation of law, transfer, death of a shareowner or any other means whatsoever, shall become entitled to any share, shall be bound by every notice in respect of such share which shall have been duly given to the shareowner from whom he derives his title to such share prior to his name and address being entered on the securities register (whether such notice was given before or after the happening of the event upon which he became so entitled) and prior to his furnishing to the Corporation the proof of authority or evidence of his entitlement as provided in the Act.

SECTION 12.08 <u>WAIVER OF NOTICE</u>. Any shareowner (or his duly appointed proxyholder), director, officer, auditor or member of a committee of the board may at any time waive any notice, or waive or abridge the time for any notice, required to be given to him under any provision of the Act, the regulations thereunder, the articles, the by-laws or otherwise and such waiver or abridgement shall cure any default in the giving or in the time of such notice, as the case may be. Any such waiver or abridgement shall be in writing except a waiver of notice of a meeting of shareowners or of the board or committee thereof which may be given in any manner.

ARTICLE THIRTEEN

FISCAL YEAR

SECTION 13.01 <u>FISCAL YEAR</u>. The fiscal period of the Corporation shall terminate on such day in each year as the board of directors may from time to time determine.

ARTICLE FOURTEEN

EFFECTIVE DATE

SECTION 14.01 <u>EFFECTIVE DATE</u>. This by-law is effective from the date of the resolution of the directors adopting same and shall continue to be effective unless amended by the directors until the next meeting of shareowners of the Corporation, whereat if same is confirmed or confirmed as amended, this by-law shall continue in effect in the form in which it was so confirmed.

SECTION 14.02 <u>REPEAL</u>. Upon the date of this by-law coming into force, By-Law No. 2002-1, of the Corporation, as amended, shall be repealed, provided that such repeal shall not affect the previous operation of any by-law so repealed or affect the validity of any act done or right, privilege, obligation or liability acquired or incurred under or the validity of any contract or agreement made pursuant to any such by-law prior to its repeal.

All officers and persons acting under any by-law so repealed shall continue to act as appointed under the provisions of this by-law and all resolutions of the shareowners or board or committee thereof with continuing effect passed under any repealed by-law shall continue to be valid except to the extent inconsistent with this by-law and until amended or repealed.

The foregoing amendments to By-law No. 2002-1 of the Corporation were approved by the directors of the Corporation at a meeting held on the 14 th day of March 2007.

The foregoing amendments to By-law No. 2002-1 of the Corporation were confirmed by the shareowners of the Corporation at the annual and special meeting of shareowners held on the 26 th day of April 2007.

The foregoing Second Amended and Restated By-law No. 2002-1 of the Corporation is signed by an officer of the Corporation and hereby made.

DATED as of the 26 th day of April, 2007.

/s/ Mark R. Halperin

Mark R. Halperin Corporate Secretary

AMENDMENTS TO BY-LAW NO. 2002-1

OF

COTT CORPORATION/CORPORATION COTT (the "Corporation")

BE IT ENACTED as an amendment to By-law No. 2002-1 of the Corporation that section 8.03 and section 8.06 are hereby deleted in their entirety and restated as follows:

"SECTION 8.03 <u>REGISTRATION OF TRANSFER</u> Subject to the provisions of the Act, no transfer of shares shall be registered in a securities register except upon presentation of the certificate representing such shares with a transfer endorsed thereon or delivered therewith duly executed by the registered holder or by his attorney or successor duly appointed, or upon proper instructions from the holder of uncertificated shares, in each case with such reasonable assurance or evidence of signature, identification and authority to transfer as the board may from time to time require, and upon payment of all applicable taxes and any fees required by the board.

SECTION 8.06 SHARE CERTIFICATES. Shares of the capital stock of the Corporation may be certificated or uncertificated. Every holder of one or more shares of the Corporation shall be entitled, at his option, to a share certificate, or to a non-transferable written acknowledgement of his right to obtain a share certificate, stating the number and class or series of shares held by him as shown on the securities register. Share certificates and acknowledgements of a shareowner's right to a share certificate respectively, shall be in such form as the board shall from time to time approve. Any share certificate shall be signed in accordance with section 2.02, and need not be under the corporate seal; provided that, unless the board otherwise determines, certificates representing shares in respect of which a transfer agent and/or registrar has been appointed shall not be valid unless countersigned by or on behalf of such transfer agent and/or registrar. The signature of one of the signing officers or, in the case of share certificates which are not valid unless counter-signed by or on behalf of a transfer agent and/or registrar, the signatures of both signing officers, may be printed or mechanically reproduced upon share certificates and every such printed or mechanically reproduced signature shall for all purposes be deemed to be the signature of the officer whose signature it reproduces and shall be binding upon the Corporation. A share certificate executed as aforesaid shall be valid notwithstanding that one or both of the officers whose facsimile signature appears thereon no longer holds office at the date of issue of the certificate."

The foregoing Amendment to By-law No. 2002-1 of the Corporation, being a by-law repealing section 8.03 and section 8.06 of By-law No. 2002-1 of the Corporation, was approved by the directors of the Corporation at a meeting held on the 14 th day of March 2007.

The foregoing Amendment to By-law No. 2002-1 of the Corporation, being a by-law repealing section 8.03 and section 8.06 of By-law No. 2002-1 of the Corporation, was confirmed by the shareholders of the Corporation at the annual general meeting of shareholders held on the 26 th day of April 2007.

The foregoing Amendment to By-law No. 2002-1 of the Corporation, being a by-law repealing section 8.03 and section 8.06 of By-law No. 2002-1 of the Corporation, is signed by an officer of the Corporation and hereby made.

DATED as of the 26 th day of April, 2007.

/s/ Mark Halperin

Mark Halperin Corporate Secretary

AMENDMENT TO COTT CORPORATION SECOND AMENDED AND RESTATED BY-LAWS

BY LAW NO. 2002-2

being a by-law relating generally to the nomination of persons for election of directors of Cott Corporation (the "Corporation"), which by-law amends Second Amended and Restated By-Law No. 2002-1.

ARTICLE ONE

ADVANCE NOTICE OF NOMINATIONS OF DIRECTORS

SECTION 1.01 NOMINATION OF DIRECTORS. Only persons who are nominated in accordance with the provisions of this By-Law No. 2002-2 shall be eligible for election as directors of the Corporation. Nominations of persons for election as directors of the Corporation at any annual meeting of shareowners, or at any special meeting of shareowners called for the purpose of electing directors as set forth in the Corporation's notice of such special meeting, may only be made:

- (a) by or at the direction of the board of directors of the Corporation, including pursuant to a notice of meeting,
- (b) by or at the direction or request of one or more shareowners pursuant to a proposal submitted to the Corporation in accordance with applicable laws or a requisition of meeting submitted to the directors in accordance with applicable laws, or
- (c) by any person (a "nominating shareowner") who, at the close of business on the date of the giving of the notice provided for below and on the record date for determining shareowners entitled to vote at such meeting, is a registered holder or beneficial owner of shares that are entitled to be voted at such meeting and complies with the notice and other procedures set forth in this By-Law No. 2002-2.

SECTION 1.02 <u>TIMELY NOTICE</u>. In addition to any other requirements in this By-Law No. 2002-2 and under applicable laws, for a nomination to be made by a nominating shareowner, the nominating shareowner must have given timely notice thereof in proper written form to the secretary of the Corporation. To be timely, a nominating shareowner's notice must be received by the Secretary at the principal executive offices of the Corporation:

- (a) in the case of an annual meeting of shareowners, not less than 30 nor more than 60 days prior to the date of the annual meeting of shareowners; provided, however, that if the annual meeting of shareowners is called for a date that is less than 50 days after the date on which the first public announcement of the date of the annual meeting was made, notice by the nominating shareowner may be made not later than the close of business on the 10 th day following the date on which the public announcement of the date of the annual meeting is first made by the Corporation; and
- (b) in the case of a special meeting of shareowners (which is not also an annual meeting of shareowners), not later than the close of business on the 15 th day following the day on which the public announcement of the date of the special meeting of shareowners is first made by the Corporation.

The adjournment or postponement of a meeting of shareowners or the announcement thereof shall not commence a new time period for the giving of a nominating shareowner's notice as described above.

SECTION 1.03 PROPER WRITTEN FORM. To be in proper written form, a nominating shareowner's notice to the Secretary must set forth:

- as to each person whom the nominating shareowner proposes to nominate for election as a director:
 - (i) the name, age, business address and residential address of the nominee,
 - (ii) the principal occupation or employment of the nominee,
 - (iii) whether the nominee is a resident Canadian within the meaning of the Canadian Business Corporation Act (the "Act"),
 - (iv) the class or series and number of shares of the Corporation which are controlled or which are owned

- beneficially or of record by the nominee as of the record date for the meeting of shareowners (if such date shall then have been made publicly available and shall have occurred) and as of the date of such notice,
- (v) any relationships, agreements or arrangements, including financial, compensation and indemnity related relationships, agreements or arrangements, between the nominee or any of its affiliates and the nominating shareowner, any person acting jointly or in concert with the nominating shareowner or any of their respective affiliates,
- (vi) any other information relating to the nominee that would be required to be disclosed in a dissident's proxy circular in connection with solicitations of proxies for election of directors pursuant to the Act and applicable securities laws, and
- (vii) duly completed personal information form in respect of the nominee in the form prescribed by the principal stock exchange on which the securities of the Corporation are then listed for trading; and
- (b) as to the nominating shareowner giving the notice,
 - (i) the name and record address of the nominating shareowner,
 - (ii) the class or series and number of shares of the Corporation which are controlled or which are owned beneficially or of record by the nominating shareowner as of the record date for the meeting of shareowners (if such date shall then have been made publicly available and shall have occurred) and as of the date of such notice.
 - (iii) any derivatives or other economic or voting interests in the Corporation and any hedges implemented with respect to the nominating shareowners' interests in the Corporation,
 - (iv) any proxy, contract, arrangement, understanding or relationship pursuant to which the nominating shareowner has a right to vote any shares of the Corporation,
 - (v) whether the nominating shareowner intends to deliver a proxy circular and form of proxy to any shareowners of the Corporation in connection with the election of directors, and
 - (vi) any other information relating to the nominating shareowner that would be required to be made in a dissident's proxy circular in connection with solicitations of proxies for election of directors pursuant to the Act and applicable securities laws.

Such notice must be accompanied by the written consent of each nominee to being named as a nominee and to serve as a director, if elected. Reference to "nominating shareowner" in this section 1.03 shall be deemed to refer to each shareowner that nominates a person for election as director in the case of a nomination proposal where more than one shareowner is involved in making such nomination proposal.

SECTION 1.04 <u>FURTHER INFORMATION</u>. The Corporation may require any proposed nominee to furnish such other information as may reasonably be required by the Corporation to determine the eligibility of such proposed nominee to serve as an independent director of the Corporation or that could be material to a reasonable shareowner's understanding of the independence, or lack thereof, of such proposed nominee.

SECTION 1.05 <u>DETERMINATION OF ELIGIBILITY</u>. The chair of the meeting of shareowners at which an election for directors is held shall have the power and duty to determine whether a nomination was made in accordance with the procedures set forth in the foregoing provisions and, if any proposed nomination is not in compliance with such foregoing provisions, to declare that such defective nomination shall be disregarded. The Board may, in its sole discretion, waive any requirement in this By-Law No. 2002-2.

SECTION 1.06 <u>DISCUSSION PERMITTED</u>. Nothing in this By-Law No. 2002-2 shall be deemed to preclude discussion by a shareowner (as distinct from the nomination of directors) at a meeting of shareowners of any matter it is entitled to discuss pursuant to the Act.

SECTION 1.07 <u>MEANING OF PUBLIC ANNOUNCEMENT</u>. For purposes of this By-Law No. 2002-2, "public announcement" shall mean disclosure in a press release reported by a national news service in Canada or the United States or in a document publicly filed by the Corporation under its profile on the System of Electronic Document

Analysis and Retrieval at www.sedar.com or the Electronic Data Gathering, Analysis and Retrieval system at www.sec.gov/edgar.shtml .

SECTION 1.08 NOTICE. Notwithstanding any other provision of the by-laws of the Corporation, notice given to the secretary pursuant to this By-Law No. 2002-2 may only be given by personal delivery, facsimile transmission or by email (at such email address as may be stipulated from time to time by the secretary for purposes of this notice), and shall be deemed to have been given and made only at the time it is served by personal delivery to the Secretary at the address of the principal executive offices of the Corporation, email (at the address as aforesaid) or sent by facsimile transmission (provided that receipt of confirmation of such transmission has been received); provided that if such delivery or electronic communication is made on a day which is a not a business day or later than 5:00 p.m. (Toronto time) on a day which is a business day, then such delivery or electronic communication shall be deemed to have been made on the subsequent day that is a business day.

ARTICLE TWO

GENERAL

SECTION 2.01 <u>EFFECTIVE DATE</u>. This By-Law No. 2002-2 is effective from the date of the resolution of the directors adopting same and shall continue to be effective unless amended by the directors until the next meeting of shareowners of the Corporation, whereat if same is confirmed or confirmed as amended, this by-law shall continue in effect in the form in which it was so confirmed.

SECTION 2.02 <u>AMENDMENT</u>. Second Amended and Restated By-Law. No 2002-1, as amended from time to time, of the by-laws of the Corporation and this By-Law No. 2002-2 shall be read together and shall have effect, so far as practicable, as though all the provisions thereof were contained in one by-law of the Corporation. All terms contained in this by-law which are defined by Second Amended and Restated by-Law No. 2002-1, as amended by from to time, of the by-laws of the Corporation shall, for all purposes hereof, have the meanings given to such terms in the said By-law, unless expressly stated otherwise or the context otherwise required.

The foregoing amendments to By-law No. 2002-1 of the Corporation were approved by the directors of the Corporation at a meeting held on the 29 h day of October, 2013.

The foregoing By-Law No. 2002-2 of the Corporation is signed by an officer of the Corporation and hereby made.

DATED as of the 29 th day of October, 2013.

/s/ Marni Morgan Poe

Marni Morgan Poe Vice President, General Counsel and Secretary

THIRD AMENDMENT TO SECOND AMENDED AND RESTATED BY-LAW NO. 2002-1 OF COTT CORPORATION CORPORATION COTT (the "Corporation")

BE IT ENACTED as a third amendment to By-law No. 2002-1 dated April 26, 2007 of the Corporation that section 10.03 is hereby deleted in its entirety and replace with the following:

" SECTION 10.03 PLACE OF MEETINGS

Meetings of shareowners of the Corporation may be held at such place inside or outside of Canada, as the directors may from time to time determine."

The foregoing Third Amendment to By-law No. 2002-1 of the Corporation, being a by-law repealing section 10.03 of By-law No. 2002-1 of the Corporation, was approved by the board of directors of the Corporation pursuant to a meeting held on February 28, 2018.

The foregoing Third Amendment to By-law No. 2002-1 of the Corporation, being a by-law repealing section 10.03 of By-law No. 2002-1 of the Corporation, was confirmed by the shareowners of the Corporation pursuant to a special meeting held on May 1, 2018.

The foregoing Third Amendment to By-law No. 2002-1 of the Corporation is signed by an officer of the Corporation and hereby made.

[Signature page to immediately follow]

 \boldsymbol{DATED} as of the 1 st day of May, 2018.

/s/ Marni Morgan Poe

Name: Marni Morgan Poe

Title: Vice-President, General Counsel and Secretary